UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

	FORM 10-Q	
(Mark C	One)	
X	QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) 1934	OF THE SECURITIES EXCHANGE ACT OF
	For the quarterly period ended Septe	mber 29, 2013
	TRANSITION REPORT UNDER SECTION 13 OR 15(d) OF TH	E SECURITIES EXCHANGE ACT OF 1934
	For the transition period fromto	0
	Commission File Number: 001-322	253
		<u> </u>
	EnerSys	
	(Exact name of registrant as specified in it	ts charter)
	Delaware	23-3058564
	(State or other jurisdiction of incorporation or organization)	(I.R.S. Employer Identification No.)
	2366 Bernville Road Reading, Pennsylvania 19605 (Address of principal executive offices) (Zi	p Code)
	Registrant's telephone number, including area co	de: 610-208-1991
12 montl	by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 of hs (or for such shorter period that the registrant was required to file such reports), and (2) has beginning \square NO.	\','
posted p	by check mark whether the registrant has submitted electronically and posted on its corporate W ursuant to Rule 405 of Regulation S-T ($\S232.405$ of this chapter) during the preceding 12 month such files). Yes \boxtimes No \square	
	by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer," "accelerated filer," and "smaller reporting company" in Rule 12b-2 of the Security	
Large ac	ccelerated filer	Accelerated filer
Non-acc	celerated filer \Box (Do not check if a smaller reporting company)	Smaller reporting company \Box
Indicate	by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Securi	ties Exchange Act of 1934). \square YES \boxtimes NO.
	Common Stock outstanding at October 31, 2013:	47,401,370 shares

ENERSYS INDEX – FORM 10-Q

		Page
PART I – F	FINANCIAL INFORMATION	
Item 1.	<u>Financial Statements</u>	
	Consolidated Condensed Balance Sheets (Unaudited) as of September 29, 2013 and March 31, 2013	<u>3</u>
	Consolidated Condensed Statements of Income (Unaudited) for the Quarters Ended September 29, 2013 and September 30, 2012	<u>4</u>
	Consolidated Condensed Statements of Income (Unaudited) for the Six Months Ended September 29, 2013 and September 30, 2012	<u>5</u>
	Consolidated Condensed Statements of Comprehensive Income (Unaudited) for the Quarters and Six Months Ended September 29, 2013 and September 30, 2012	<u>6</u>
	Consolidated Condensed Statements of Cash Flows (Unaudited) for the Six Months Ended September 29, 2013 and September 30, 2012	<u>s</u> Z
	Notes to Consolidated Condensed Financial Statements (Unaudited)	<u>8</u>
	1. Basis of Presentation	<u>8</u>
	2. Inventories	<u>8</u>
	3. Fair Value of Financial Instruments	<u>9</u>
	4. Derivative Financial Instruments	<u>10</u>
	5. Income Taxes	<u>13</u>
	6. Warranties	<u>13</u>
	7. Commitments, Contingencies and Litigation	<u>14</u>
	8. Restructuring Plans	<u>14</u>
	9. <u>Debt</u>	<u>15</u>
	10. Retirement Plans	<u>17</u>
	11. Stock-Based Compensation	<u>17</u>
	12. Stockholders' Equity and Noncontrolling Interests	<u>18</u>
	13. Earnings Per Share	20
	14. Business Segments	21
	15. Subsequent Events	<u>21</u>
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>22</u>
Item 3.	Quantitative and Qualitative Disclosures about Market Risk	<u>31</u>
Item 4.	Controls and Procedures	<u>33</u>
PART II –	OTHER INFORMATION	
Item 1.	<u>Legal Proceedings</u>	<u>34</u>
Item 1A.	Risk Factors	<u>34</u>
Item 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	<u>34</u>
Item 4.	Mine Safety Disclosures	<u>35</u>
Item 6.	<u>Exhibits</u>	<u>36</u>
SIGNATU:	<u>res</u>	<u>37</u>

<u>37</u>

PART I – FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

ENERSYS Consolidated Condensed Balance Sheets (Unaudited) (In Thousands, Except Share and Per Share Data)

		September 29, 2013		March 31, 2013
Assets				
Current assets:				
Cash and cash equivalents	\$	272,690	\$	249,348
Accounts receivable, net of allowance for doubtful accounts: September 29, 2013 - \$9,646; March 31, 2013 - \$9,292		467,704		448,068
Inventories, net		352,134		353,941
Deferred taxes		36,866		37,786
Prepaid and other current assets		65,599		63,819
Total current assets		1,194,993	_	1,152,962
Property, plant, and equipment, net		354,942		350,126
Goodwill		351,911		345,499
Other intangible assets, net		103,040		103,701
Other assets		36,187		35,579
Total assets	\$	2,041,073	\$	1,987,867
Liabilities and Equity	_		_	
Current liabilities:				
Short-term debt	\$	20,676	\$	22,702
Current portion of long-term debt and capital lease obligations		234		311
Accounts payable		233,502		249,359
Accrued expenses		208,403		195,187
Total current liabilities	_	462,815	_	467,559
Long-term debt and capital lease obligations		159,134		155,476
Deferred taxes		89,220		88,036
Other liabilities		78,220		90,418
Total liabilities	_	789,389		801,489
Commitments and contingencies		_		_
Redeemable noncontrolling interests		9,439		11,095
Equity:				
Common Stock, \$0.01 par value per share, 135,000,000 shares authorized; 53,257,321 shares issued and 47,465,952 outstanding at September 29, 2013; 52,970,281 shares issued and 47,840,204 shares outstanding		500		500
at March 31, 2013		532		529
Additional paid-in capital		505,903		501,646
Treasury stock, at cost, 5,791,369 shares held as of September 29, 2013; 5,130,077 shares held as of March 31, 2013		(134,369)		(100,776)
Retained earnings		797,331		727,347
Accumulated other comprehensive income		66,940		40,655
Total EnerSys stockholders' equity		1,236,337		1,169,401
Nonredeemable noncontrolling interests		5,908		5,882
Total equity		1,242,245		1,175,283
Total liabilities and equity	\$	2,041,073	\$	1,987,867

See accompanying notes.

ENERSYS Consolidated Condensed Statements of Income (Unaudited) (In Thousands, Except Share and Per Share Data)

		Quarter ended			
		September 29, 2013	Sej	ptember 30, 2012	
Net sales	\$	568,847	\$	554,212	
Cost of goods sold		424,497		415,873	
Gross profit		144,350		138,339	
Operating expenses		82,226		74,159	
Restructuring charges		1,119		1,295	
Operating earnings		61,005		62,885	
Interest expense		4,119		4,942	
Other (income) expense, net		515		(1,794)	
Earnings before income taxes		56,371		59,737	
Income tax expense		15,220		16,726	
Net earnings		41,151		43,011	
Net losses attributable to noncontrolling interests		(188)		(779)	
Net earnings attributable to EnerSys stockholders	\$	41,339	\$	43,790	
Net earnings per common share attributable to EnerSys stockholders:	_				
Basic	\$	0.87	\$	0.91	
Diluted	\$	0.84	\$	0.90	
Dividends per common share	\$	0.125	\$	_	
Weighted-average number of common shares outstanding:	-				
Basic		47,573,496		48,188,331	
Diluted		49,405,818		48,719,916	

See accompanying notes.

ENERSYS Consolidated Condensed Statements of Income (Unaudited) (In Thousands, Except Share and Per Share Data)

		Six months ended				
	Sept	ember 29, 2013	Sept	ember 30, 2012		
Net sales	\$	1,166,144	\$	1,148,122		
Cost of goods sold		881,655		861,477		
Gross profit		284,489		286,645		
Operating expenses		159,336		151,840		
Restructuring charges		1,540		1,665		
Operating earnings		123,613		133,140		
Interest expense		8,390		9,674		
Other (income) expense, net		2,873		(544)		
Earnings before income taxes		112,350		124,010		
Income tax expense		30,782		35,435		
Net earnings		81,568		88,575		
Net losses attributable to noncontrolling interests		(618)		(1,019)		
Net earnings attributable to EnerSys stockholders	\$	82,186	\$	89,594		
Net earnings per common share attributable to EnerSys stockholders:						
Basic	\$	1.72	\$	1.86		
Diluted	\$	1.67	\$	1.84		
Dividends per common share	\$	0.25	\$	_		
Weighted-average number of common shares outstanding:						
Basic		47,721,239		48,044,767		
Diluted		49,355,381		48,573,454		

ENERSYS Consolidated Condensed Statements of Comprehensive Income (Unaudited) (In Thousands)

	Quarte	er ended	Six mon	nths ended	
	September 29, 2013	September 30, 2012	September 29, 2013	September 30, 2012	
Net earnings	\$ 41,151	\$ 43,011	\$ 81,568	\$ 88,575	
Other comprehensive income (loss):					
Net unrealized gain on derivative instruments, net of tax	3,778	7,937	2,290	2,591	
Pension funded status adjustment, net of tax	115	(131)	316	84	
Foreign currency translation adjustment	24,601	15,794	22,667	(16,181)	
Total other comprehensive income (loss), net of tax	28,494	23,600	25,273	(13,506)	
Total comprehensive income	69,645	66,611	106,841	75,069	
Comprehensive loss attributable to noncontrolling interests	(399)	(676)	(1,630)	(1,440)	
Comprehensive income attributable to EnerSys stockholders	\$ 70,044	\$ 67,287	\$ 108,471	\$ 76,509	

See accompanying notes.

ENERSYS Consolidated Condensed Statements of Cash Flows (Unaudited) (In Thousands)

		Six mon	ths er	ıded
	Septe	mber 29, 2013		September 30, 2012
Cash flows from operating activities				
Net earnings	\$	81,568	\$	88,575
Adjustments to reconcile net earnings to net cash provided by operating activities:				
Depreciation and amortization		25,320		25,070
Derivatives not designated in hedging relationships:				
Net losses		180		510
Cash settlements		(217)		(1,336)
Provision for doubtful accounts		612		(120)
Deferred income taxes		144		(902)
Non-cash interest expense		4,404		4,310
Stock-based compensation		7,592		7,198
Gain on disposal of property, plant, and equipment		(310)		(74)
Changes in assets and liabilities:				
Accounts receivable		(15,508)		(11,533)
Inventories		4,104		(14,077)
Prepaid and other current assets		1,314		(1,228)
Other assets		858		1,858
Accounts payable		(18,936)		(15,528)
Accrued expenses		237		(2,557)
Other liabilities		(185)		(1,396)
Net cash provided by operating activities	-	91,177		78,770
Cash flows from investing activities				
Capital expenditures		(24,736)		(26,674)
Purchases of businesses		(856)		_
Proceeds from disposal of property, plant, and equipment		1,263		89
Net cash used in investing activities		(24,329)	_	(26,585)
Cash flows from financing activities		, , ,		, ,
Net increase in short-term debt		443		5,402
Proceeds from revolving credit borrowings		_		173,250
Repayments of revolving credit borrowings		_		(182,800)
Proceeds from long-term debt - other		_		5,556
Payments of long-term debt - other		_		(10,638)
Deferred financing costs		(823)		
Capital lease obligations		(161)		(282)
Option proceeds (taxes paid related to net share settlement of equity awards), net		(7,947)		8,922
Excess tax benefits from exercise of stock options and vesting of equity awards		4,614		4,965
Purchase of treasury stock		(33,593)		_
Dividends paid to stockholders		(11,902)		_
Purchase of noncontrolling interests				(2,131)
Proceeds from noncontrolling interests		<u> </u>		613
Net cash (used in) provided by financing activities		(49,369)		2,857
Effect of exchange rate changes on cash and cash equivalents		5,863		(3,190)
Net increase in cash and cash equivalents		23,342	_	51,852
Cash and cash equivalents at beginning of period		249,348		160,490
Cash and cash equivalents at end of period	\$	272,690	\$	212,342

See accompanying notes.

ENERSYS NOTES TO CONSOLIDATED CONDENSED FINANCIAL STATEMENTS (Unaudited) (In Thousands, Except Share and Per Share Data)

1. Basis of Presentation

The accompanying interim unaudited consolidated condensed financial statements of EnerSys (the "Company") have been prepared in accordance with accounting principles generally accepted in the United States for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all the information and notes required for complete financial statements. In the opinion of management, the unaudited consolidated condensed financial statements include all normal recurring adjustments considered necessary for the fair presentation of the financial position, results of operations, and cash flows for the interim periods presented. The financial statements should be read in conjunction with the Consolidated Financial Statements and Notes thereto included in the Company's 2013 Annual Report on Form 10-K (SEC File No. 001-32253), which was filed on May 28, 2013.

The Company reports interim financial information for 13-week periods, except for the first quarter, which always begins on April 1, and the fourth quarter, which always ends on March 31. The four quarters in fiscal 2014 end on June 30, 2013, September 29, 2013, December 29, 2013, and March 31, 2014, respectively. The four quarters in fiscal 2013 ended on July 1, 2012, September 30, 2012, December 30, 2012, and March 31, 2013, respectively.

The consolidated condensed financial statements include the accounts of the Company and its wholly-owned subsidiaries and any partially owned subsidiaries that the Company has the ability to control. All intercompany transactions and balances have been eliminated in consolidation.

The Company also consolidates certain subsidiaries in which the noncontrolling interest party has within its control the right to require the Company to redeem all or a portion of its interest in the subsidiary. The redeemable noncontrolling interests are reported at their estimated redemption value. Any adjustment to the redemption value impacts retained earnings but does not impact net income or comprehensive income. Noncontrolling interests which are redeemable only upon future events, the occurrence of which is not currently probable, are recorded at carrying value.

Recently Adopted Accounting Pronouncements

In February 2013, the Financial Accounting Standards Board ("FASB") finalized the disclosure requirements on how entities should present financial information about reclassification adjustments from accumulated other comprehensive income in ASU No. 2013-02, "Comprehensive Income (Topic 220): Reporting of Amounts Reclassified Out of Accumulated Other Comprehensive Income." The standard requires that companies present either in a single note or parenthetically on the face of the financial statements, the effect of significant amounts reclassified from each component of accumulated other comprehensive income based on its source and the income statement line items affected by the reclassification. If a component is not required to be reclassified to net income in its entirety, companies would instead cross-reference to the related footnote for additional information. The Company adopted ASU No. 2013-02 as of April 1, 2013, and the adoption did not have a material impact on the Company's consolidated condensed financial statements.

2. Inventories

Inventories, net consist of:

	Septen	nber 29, 2013	March 31, 2013		
Raw materials	\$	88,985	\$ 88,787		
Work-in-process		113,922	113,119		
Finished goods		149,227	152,035		
Total	\$	352,134	\$ 353,941		

3. Fair Value of Financial Instruments

Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. The Company uses the following valuation techniques to measure fair value for its financial assets and financial liabilities:

Level 1 Inputs are unadjusted quoted prices in active markets for identical assets or liabilities.

Level 2 Inputs are quoted prices for similar assets or liabilities in an active market, quoted prices for identical or similar assets or liabilities in markets that are not active, inputs other than quoted prices that are observable and market-corroborated inputs which are derived principally from or corroborated by observable market data.

Level 3 Inputs are derived from valuation techniques in which one or more significant inputs or value drivers are unobservable.

The following tables represent the financial assets and (liabilities), measured at fair value on a recurring basis as of September 29, 2013 and March 31, 2013 and the basis for that measurement:

	Total Fair Value Measurement September 29, 2013		Quoted Price in Active Markets for Identical Assets (Level 1)		Significant Other Observable Inputs (Level 2)	Significant Unobservable Inputs (Level 3)	
Lead forward contracts	\$ 1,079			\$	1,079	\$	_
Foreign currency forward contracts	(559)		_		(559)		_
Total derivatives	\$ 520	\$		\$	520	\$	_

	Total Fair Value		Quoted Price in Active Markets for Identical Assets (Level 1)		Significant Other Observable Inputs (Level 2)		Significant Unobservable Inputs (Level 3)	
Interest rate swap agreements	\$	(654)	\$	_	\$	(654)	\$	_
Lead forward contracts		(2,832)		_		(2,832)		_
Foreign currency forward contracts		(11)		_		(11)		_
Total derivatives	\$	(3,497)	\$	_	\$	(3,497)	\$	_

The fair values of interest rate swap agreements are based on observable prices as quoted for receiving the variable three month LIBOR and paying fixed interest rates and, therefore, were classified as Level 2. At September 29, 2013, the Company had no interest rate swap agreements.

The fair values of lead forward contracts are calculated using observable prices for lead as quoted on the London Metal Exchange ("LME") and, therefore, were classified as Level 2.

The fair values for foreign currency forward contracts are based upon current quoted market prices and are classified as Level 2 based on the nature of the underlying market in which these derivatives are traded.

Financial Instruments

The fair values of the Company's cash and cash equivalents, accounts receivable and accounts payable approximate carrying value due to their short maturities.

The fair value of the Company's short-term debt approximates its carrying value, as it is variable rate debt and the terms are comparable to market terms as of the balance sheet dates and is classified as Level 2.

The Company's senior unsecured 3.375% convertible notes ("Convertible Notes"), with a face value of \$172,500, were issued when the Company's stock price was trading at \$30.19 per share. On September 29, 2013, the Company's stock price closed at \$60.55 per share. The Convertible Notes have a conversion option at \$40.60 per share. The fair value of these notes represent the trading values based upon quoted market prices and are classified as Level 2. The Convertible Notes were trading at 154% of face value on September 29, 2013, and 126% of face value on March 31, 2013.

See Note 8 to the Consolidated Financial Statements included in the Company's 2013 Annual Report on Form 10-K for more details.

The carrying amounts and estimated fair values of the Company's derivatives and Convertible Notes at September 29, 2013 and March 31, 2013 were as follows:

	 Sep	tember 29,	2013								
	arrying Amount			Fair Value		Carrying Amount					
Financial assets:										_	
Derivatives (1)	\$ 1,079		\$	1,079		\$	241		\$	241	
Financial liabilities:											
Convertible Notes	\$ 159,000	(2)	\$	265,650	(3)	\$	155,273	(2)	\$	217,350	(3)
Derivatives (1)	559			559			3,738			3,738	

- (1) Represents interest rate swap agreements, lead and foreign currency hedges (see Note 4 for asset and liability positions of the interest rate swap agreements, lead and foreign currency hedges at September 29, 2013 and March 31, 2013).
- (2) The carrying amounts of the Convertible Notes at September 29, 2013 and March 31, 2013 represent the \$172,500 principal value, less the unamortized debt discount (see Note 9 for further details).
- (3) The fair value amounts of the Convertible Notes at September 29, 2013 and March 31, 2013 represent the trading values of the Convertible Notes with a principal value of \$172.500.

4. Derivative Financial Instruments

The Company utilizes derivative instruments to reduce its exposure to fluctuations in commodity prices, foreign exchange rates and interest rates, under established procedures and controls. The Company does not enter into derivative contracts for speculative purposes. The Company's agreements are with creditworthy financial institutions and the Company anticipates performance by counterparties to these contracts and therefore no material loss is expected.

Derivatives in Cash Flow Hedging Relationships

Lead Hedge Forward Contracts

The Company enters into lead hedge forward contracts to fix the price for a portion of its lead purchases. Management considers the lead hedge forward contracts to be effective against changes in the cash flows of the underlying lead purchases. The vast majority of such contracts are for a period not extending beyond one year and the notional amounts at September 29, 2013 and March 31, 2013 were 94.2 million pounds and 56.3 million pounds, respectively.

Foreign Currency Forward Contracts

The Company purchases lead and other commodities in certain countries where the foreign currency exposure is different from the functional currency of that country. The Company uses foreign currency forward contracts to hedge a portion of the Company's foreign currency exposures for lead as well as other foreign currency exposures so that gains and losses on these contracts offset changes in the underlying foreign currency denominated exposures. The vast majority of such contracts are for a period not extending beyond one year. As of September 29, 2013 and March 31, 2013, the Company had entered into a total of \$83,443 and \$51,366, respectively, of such contracts.

In the coming twelve months, the Company anticipates that \$2,242 of pretax gain relating to lead and foreign currency forward contracts will be reclassified from accumulated other comprehensive income ("AOCI") as part of cost of goods sold. This amount represents the current unrealized impact of hedging lead and foreign exchange rates, which will change as market rates change in the future, and will ultimately be realized in the income statement as an offset to the corresponding actual changes in lead costs to be realized in connection with the variable lead cost and foreign exchange rates being hedged.

Derivatives not Designated in Hedging Relationships

Interest Rate Swap Agreements

As of March 31, 2013, the Company maintained interest rate swap agreements that converted \$65,000 of variable-rate debt to a fixed-rate basis, utilizing the three-month LIBOR, as a floating rate reference. These agreements expired during the first quarter ended June 30, 2013, and the Company had no interest rate swap agreements as of September 29, 2013. The Company recorded expense relating to changes in the fair value of these agreements in the consolidated condensed statements of income, within other (income) expense, net of \$77 and \$92 during the second quarter and six months ended September 30, 2012, respectively.

Foreign Currency Forward Contracts

The Company also enters into foreign currency forward contracts to economically hedge foreign currency fluctuations on intercompany loans and foreign currency denominated receivables and payables. These are not designated as hedging instruments and changes in fair value of these instruments are recorded directly in the consolidated condensed statements of income. As of September 29, 2013 and March 31, 2013, the notional amount of these contracts was \$16,251 and \$21,749, respectively. The Company recorded income in the consolidated condensed

statements of income within other (income) expense, net of \$46 during the second quarter of fiscal 2014 and expense of \$304 during the second quarter of fiscal 2013, respectively, and expense of \$180 and \$418 for the six months ended September 29, 2013 and September 30, 2012, respectively.

Presented below in tabular form is information on the location and amounts of derivative fair values in the consolidated condensed balance sheets and derivative gains and losses in the consolidated condensed statements of income:

Fair Value of Derivative Instruments September 29, 2013 and March 31, 2013

	Derivatives and	Hedging A Flow	ies Designated as Cash es	Derivatives and Hedging Activities Not Designate Hedging Instruments				
	September 29	, 2013	March 31, 2013	September 29, 2013			March 31, 2013	
Prepaid and other current assets								
Foreign currency forward contracts	\$	_	\$ _	\$	_	\$	241	
Lead hedge forward contracts		1,004	_		_		_	
Other assets								
Lead hedge forward contracts		75	_		_		_	
Total assets	\$	1,079	\$ _	\$	_	\$	241	
Accrued expenses								
Interest rate swap agreements	\$	_	\$ _	\$	_	\$	654	
Lead hedge forward contracts		_	2,832		_		_	
Foreign currency forward contracts		182	252		377		_	
Total liabilities	\$	182	\$ 3,084	\$	377	\$	654	

The Effect of Derivative Instruments on the Consolidated Condensed Statements of Income For the quarter ended September 29, 2013

Derivatives Designated as Cash Flow Hedges	Pretax Gain (Loss) Recognized in AOCI on Derivative (Effective Portion	Location of Gain (Loss) Reclassified from AOCI into Income (Effective Portion)	Pretax Gain (Loss) Reclassified from AOCI into Income (Effective Portion)
Lead hedge contracts	\$ 2,427	Cost of goods sold	\$ (3,661)
Foreign currency forward contracts	(492)	Cost of goods sold	(411)
Total	\$ 1,935		\$ (4,072)

Derivatives Not Designated as Hedging Instruments	Location of Gain (Loss) Recognized in Income on Derivative	Pre	etax Gain (Loss)
Foreign currency forward contracts	Other (income) expense, net	\$	46
Total		\$	46

The Effect of Derivative Instruments on the Consolidated Condensed Statements of Income For the quarter ended September 30, 2012

Derivatives Designated as Cash Flow Hedges	Pretax Gain (Recognized in A Derivative (Effecti	OCÍ on	Location of Gain (Loss) Reclassified from AOCI into Income (Effective Portion)	Pretax Ga Reclassified fr Income (Effe	om AOCÍ into
Lead hedge contracts	\$	12,476	Cost of goods sold	\$	(1,967)
Foreign currency forward contracts		(706)	Cost of goods sold		1,094
Total	\$	11,770		\$	(873)

Derivatives Not Designated as Hedging Instruments	Location of Gain (Loss) Recognized in Income on Derivative	Pretax Gain (Loss)
Interest rate swap contracts	Other (income) expense, net	\$ (77)
Foreign currency forward contracts	Other (income) expense, net	(304)
Total		\$ (381)

The Effect of Derivative Instruments on the Consolidated Condensed Statements of Income For the six months ended September 29, 2013

Derivatives Designated as Cash Flow Hedges	Pretax Gain (Loss Recognized in AOCI Derivative (Effective Po	on Reclassified fron	n AÒCI ínto Recl	Pretax Gain (Loss) lassified from AOCI into come (Effective Portion)
Lead hedge contracts	\$ 2	,124 Cost of goo	ods sold \$	(1,615)
Foreign currency forward contracts	(1	,098) Cost of goo	ods sold	(1,002)
Total	\$ 1	,026	\$	(2,617)

Derivatives Not Designated as Hedging Instruments	Location of Gain (Loss) Recognized in Income on Derivative	Pretax Gain (Loss)
Foreign currency forward contracts	Other (income) expense, net	\$ (180)
Total		\$ (180)

The Effect of Derivative Instruments on the Consolidated Condensed Statements of Income For the six months ended September 30, 2012

Derivatives Designated as Cash Flow Hedges	Pretax Gain (Loss) Recognized in AOCI on Derivative (Effective Portion)	Location of Gain (Loss) Reclassified from AOCI into Income (Effective Portion)	Pretax Gain (Loss) Reclassified from AOCI into Income (Effective Portion)
Lead hedge contracts	\$ 6,063	Cost of goods sold	\$ (469)
Foreign currency forward contracts	1	Cost of goods sold	2,372
Total	\$ 6,064		\$ 1,903

Derivatives Not Designated as Hedging Instruments	Location of Gain (Loss) Recognized in Income on Derivative	Pretax Gain (Loss)
Interest rate swap contracts	Other (income) expense, net	\$ (92)
Foreign currency forward contracts	Other (income) expense, net	(418)
Total		\$ (510)

5. Income Taxes

The Company's income tax provisions consist of federal, state and foreign income taxes. The tax provisions for the second quarters of fiscal 2014 and 2013 were based on the estimated effective tax rates applicable for the full years ending March 31, 2014 and March 31, 2013, respectively, after giving effect to items specifically related to the interim periods.

The effective income tax rates for the second quarters of fiscal 2014 and 2013 were 27.0% and 28.0%, respectively. The effective income tax rates for the first six months of fiscal 2014 and 2013 were 27.4% and 28.6%, respectively. The rate decrease in the second quarter and six months of fiscal 2014 compared to the prior year periods are primarily due to changes in the mix of earnings among tax jurisdictions and a reduction in income taxes related to a legal entity reorganization of certain foreign subsidiaries.

6. Warranties

The Company provides for estimated product warranty expenses when the related products are sold, with related liabilities included within accrued expenses and other liabilities. Because warranty estimates are forecasts that are based on the best available information, primarily historical claims experience, claims costs may differ from amounts provided. An analysis of changes in the liability for product warranties is as follows:

	Quarter ended				Six months ended			
	Septem	ber 29, 2013	Sej	ptember 30, 2012	0, September 29, 2013			
Balance at beginning of period	\$	42,202	\$	42,488	\$	42,591	\$	42,067
Current period provisions		2,302		5,892		7,645		11,488
Costs incurred		(4,477)		(6,083)		(10,230)		(10,578)
Foreign currency translation adjustment		1,497		342		1,518		(338)
Balance at end of period	\$	41,524	\$	42,639	\$	41,524	\$	42,639

7. Commitments, Contingencies and Litigation

Litigation and Other Legal Matters

The Company is involved in litigation incidental to the conduct of its business, the results of which, in the opinion of management, are not likely to be material to the Company's financial position, results of operations, or cash flows. See Note 18 to the Consolidated Financial Statements included in the Company's 2013 Annual Report on Form 10-K. There have been no significant changes since March 31, 2013.

Environmental Issues

As a result of its operations, the Company is subject to various federal, state, and local, as well as international environmental laws and regulations and is exposed to the costs and risks of handling, processing, storing, transporting, and disposing of hazardous substances, especially lead and acid. The Company's operations are also subject to federal, state, local and international occupational safety and health regulations, including laws and regulations relating to exposure to lead in the workplace. See Note 18 to the Consolidated Financial Statements included in the Company's 2013 Annual Report on Form 10-K for a full description of environmental issues. There have been no significant changes since March 31, 2013.

Lead Contracts

To stabilize its costs, the Company has entered into contracts with financial institutions to fix the price of lead. The vast majority of such contracts are for a period not extending beyond one year. Under these contracts, at September 29, 2013 and March 31, 2013, the Company has hedged the price to purchase 94.2 million pounds and 56.3 million pounds of lead, respectively, for a total purchase price of \$88,546 and \$56,601, respectively.

Foreign Currency Forward Contracts

The Company quantifies and monitors its global foreign currency exposures. On a selective basis, the Company will enter into foreign currency forward and option contracts to reduce the volatility from currency movements that affect the Company. The vast majority of such contracts are for a period not extending beyond one year. The Company's largest exposure is from the purchase and conversion of U.S. dollar based lead costs into local currencies in EMEA. Additionally, the Company has currency exposures from intercompany and third party trade transactions. To hedge these exposures, the Company has entered into a total of \$99,694 and \$73,115, respectively, of foreign currency forward contracts with financial institutions as of September 29, 2013 and March 31, 2013.

Interest Rate Swap Agreements

The Company is exposed to changes in variable U.S. interest rates on borrowings under its 2011 Credit Facility. On a selective basis, from time to time, the Company enters into interest rate swap agreements to reduce the negative impact that increases in interest rates could have on its outstanding variable rate debt. These agreements expired during the first quarter ended June 30, 2013 and the Company had no interest rate swap agreements at September 29, 2013 as well as no borrowings under its 2011 Credit Facility. At March 31, 2013, such agreements converted \$65,000 of variable-rate debt to a fixed-rate basis, utilizing the three-month LIBOR, as a floating rate reference. Fluctuations in LIBOR and fixed rates affect both the Company's net financial investment position and the amount of cash to be paid or received under these agreements.

8. Restructuring Plans

During fiscal 2012, the Company announced restructuring plans related to its operations in EMEA, primarily consisting of the transfer of manufacturing of select products between certain of its manufacturing operations and restructuring of its selling, general and administrative operations, which resulted in the reduction of approximately 85 employees upon completion at the end of the second quarter of fiscal 2014. The total charges for these actions amounted to \$3,545, primarily from cash expenses for employee severance-related payments. The Company recorded restructuring charges of \$3,545 through fiscal 2013, with no additional charges during the first six months of fiscal 2014. The Company incurred \$3,346 of costs against the accrual through fiscal 2013, with \$185 additional costs incurred against the accrual during the first six months of fiscal 2014. This plan has been completed as of September 29, 2013.

During fiscal 2013, the Company announced a further restructuring related to improving the efficiency of its manufacturing operations in EMEA. The Company estimates that the total charges for these actions will amount to approximately \$8,100, primarily from cash expenses for employee severance-related payments and non-cash expenses associated with the write-off of certain fixed assets and inventory. The Company estimates that these actions will result in the reduction of approximately 130 employees upon completion. During fiscal 2013, the Company recorded restructuring charges of \$3,998, consisting of non-cash charges of \$1,399 related to the write-off of fixed assets and inventory, along with cash charges related to employee severance and other charges of \$2,599. The Company recorded an additional \$1,387 in charges during the first six months of fiscal 2014. During fiscal 2013, the Company incurred \$952 of costs against the accrual, with an additional \$1,483 of costs incurred during the first six months of fiscal 2014. As of September 29, 2013, the reserve balance associated with these actions is \$1,559. The Company expects to be committed to an additional \$1,324 of restructuring charges in fiscal 2014 related to these actions, and expects to complete the program during fiscal 2015.

During fiscal 2013, the Company announced a restructuring related to the closure of its manufacturing facility located in Chaoan, People's Republic of China, pursuant to which the Company will transfer the manufacturing at that location to its other facilities in the People's Republic of China, to improve operational efficiencies. The Company estimates that the total charges related to this action will amount to

\$3,400. During fiscal 2013, the Company recorded restructuring charges of \$2,691, consisting of non-cash charges of \$2,290 related to the write-off of fixed assets and inventory, along with cash charges related to employee severance and other charges of \$401. During the first six months of fiscal 2014, no additional restructuring charges were accrued. During fiscal 2013, the Company incurred \$221 in costs against the accrual, with an additional \$136 of costs incurred during the first six months of fiscal 2014. As of September 29, 2013, the reserve balance associated with this action is \$44. The Company expects to be committed to an additional \$700 of restructuring related to these actions and expects to complete the restructuring during fiscal 2014.

During fiscal 2014, the Company announced a further restructuring related to improving the efficiency of its manufacturing operations in EMEA. The Company estimates that the total charges for these actions will amount to approximately \$1,300, primarily from cash expenses for employee severance-related payments and contract obligations. The Company estimates that these actions will result in the reduction of approximately 20 employees upon completion. During fiscal 2014, the Company recorded restructuring charges of \$153 consisting of cash charges related to employee severance and incurred \$139 of costs against the accrual. As of September 29, 2013, the reserve balance associated with these actions is \$14. The Company expects to be committed to an additional \$850 of restructuring charges in fiscal 2014 related to these actions, and expects to complete the program during fiscal 2015.

A roll-forward of the restructuring reserve is as follows:

	Employee Severance	Other	Total
Balance as of March 31, 2013	\$ 1,738	\$ 221	\$ 1,959
Accrued	1,540	_	1,540
Costs incurred	(1,786)	(157)	(1,943)
Foreign currency impact and other	56	5	61
Balance as of September 29, 2013	\$ 1,548	\$ 69	\$ 1,617

9. Debt

The following summarizes the Company's long-term debt including capital lease obligations:

	Septe	mber 29, 2013	March 31, 2013		
3.375% Convertible Notes, net of discount, due 2038	\$	159,000	\$	155,273	
Capital lease obligations and other		368		514	
		159,368		155,787	
Less current portion		234		311	
Total long-term debt and capital lease obligations	\$	159,134	\$	155,476	

The Convertible Notes will mature on June 1, 2038. Prior to maturity, the holders may convert their Convertible Notes into shares of the Company's common stock at any time after March 1, 2015 or prior to that date under certain circumstances. When issued, the initial conversion rate was 24.6305 shares per \$1,000 principal amount of Convertible Notes, which was equivalent to an initial conversion price of \$40.60 per share. The conversion price is subject to adjustment under certain circumstances. It is the Company's current intent to settle the principal amount of any conversions in cash, and any additional conversion consideration in cash, shares of EnerSys common stock or a combination of cash and shares. The Convertible Notes will mature on June 1, 2038, unless earlier converted, redeemed or repurchased.

Holders may convert their Convertible Notes prior to March 1, 2015, at the option of the holder, under the following circumstances: (i) during any calendar quarter (and only during such calendar quarter), if the last reported sale price (as defined in the indenture for the Convertible Notes) of a share of EnerSys common stock for 20 or more trading days in a period of 30 consecutive trading days ending on the last trading day of the immediately preceding calendar quarter exceeds 130% of the applicable conversion price in effect for the Convertible Notes on the last trading day of the immediately preceding calendar quarter, (ii) upon the occurrence of specified corporate events, or (iii) during the five business-day period after any five consecutive trading day period (the "measurement period") in which the "trading price" (as defined in the indenture for the Convertible Notes) of the Convertible Notes for each day of the measurement period was less than 98% of the product of the "last reported sale price" (as defined in the indenture for the Convertible Notes) of a share of EnerSys common stock and the applicable conversion rate on such day.

The Convertible Notes are represented by a liability component, which is reported herein as long-term debt, net of discount and an equity component representing the convertible feature, which is included in additional paid-in capital in EnerSys stockholders' equity. See Note 8 to the Consolidated Financial Statements included in the Company's 2013 Annual Report on Form 10-K for early redemption features.

The following represents the principal amount of the liability component, the unamortized discount, and the net carrying amount of the Convertible Notes as of September 29, 2013 and March 31, 2013:

	September 29, 2013			March 31, 2013		
Principal	\$	172,500	\$	172,500		
Unamortized discount		(13,500)		(17,227)		
Net carrying amount	\$	159,000	\$	155,273		
Carrying amount of equity component	\$	29,850	\$	29,850		

As of September 29, 2013, the remaining discount will be amortized over a period of 20 months. The conversion price of the \$172,500 in aggregate principal amount of the Convertible Notes is \$40.60 per share and the number of shares on which the aggregate consideration is to be delivered upon conversion is 4,248,761.

The effective interest rate on the liability component of the Convertible Notes was 8.50%. The amount of interest cost recognized for the amortization of the discount on the liability component of the Convertible Notes was \$1,883 and \$1,731, respectively, during the quarters ended September 29, 2013 and September 30, 2012 and \$3,727 and \$3,427, respectively, during the six months ended September 29, 2013 and September 30, 2012.

Short-Term Debt

As of September 29, 2013 and March 31, 2013, the Company had \$20,676 and \$22,702, respectively, of short-term borrowings from banks. The weighted-average interest rates on these borrowings were approximately 11% and 14%, respectively, at September 29, 2013 and September 30, 2012.

Available Lines of Credit

As of September 29, 2013 and March 31, 2013, the Company had available and undrawn, under all its lines of credit, \$480,502 and \$469,123, respectively. Included in the September 29, 2013 and March 31, 2013 were \$131,752 and \$120,373, respectively, of uncommitted lines of credit.

As of September 29, 2013 and March 31, 2013, the Company had \$6,970 and \$11,854, respectively, of standby letters of credit.

Amendment to 2011 Credit Facility

In August 2013, the Company amended its 2011 Credit Facility. The key amendments to the agreement were extending the maturity until September 2018, increasing flexibility for entering into acquisitions and joint ventures, stock buybacks and cash dividend payments. The Company incurred \$823 in new deferred financing fees in connection with this amendment.

10. Retirement Plans

The following tables present the components of the Company's net periodic benefit cost related to its defined benefit pension plans:

		United St	tates Plans		International Plans				
		Quarte	er ended			Quarter ended			
	Septem	ber 29, 2013	September 30, 2012		September 29, 2013		September 30, 2012		
Service cost	\$	90	\$	81	\$	206	\$	176	
Interest cost		156		163		596		591	
Expected return on plan assets		(198)		(188)		(525)		(465)	
Amortization and deferral		134		101		107		51	
Net periodic benefit cost	\$	182	\$	157	\$	384	\$	353	

		United St	tates Plans		International Plans				
		Six mon			Six months ended				
	Septer	September 29, 2013 September 30, 2			Septer	nber 29, 2013	September 30, 2012		
Service cost	\$	181	\$	162	\$	406	\$	353	
Interest cost		312		327		1,176		1,185	
Expected return on plan assets		(397)		(377)		(1,036)		(929)	
Amortization and deferral		267		201		211		103	
Net periodic benefit cost	\$	363	\$	313	\$	757	\$	712	

11. Stock-Based Compensation

As of September 29, 2013, the Company maintains the EnerSys 2010 Equity Incentive Plan ("2010 EIP"). The 2010 EIP reserved 3,177,477 shares of common stock for the grant of various types of equity awards including nonqualified stock options, restricted stock, restricted stock units, market share units and other forms of stock-based compensation.

The Company recognized stock-based compensation expense associated with its equity incentive plans of \$4,569 for the second quarter of fiscal 2014 and \$3,825 for the second quarter of fiscal 2013. The Company recognized stock-based compensation expense associated with its equity incentive plans of \$7,592 for the six months ended September 29, 2013 and \$7,198 for the six months ended September 30, 2012.

During the six months ended September 29, 2013, the Company granted to non-employee directors 20,148 restricted stock units, pursuant to the EnerSys Deferred Compensation Plan for non-employee directors.

During the six months ended September 29, 2013, the Company granted to management and other key employees 150,629 restricted stock units that vest 25% each year over four years from the date of grant, and 189,438 market share units that vest three years from the date of grant.

Common stock activity during the six months ended September 29, 2013 included the exercise of 5,786 options and the vesting of 216,432 restricted stock units and 222,123 market share units.

 $As of September 29, 2013, there were 72, 200 non-qualified stock options, 556, 847 \ restricted stock units and 668, 208 \ market share units outstanding.$

12. Stockholders' Equity and Noncontrolling Interests

Common Stock

The following demonstrates the change in the number of shares of Common Stock outstanding during the six months ended September 29, 2013:

Shares outstanding as of March 31, 2013	47,840,204
Purchase of treasury stock	(661,292)
Shares issued as part of equity-based compensation plans, net of equity awards surrendered for option price and taxes	287,040
Shares outstanding as of September 29, 2013	47,465,952

Treasury Stock

During the six months ended September 29, 2013, the Company purchased 661,292 shares of its common stock for \$33,593. At September 29, 2013 and March 31, 2013, the Company held 5,791,369 and 5,130,077 shares as treasury stock, respectively.

Accumulated Other Comprehensive Income ("AOCI")

The components of AOCI, net of tax, are as follows:

	March 31, 2013			Before Reclassifications	A	Amounts Reclassified from AOCI	September 29, 2013		
Pension funded status adjustment	\$	(13,169)	\$	_	\$	316	\$	(12,853)	
Net unrealized loss (gain) on derivative instruments		(832)		644		1,646		1,458	
Foreign currency translation adjustment		54,656		23,679		_		78,335	
Accumulated other comprehensive income	\$	40,655	\$	24,323	\$	1,962	\$	66,940	

The following table presents reclassifications from other AOCI during the second quarter ended September 29, 2013:

Components of AOCI	Amounts Re	classified from AOCI	Location of Loss Recognized on Income Statement					
Derivatives in Cash Flow Hedging Relationships:								
Net unrealized loss on derivative instruments	\$	4,072						
Tax benefit		(1,511)						
Net unrealized loss on derivative instruments, net of tax	\$	2,561	Cost of goods sold					
Defined benefit pension costs:								
Prior service costs and deferrals	\$	235						
Tax benefit		(120)						
Net periodic benefit cost, net of tax	\$	115	Net periodic benefit cost, included in Cost of goods sold, Operating expenses - See Note 10					

The following table presents reclassifications from AOCI during the six months ended September 29, 2013:

Components of AOCI	Amounts Reclassifie	ed from AOCI	Location of Loss Recognized on Income Statement
Derivatives in Cash Flow Hedging Relationships:			
Net unrealized loss on derivative instruments	\$	2,617	
Tax benefit		(971)	
Net unrealized loss on derivative instruments, net of tax	\$	1,646	Cost of goods sold
Defined benefit pension costs:			
Prior service costs and deferrals	\$	472	
Tax benefit		(156)	
Net periodic benefit cost, net of tax	\$	316	Net periodic benefit cost, included in Cost of goods sold, Operating expenses - See Note 10

The following demonstrates the change in equity attributable to EnerSys stockholders and nonredeemable noncontrolling interests during the six months ended September 29, 2013:

	quity Attributable to nerSys Stockholders	Nonredeemable Noncontrolling Interests	Total Equity
Balance as of March 31, 2013	\$ 1,169,401	\$ 5,882	\$ 1,175,283
Total comprehensive income:			
Net earnings	82,186	12	82,198
Net unrealized gain on derivative instruments, net of tax	2,290	_	2,290
Pension funded status adjustment, net of tax	(196)	_	(196)
Foreign currency translation adjustment	24,191	14	24,205
Total other comprehensive income, net of tax	26,285	14	26,299
Total comprehensive income	108,471	26	108,497
Other changes in equity:			
Purchase of treasury stock	(33,593)	_	(33,593)
Cash dividends - common stock (\$0.25 per share)	(11,902)	_	(11,902)
Other, including activity related to equity awards	3,960	_	3,960
Balance as of September 29, 2013	\$ 1,236,337	\$ 5,908	\$ 1,242,245

The following demonstrates the change in redeemable noncontrolling interests during the six months ended September 29, 2013:

	I	Redeemable Noncontrolling Interests
Balance as of March 31, 2013	\$	11,095
Net losses		(630)
Foreign currency translation adjustment		(1,026)
Balance as of September 29, 2013	\$	9,439

13. Earnings Per Share

The following table sets forth the reconciliation from basic to diluted weighted-average number of common shares outstanding and the calculations of net earnings per common share attributable to EnerSys stockholders.

		Quarte	r end	led	Six months ended			
	Sept	ember 29, 2013	S	September 30, 2012	S	September 29, 2013		September 30, 2012
Net earnings attributable to EnerSys stockholders	\$	41,339	\$	43,790	\$	82,186	\$	89,594
Weighted-average number of common shares outstanding:								
Basic		47,573,496		48,188,331		47,721,239		48,044,767
Dilutive effect of:								
Common shares from exercise and lapse of equity awards, net of shares assumed reacquired		801,839		531,585		815,674		528,687
Convertible Notes		1,030,483		_		818,468		_
Diluted weighted-average number of common shares outstanding		49,405,818		48,719,916		49,355,381		48,573,454
Basic earnings per common share attributable to EnerSys stockholders	\$	0.87	\$	0.91	\$	1.72	\$	1.86
Diluted earnings per common share attributable to EnerSys stockholders	\$	0.84	\$	0.90	\$	1.67	\$	1.84
Anti-dilutive equity awards not included in diluted weighted-average common shares		11,010		287,566		13,321		197,392

The aggregate number of common shares that the Company could be obligated to issue upon conversion of its Convertible Notes that the Company sold in May 2008 is 4,248,761. It is the Company's current intent to settle the principal amount of any conversions in cash, and any additional conversion consideration in cash, shares of the Company's common stock or a combination of cash and shares. During the second quarter of fiscal 2014, the average price of our common stock at \$53.60 per share exceeded the conversion price of \$40.60 per share on the Convertible Notes. For the current quarter and six months ended September 29, 2013, 1,030,483 and 818,468 shares, respectively, relating to the conversion premium on the Convertible Notes were included in the diluted earnings per share using the treasury stock method.

14. Business Segments

The Company has three reportable business segments based on geographic regions, defined as follows:

- · Americas, which includes North and South America, with segment headquarters in Reading, Pennsylvania, USA;
- · EMEA, which includes Europe, the Middle East and Africa, with segment headquarters in Zurich, Switzerland; and
- Asia, which includes Asia, Australia and Oceania, with segment headquarters in Singapore.

Summarized financial information related to the Company's reportable segments for the quarters and six months ended September 29, 2013 and September 30, 2012, is shown below:

	Quarter ended					Six months ended			
	September 29, 2013		Se	September 30, 2012		September 29, 2013		eptember 30, 2012	
Net sales by segment to unaffiliated customers									
EMEA	\$	223,374	\$	215,453	\$	454,341	\$	452,504	
Americas		287,683		276,704		603,306		565,628	
Asia		57,790		62,055		108,497		129,990	
Total net sales	\$	568,847	\$	554,212	\$	1,166,144	\$	1,148,122	
Net sales by product line									
Reserve power	\$	279,494	\$	285,286	\$	572,313	\$	574,580	
Motive power		289,353		268,926		593,831		573,542	
Total net sales	\$	568,847	\$	554,212	\$	1,166,144	\$	1,148,122	
Intersegment sales			-						
EMEA	\$	18,276	\$	23,222	\$	36,573	\$	45,384	
Americas		8,176		9,143		18,269		19,697	
Asia		10,052		7,785		17,763		14,144	
Total intersegment sales (1)	\$	36,504	\$	40,150	\$	72,605	\$	79,225	
Operating earnings by segment									
EMEA	\$	15,243	\$	14,021	\$	31,326	\$	31,241	
Americas		43,143		43,572		84,868		88,086	
Asia		3,738		6,587		8,959		15,478	
Restructuring charges - EMEA		(1,119)		(1,295)		(1,540)		(1,665)	
Total operating earnings (2)	\$	61,005	\$	62,885	\$	123,613	\$	133,140	

¹⁾ Intersegment sales are presented on a cost-plus basis which takes into consideration the effect of transfer prices between legal entities.

15. Subsequent Events

On October 8, 2013, the Company completed the acquisition of Purcell Systems, Inc., a designer, manufacturer and marketer of thermally managed electronic equipment and battery cabinet enclosures, headquartered in Spokane, WA, from Weston Presidio and other stockholders, representing an investment of approximately \$115,000.

On October 31, 2013, the Company announced the declaration of a quarterly cash dividend of \$0.125 per share of common stock to be paid on December 27, 2013, to stockholders of record as of December 13, 2013.

On November 1, 2013, the Company completed the acquisition of Quallion, LLC, a manufacturer of lithium ion cells and batteries for medical devices, defense, aviation and space, headquartered in Sylmer, California, representing an investment of approximately \$30,000.

⁽²⁾ The Company does not allocate interest expense or other (income) expense to the reportable segments.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS FORWARD-LOOKING STATEMENTS

The Private Securities Litigation Reform Act of 1995 (the "Reform Act") provides a safe harbor for forward-looking statements made by or on behalf of EnerSys. EnerSys and its representatives may, from time to time, make written or verbal forward-looking statements, including statements contained in our filings with the Securities and Exchange Commission and its reports to stockholders. Generally, the inclusion of the words "believe," "expect," "intend," "estimate," "anticipate," "will," and similar expressions identify statements that constitute "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934 and that are intended to come within the safe harbor protection provided by those sections. All statements addressing operating performance, events, or developments that EnerSys expects or anticipates will occur in the future, including statements relating to sales growth, earnings or earnings per share growth, and market share, as well as statements expressing optimism or pessimism about future operating results, are forward-looking statements within the meaning of the Reform Act. The forward-looking statements are and will be based on management's then-current views and assumptions regarding future events and operating performance, and are applicable only as of the dates of such statements.

Forward-looking statements involve risks, uncertainties and assumptions. Although we do not make forward-looking statements unless we believe we have a reasonable basis for doing so, we cannot guarantee their accuracy. Actual results may differ materially from those expressed in these forward-looking statements due to a number of uncertainties and risks, including the risks described in the Company's 2013 Annual Report on Form 10-K and other unforeseen risks. You should not put undue reliance on any forward-looking statements. These statements speak only as of the date of this Quarterly Report on Form 10-Q, even if subsequently made available on our website or otherwise, and we undertake no obligation to update or revise these statements to reflect events or circumstances occurring after the date of this Quarterly Report on Form 10-Q.

Our actual results may differ materially from those contemplated by the forward-looking statements for a number of reasons, including the following factors:

- · general cyclical patterns of the industries in which our customers operate;
- · the extent to which we cannot control our fixed and variable costs;
- · the raw materials in our products may experience significant fluctuations in market price and availability;
- certain raw materials constitute hazardous materials that may give rise to costly environmental and safety claims;
- legislation regarding the restriction of the use of certain hazardous substances in our products;
- risks involved in our operations such as disruption of markets, changes in import and export laws, environmental regulations, currency restrictions and currency exchange rate fluctuations:
- our ability to raise our selling prices to our customers when our product costs increase;
- · the extent to which we are able to efficiently utilize our global manufacturing facilities and optimize our capacity;
- · general economic conditions in the markets in which we operate;
- · competitiveness of the battery markets throughout the world;
- our timely development of competitive new products and product enhancements in a changing environment and the acceptance of such products and product enhancements by customers;
- · our ability to adequately protect our proprietary intellectual property, technology and brand names;
- litigation and regulatory proceedings to which we might be subject;
- changes in our market share in the geographic business segments where we operate;
- our ability to implement our cost reduction initiatives successfully and improve our profitability;
- quality problems associated with our products;
- our ability to implement business strategies, including our acquisition strategy, manufacturing expansion and restructuring plans;
- our acquisition strategy may not be successful in locating advantageous targets;
- our ability to successfully integrate any assets, liabilities, customers, systems and management personnel we acquire into our operations and our ability to realize related revenue synergies and cost savings within expected time frames;
- our debt and debt service requirements which may restrict our operational and financial flexibility, as well as imposing unfavorable interest and financing costs;
- · our ability to maintain our existing credit facilities or obtain satisfactory new credit facilities;
- · adverse changes in our short and long-term debt levels under our credit facilities;
- our exposure to fluctuations in interest rates on our variable-rate debt;
- our ability to attract and retain qualified personnel;

- · our ability to maintain good relations with labor unions;
- credit risk associated with our customers, including risk of insolvency and bankruptcy;
- our ability to successfully recover in the event of a disaster affecting our infrastructure;
- terrorist acts or acts of war, could cause damage or disruption to our operations, our suppliers, channels to market or customers, or could cause costs to increase, or create
 political or economic instability; and
- · the operation, capacity and security of our information systems and infrastructure.

This list of factors that may affect future performance is illustrative, but by no means exhaustive. Accordingly, all forward-looking statements should be evaluated with the understanding of their inherent uncertainty.

In the following discussion and analysis of results of operations and financial condition, certain financial measures may be considered "non-GAAP financial measures" under Securities and Exchange Commission rules. These rules require supplemental explanation and reconciliation, which is provided in this Quarterly Report on Form 10-Q. EnerSys' management uses the non-GAAP measures "primary working capital", "primary working capital percentage" (see definitions in "Liquidity and Capital Resources" below) and capital expenditures in its evaluation of business segment cash flow and financial position performance. These disclosures have limitations as an analytical tool, should not be viewed as a substitute for cash flow determined in accordance with GAAP, and should not be considered in isolation or as a substitute for analysis of the Company's results as reported under GAAP, nor are they necessarily comparable to non-GAAP performance measures that may be presented by other companies. Management believes that this non-GAAP supplemental information is helpful in understanding the Company's ongoing operating results.

Overview

EnerSys (the "Company," "we," or "us") is the world's largest manufacturer, marketer and distributor of industrial batteries. We also manufacture, market and distribute related products such as chargers, power equipment and battery accessories, and we provide related after-market and customer-support services for industrial batteries. We market and sell our products globally to over 10,000 customers in more than 100 countries through a network of distributors, independent representatives and our internal sales force.

We operate and manage our business in three geographic regions of the world—Americas, EMEA and Asia, as described below. Our business is highly decentralized with manufacturing locations throughout the world. More than half of our manufacturing capacity is located outside the United States, and approximately 60% of our net sales were generated outside the United States. The Company has three reportable business segments based on geographic regions, defined as follows:

- Americas, which includes North and South America, with our segment headquarters in Reading, Pennsylvania, USA;
- EMEA, which includes Europe, the Middle East and Africa, with our segment headquarters in Zurich, Switzerland; and
- Asia, which includes Asia, Australia and Oceania, with our segment headquarters in Singapore.

We evaluate business segment performance based primarily upon operating earnings exclusive of highlighted items. Highlighted items are those that the Company deems are not indicative of ongoing operating results, including those charges that the Company incurs as a result of restructuring activities and those charges and credits that are not directly related to ongoing business segment performance. All corporate and centrally incurred costs are allocated to the business segments based principally on net sales. We evaluate business segment cash flow and financial position performance based primarily upon capital expenditures and primary working capital levels (see definition of primary working capital in "Liquidity and Capital Resources" below). Although we monitor the three elements of primary working capital (receivables, inventory and payables), our primary focus is on the total amount due to the significant impact it has on our cash flow.

Our management structure, financial reporting systems, and associated internal controls and procedures, are all consistent with our three geographic business segments. We report on a March 31 fiscal year-end. Our financial results are largely driven by the following factors:

- global economic conditions and general cyclical patterns of the industries in which our customers operate;
- changes in our selling prices and, in periods when our product costs increase, our ability to raise our selling prices to pass such cost increases through to our customers:
- the extent to which we are able to efficiently utilize our global manufacturing facilities and optimize our capacity;
- the extent to which we can control our fixed and variable costs, including those for our raw materials, manufacturing, distribution and operating activities;
- · changes in our level of debt and changes in the variable interest rates under our credit facilities; and
- the size and number of acquisitions and our ability to achieve their intended benefits.

We have two primary industrial battery product lines: reserve power products and motive power products. Net sales classifications by product line are as follows:

- Reserve power products are used for backup power for the continuous operation of critical applications in telecommunications systems, uninterruptible power
 systems, or "UPS" applications for computer and computer-controlled systems, and other specialty power applications, including security systems, premium starting,
 lighting and ignition applications, in switchgear, electrical control systems used in electric utilities, large-scale energy storage, energy pipelines, in commercial
 aircraft, satellites, military aircraft, submarines, ships and tactical vehicles.
- **Motive power products** are used to provide power for manufacturing, warehousing and other material handling equipment, primarily electric industrial forklift trucks, mining equipment, diesel locomotive starting and other rail equipment.

On October 8, 2013, the Company completed the acquisition of Purcell Systems, Inc., a designer, manufacturer and marketer of thermally managed electronic equipment and battery cabinet enclosures, headquartered in Spokane, WA.With the acquisition of Purcell Systems, Inc., we will be supplementing our Reserve Power Products with thermally managed cabinets and enclosures for electronic equipment and batteries.

Economic Climate

Recent indicators continue to suggest a mixed trend in economic activity among the different geographical regions. The Americas and Asia's economic expansion continues but at a slower rate. The ongoing financial crisis and austerity measures in Europe are a factor in slowing overall economic growth in this region.

Volatility of Commodities and Foreign Currencies

Our most significant commodity and foreign currency exposures are related to lead and the euro. Historically, volatility of commodity costs and foreign currency exchange rates have caused large swings in our production costs. As the global economic climate changes, we anticipate that our commodity costs may continue to fluctuate as they have in the past several years. Overall, on a consolidated basis, we have experienced stable trends more recently in our revenue and order rates and commodity cost changes have not been substantial.

Customer Pricing

Our selling prices fluctuated during the last several years to offset the volatile cost of commodities. Approximately 35% of our revenue is currently subject to agreements that adjust pricing to a market-based index for lead. During the current quarter and six months of fiscal 2014, our selling prices increased slightly, compared to the comparable prior year periods.

Liquidity and Capital Resources

Our capital structure and liquidity remain strong. As of September 29, 2013, we had \$272.7 million of cash and cash equivalents and \$481 million, undrawn and available under all our lines of credit including approximately \$132 million of uncommitted credit lines. A substantial majority of the Company's cash and investments are held by foreign subsidiaries and are considered to be indefinitely reinvested and expected to be utilized to fund local operating activities, capital expenditure requirements and acquisitions. The Company believes that it has sufficient sources of domestic and foreign liquidity.

Results of Operations

Net Sales

	Quarter ended September 29, 2013				r ended er 30, 2012	Increase (Decrease)			
Current quarter by segment	In Millions	Percentage of Total Net Sales	In Millions		Percentage of Total Net Sales		In Millions	%	
EMEA	\$ 223.3	39.3%	\$	215.4	38.9%	\$	7.9	3.7%	
Americas	287.7	50.6		276.7	49.9		11.0	4.0	
Asia	57.8	10.1		62.1	11.2		(4.3)	(6.9)	
Total net sales	\$ 568.8	100.0%	\$	5 554.2	100.0%	\$	14.6	2.6%	

	Six months ended September 29, 2013				ths ended er 30, 2012	Increase (Decrease)		
Year to date by segment	In Millions	Percentage of Total Net Sales		In Millions	Percentage of Total Net Sales		In Millions	%
EMEA	\$ 454.3	39.0%	\$	452.5	39.4%	\$	1.8	0.4%
Americas	603.3	51.7		565.6	49.3		37.7	6.7
Asia	108.5	9.3		130.0	11.3		(21.5)	(16.5)
Total net sales	\$ 1,166.1	100.0%	\$	1,148.1	100.0%	\$	18.0	1.6%

Net sales increased \$14.6 million or 2.6% in the second quarter of fiscal 2014 from the comparable period in fiscal 2013. This increase for the quarter was the result of a 1% increase in each of organic volume, pricing and currency translation impact.

Net sales increased \$18.0 million or 1.6% in the six months of fiscal 2014 from the comparable period in fiscal 2013. This increase for the six months was the result of a slight increase in organic volume, price and currency translation impact.

Segment sales

The EMEA segment's net sales increased \$7.9 million or 3.7% in the second quarter of fiscal 2014, as compared to the second quarter of fiscal 2013, primarily due to foreign currency translation impact. Net sales increased \$1.8 million or 0.4% in the six months of fiscal 2014, as compared to the six months of fiscal 2013, primarily due to an increase of approximately 2% related to currency translation impact, and 1% increase in pricing partially offset by a decrease of 2% in organic volume.

The Americas segment's net sales increased \$11.0 million or 4.0% in the second quarter of fiscal 2014, as compared to the second quarter of fiscal 2013, primarily due to an increase of approximately 4% in organic volume and a 1% increase in pricing, partially offset by a 1% decrease due to foreign currency translation impact. Net sales increased \$37.7 million or 6.7% six months of fiscal 2014, as compared to the six months of fiscal 2013, primarily due to higher organic volume which contributed approximately 6% and a 1% increase due to pricing.

The Asia segment's net sales decreased \$4.3 million or 6.9% in the second quarter of fiscal 2014, as compared to the second quarter of fiscal 2013, primarily due to lower organic volume and foreign currency translation impact of approximately 3% each and a 1% decrease in pricing. Net sales decreased \$21.5 million or 16.5% in the six months of fiscal 2014, as compared to the six months of fiscal 2013, primarily due to organic volume decrease of approximately 12% and a decrease of 3% and 1%, due to pricing and foreign currency translation, respectively. The prior periods' sales included a large project in Japan which has concluded.

Product line sales

	Quarter ended September 29, 2013				r ended er 30, 2012	Increase (Decrease)			
	In Millions	Percentage of Total Net Sales		In Millions	Percentage of Total Net Sales		In Millions	%	
Reserve power	\$ 279.5	49.1%	\$	285.3	51.5%	\$	(5.8)	(2.0)%	
Motive power	289.3	50.9		268.9	48.5		20.4	7.6	
Total net sales	\$ 568.8	100.0%	\$	554.2	100.0%	\$	14.6	2.6 %	

	Six months ended September 29, 2013				ths ended er 30, 2012	Increase (Decrease)			
	In Millions	Percentage of Total Net Sales		In Millions	Percentage of Total Net Sales		In Millions	%	
Reserve power	\$ 572.3	49.1%	\$	574.6	50.0%	\$	(2.3)	(0.4)%	
Motive power	593.8	50.9		573.5	50.0		20.3	3.5	
Total net sales	\$ 1,166.1	100.0%	\$	1,148.1	100.0%	\$	18.0	1.6 %	

Net sales of our reserve power products in the second quarter of fiscal 2014 decreased \$5.8 million or 2.0% compared to the second quarter of fiscal 2013. Organic volume decreased approximately 3% partially offset by an increase in pricing of approximately 1%. Net sales in the six months of fiscal 2014 decreased \$2.3 million or 0.4% compared to the six months of fiscal 2013. Organic volume decreased by approximately 2% and was partially offset by both pricing and currency translation impact of approximately 1%.

Net sales of our motive power products in the second quarter of fiscal 2014 increased by \$20.4 million or 7.6% compared to the second quarter of fiscal 2013. An improvement of approximately 6% in organic volume and a 1% increase due to foreign currency translation impact contributed to this improvement in the second quarter of fiscal 2014. Net sales in the six months of fiscal 2014 increased \$20.3 million or

3.5% compared to the six months of fiscal 2013, primarily due to a 2% increase in organic volume and a combined 1% increase due to pricing and foreign currency translation impact.

Gross Profit

	Quarter ended September 29, 2013	Quarter ended September 30, 2012			Increase (Decrease)			
	In Millions	Percentage of Total Net Sales		In Millions	Percentage of Total Net Sales		In Millions	%
Gross Profit	\$ 144.3	25.4%	\$	138.3	25.0%	\$	6.0	4.3%
	 Six montl Septembe	ns ended r 29, 2013		Six month September			Increase (Decrease)
	In Millions	Percentage of Total Net Sales		In Millions	Percentage of Total Net Sales		In Millions	%
Gross Profit	\$ 284.4	24.4%	\$	286.6	25.0%	\$	(2.2)	(0.8)%

Gross profit increased \$6.0 million or 4.3% in the second quarter of fiscal 2014 and decreased \$2.2 million or 0.8% in the six months of fiscal 2014 compared to the comparable periods of fiscal 2013. Gross profit, as a percentage of net sales increased 40 basis points in the second quarter but decreased 60 basis points in the six months of fiscal 2014, when compared to the comparable prior year periods of fiscal 2013. The increase in the current quarter is primarily attributed to higher volume and pricing partially offset by higher commodity costs. The decrease in the six months of fiscal 2014 compared to the prior year six months is primarily due to higher commodity costs partially offset by organic volume and pricing.

Operating Items

	Quarter ended September 29, 2013			Quarte Septemb	ended er 30, 2012	Increase (Decrease)		
	In Millions	Percentage of Total Net Sales		In Millions	Percentage of Total Net Sales		In Millions	%
Operating expenses	\$ 82.2	14.5%	\$	74.1	13.4%	\$	8.1	10.9 %
Restructuring charges	\$ 1.1	0.2%	\$	1.3	0.2%	\$	(0.2)	(13.6)%

		Six months ended September 29, 2013			Six mont Septembe	hs ended er 30, 2012	Increase (Decrease)				
	1	In Millions	Percentage of Total Net Sales		In Millions	Percentage of Total Net Sales		In Millions	%		
Operating expenses	\$	159.3	13.7%	\$	151.8	13.2%	\$	7.5	4.9 %		
Restructuring charges	\$	1.5	0.1%	\$	1.7	0.1%	\$	(0.2)	(7.5)%		

Operating expenses as a percentage of net sales increased 110 and 50 basis points in the second quarter and six months of fiscal 2014 compared to the second quarter and six months of fiscal 2013. Operating expenses, excluding the effect of foreign currency translation, increased \$7.8 million or 10.6% in the second quarter of fiscal 2014 and increased \$8.0 million or 5.3% in the six months of fiscal 2014 compared to the six months of fiscal 2013. Increase in operating expenses in both the quarter and six months is primarily on account of increased sales volume, bad debt expense, acquisition related fees and payroll related expenses. Selling expenses, our main component of operating expenses, were 56.8% and 58.7% of total operating expenses in the second quarter and six months of fiscal 2014, respectively, compared to 59.5% and 60.1% of total operating expenses in the second quarter and six months of fiscal 2013, respectively.

Restructuring charges

Included in our second quarter and six months of fiscal 2014, operating results are \$1.1 million and \$1.5 million of restructuring charges, respectively, primarily for staff reductions in EMEA. Included in our second quarter and six months of fiscal 2013 operating results are \$1.3 million and \$1.7 million of restructuring charges, respectively, primarily for staff reductions and write-off of fixed assets and inventory in EMEA.

Operating Earnings

		r ended er 29, 2013		er ended er 30, 2012	Increase (Decrease)		
Current quarter by segment	In Millions	Percentage of Total Net Sales (1)	In Millions	Percentage of Total Net Sales (1)		In Millions	%
EMEA	\$ 15.2	6.8 %	\$ 14.1	6.5 %	\$	1.1	8.7 %
Americas	43.2	15.0	43.5	15.7		(0.3)	(1.0)
Asia	3.7	6.5	6.6	10.6		(2.9)	(43.3)
Subtotal	62.1	10.9	64.2	11.6		(2.1)	(3.2)
Restructuring charges-EMEA	(1.1)	(0.5)	(1.3)	(0.6)		0.2	(13.6)
Total operating earnings	\$ 61.0	10.7 %	\$ 62.9	11.4 %	\$	(1.9)	(3.0)%

⁽¹⁾ The percentages shown for the segments are computed as a percentage of the applicable segment's net sales.

	Six mont Septembe	hs ended er 29, 2013	Six mont Septembe	ths ended er 30, 2012	Increase (Decrease)			
Year to date by segment	In Millions	Percentage of Total Net Sales (1)	In Millions	Percentage of Total Net Sales (1)	In Millions	%		
EMEA	\$ 31.3	6.9 %	\$ 31.3	6.9 %	\$ 	0.3 %		
Americas	84.9	14.1	88.0	15.6	(3.1)	(3.7)		
Asia	8.9	8.3	15.5	11.9	(6.6)	(42.1)		
Subtotal	125.1	10.7	 134.8	11.7	(9.7)	(7.2)		
Restructuring charges-EMEA	(1.5)	(0.3)	(1.7)	(0.4)	0.2	(7.5)		
Total operating earnings	\$ 123.6	10.6 %	\$ 133.1	11.6 %	\$ (9.5)	(7.2)%		

⁽¹⁾ The percentages shown for the segments are computed as a percentage of the applicable segment's net sales.

Operating earnings decreased \$1.9 million or 3.0% in the second quarter and decreased \$9.5 million or 7.2% in the six months of fiscal 2014 compared to the second quarter and six months of fiscal 2013. Operating earnings as a percentage of net sales, as shown in the table above, decreased 70 basis points in the second quarter of fiscal 2014 and decreased 100 basis points in the six months of fiscal 2014 when compared to the comparable periods of fiscal 2013.

The EMEA segment's operating earnings, excluding restructuring charges discussed above, increased in the second quarter of fiscal 2014 compared to the second quarter of fiscal 2013, with the operating margin increasing 30 basis points to 6.8%. Operating earnings remained flat in the six months of fiscal 2014 in comparison to the comparable period in the prior year. This flat trend in EMEA earnings primarily reflects the current economic environment in this region.

The Americas segment had a slight decrease in operating earnings in the second quarter of fiscal 2014 compared to the second quarter of fiscal 2013, with the operating margin decreasing 70 basis points to 15.0%. Operating earnings decreased in the six months of fiscal 2014 in comparison to the comparable period in the prior year, with the operating margin decreasing 150 basis points to 14.1%. This decrease in both the quarter and six months of fiscal 2014 is primarily attributable to higher commodity costs and operating expenses and a less favorable product mix.

Operating earnings decreased in the Asia segment in the second quarter and six months of fiscal 2014 compared to the prior year periods, with the operating margin decreasing by 410 basis points and 360 basis points, respectively, in the current quarter and six months to 6.5% and 8.3%, respectively. The decrease in operating margin was primarily on account of a less favorable product mix and increased bad debt expense in the current quarter and six months of fiscal 2014 compared to the comparable prior year periods.

Interest Expense

			r ended er 29, 2013			r ended er 30, 2012		Increase	(Decrease)
		In Millions	Percentage of Total Net Sales		In Millions	Percentage of Total Net Sales		In Millions	%
Interest expense	\$	4.1	0.7%	\$	5.0	0.9%	\$	(0.9)	(16.7)%
			ths ended er 29, 2013			ths ended er 30, 2012		Increase	(Decrease)
		In Millions	Percentage of Total Net Sales		In Millions	Percentage of Total Net Sales		In Millions	%
Interest expense	ď	8.4	0.7%	ф	9.7	0.8%	ф	(1.3)	(13.3)%

Interest expense of \$4.1 million in the second quarter of fiscal 2014 (net of interest income of \$0.1 million) was \$0.9 million lower than the interest expense of \$5.0 million in the second quarter of fiscal 2013 (net of interest income of \$0.2 million). Interest expense of \$8.4 million in the six months of fiscal 2014 (net of interest income of \$0.4 million) was \$1.3 million lower than the interest expense of \$9.7 million in the six months of fiscal 2013 (net of interest income of \$0.4 million).

The decrease in interest expense in the second quarter and six months of fiscal 2014 compared to the comparable prior year periods is primarily due to lower average debt outstanding.

Included in interest expense are non-cash charges for deferred financing fees of \$0.3 million and \$0.6 million, respectively, in the second quarter and six months of fiscal 2014 and \$0.3 million and \$0.6 million, respectively, in the second quarter and six months of fiscal 2013.

Included in interest expense is non-cash, accreted interest on the Convertible Notes of \$1.9 million and \$3.7 million, respectively, in the second quarter and six months of fiscal 2014 and \$1.7 million and \$3.4 million, respectively, in the second quarter and six months of fiscal 2013 (see Note 9 to the Consolidated Condensed Financial Statements).

Our average debt outstanding (reflecting the reduction of the Convertible Notes discount) was \$179.0 million and \$178.2 million in the second quarter and six months of fiscal 2014, respectively, compared to \$262.3 million and \$262.8 million in the second quarter and six months of fiscal 2013. This decrease was mainly due to the repayment of outstanding debt under our 2011 Credit Facility and term loans in Asia. The average Convertible Notes discount excluded from our average debt outstanding was \$14.5 million and \$15.4 million in the second quarter and six months of fiscal 2014 and \$21.7 million and \$22.6 million, respectively, in the second quarter of fiscal 2013.

Other (Income) Expense, Net

		r ended er 29, 2013		er ended ber 30, 2012	 Increase (Decrease)			
	In Millions	Percentage of Total Net Sales	In Millions	Percentage of Total Net Sales	In Millions	%		
Other (income) expense, net	\$ 0.5	0.1%	\$ (1.8)	(0.3)%	\$ 2.3	NM		

			ths ended er 29, 2013	Six montl Septembe	hs ended er 30, 2012	Increase (Decrease)			
	M	In Iillions	Percentage of Total Net Sales	In Millions	Percentage of Total Net Sales	N	In Millions	%	
Other (income) expense, net	\$	2.8	0.3%	\$ (0.6)	—%	\$	3.4	NM	

NM = not meaningful

Other (income) expense, net in the second quarter of fiscal 2014 was an expense of \$0.5 million compared to an income of \$1.8 million in the second quarter of fiscal 2013. Other (income) expense, net in the six months of fiscal 2014 was an expense of \$2.8 million compared to an income of \$0.6 million in the six months of fiscal 2013. The unfavorable impact in the second quarter and six months of fiscal 2014 is mainly attributable to higher foreign currency losses of \$0.8 million and \$2.6 million, respectively, in the current quarter and six months compared to \$0.4 million and \$1.2 million, respectively in the comparable prior year periods. The prior year second quarter and six months also had insurance recoveries of \$1.8 million in each of the periods.

Earnings Before Income Taxes

	Quarter Septembe		Quarter Septembe	ended r 30, 2012	Increase (l	Decrease)
	 In Millions	Percentage of Total Net Sales	 In Millions	Percentage of Total Net Sales	In Millions	%
Earnings before income taxes	\$ 56.4	9.9%	\$ 59.7	10.8%	\$ (3.3)	(5.6)%

		Six mont Septembe	hs ended er 29, 2013			ths ended er 30, 2012	 Increase (Decrease)
	М	In illions	Percentage of Total Net Sales	1	In Millions	Percentage of Total Net Sales	In Millions	%
Earnings before income taxes	\$	112.4	9.6%	\$	124.0	10.8%	\$ (11.6)	(9.4)%

As a result of the above, earnings before income taxes in the second quarter of fiscal 2014 decreased \$3.3 million or 5.6% compared to the second quarter of fiscal 2013 and earnings before income taxes in the six months of fiscal 2014 decreased \$11.6 million or 9.4% compared to the six months of fiscal 2013. Earnings before income taxes as a percentage of net sales were 9.9% for the second quarter of fiscal 2014 compared to 10.8% in the second quarter of fiscal 2013 and 9.6% for the six months of fiscal 2014 compared to 10.8% for the six months of fiscal 2013.

Income Tax Expense

	Quarter er September 2		Quarter ended September 30, 2012				Decrease)	
	In Millions	Percentage of Total Net Sales		In Millions	Percentage of Total Net Sales		In Millions	%
Income tax expense	\$ 15.2	2.7%	\$	16.7	3.0%	\$	(1.5)	(9.0)%
Effective tax rate	 27.0%			28.0	0%	(1.0)%		

	Six months ended September 29, 2013				Six month September		Increase (Decrease)			
	N	In Aillions	Percentage of Total Net Sales		In Millions	Percentage of Total Net Sales	1	In Millions	%	
Income tax expense	\$	30.8	2.6%	\$	35.4	3.1%	\$	(4.6)	(13.1)%	
Effective tax rate	27.4%			28.6%				(1.2)%		

The Company's income tax provisions consist of federal, state and foreign income taxes. The tax provisions for the second quarters of fiscal 2014 and fiscal 2013 were based on the estimated effective tax rates applicable for the full years ending March 31, 2014 and March 31, 2013, respectively, after giving effect to items specifically related to the interim periods.

The effective income tax rates for the second quarters of fiscal 2014 and fiscal 2013 were 27.0% and 28.0%, respectively. The effective income tax rates for the six months of fiscal 2014 and fiscal 2013 were 27.4% and 28.6%, respectively. The rate decrease in the second quarter and six months of fiscal 2014 as compared to the comparable prior year periods are primarily due to changes in the mix of earnings among tax jurisdictions and a reduction in income taxes related to a legal entity reorganization of certain foreign subsidiaries.

Critical Accounting Policies and Estimates

There have been no material changes to our critical accounting policies from those discussed under the caption "Critical Accounting Policies and Estimates" in "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations" in our 2013 Annual Report on Form 10-K.

Liquidity and Capital Resources

Operating activities provided cash of \$91.2 million in the six months of fiscal 2014 compared to \$78.8 million in the comparable period of fiscal 2013. In the six months of fiscal 2014, net earnings of \$81.6 million and depreciation and amortization of \$25.3 million were partially offset by cash used for the increase in primary working capital of \$30.3 million, net of currency translation changes. In the first six months of fiscal 2013, net earnings of \$88.6 million and depreciation and amortization of \$25.1 million were offset by cash used for the increase in primary working capital of \$41.1 million, net of currency translation changes.

Primary working capital for this purpose is trade accounts receivable, plus inventories, minus trade accounts payable. The resulting net amount is divided by the trailing three month net sales (annualized) to derive a primary working capital percentage. Primary working capital was \$586.3 million (yielding a primary working capital percentage of 25.8%) at September 29, 2013, \$552.7 million (yielding a primary working capital percentage of 24.2%) at March 31, 2013 and \$607.2 million at September 30, 2012 (yielding a primary working capital percentage of 25.8% at September 29, 2013 is 160 basis points higher than that for March 31, 2013, and 160 basis points lower than that for the prior year period.

Primary working capital increased during the six months of fiscal 2014, largely due to an increase in receivables and a decrease in payables.

Primary working capital and primary working capital percentages at September 29, 2013, March 31, 2013 and September 30, 2012 are computed as follows:

		(1n Millior	is)				
Balance At	Trade eivables	Inventory		Accounts Payable	Total	Quarter Revenue Annualized	Primary Working Capital %
September 29, 2013	\$ 467.7	\$ 352.1	\$	(233.5)	\$ 586.3	\$ 2,275.4	25.8%
March 31, 2013	448.1	353.9		(249.3)	552.7	2,288.5	24.2
September 30, 2012	472.8	366.5		(232.1)	607.2	2,216.8	27.4

Investing activities used cash of \$24.3 million in the six months of fiscal 2014, primarily comprised of capital expenditures, compared to \$26.6 million used in the comparable period in fiscal 2013.

Financing activities used cash of \$49.4 million in the six months of fiscal 2014 primarily due to the repurchase of our common stock for \$33.6 million and payment of a cash dividend to our stockholders of \$11.9 million. Taxes paid related to net share settlement of equity awards, net of option proceeds and related tax benefits resulted in a net outflow of \$3.3 million. Net borrowings on short-term debt were \$0.4 million. In the six months of fiscal 2013, financing activities provided cash of \$2.9 million, primarily reflecting borrowings and repayments of \$173.3 million and \$182.8 million, respectively, on our revolver. Borrowings on long-term and short-term debt were \$5.6 million and \$5.4 million, respectively, which were partially offset by repayments of long-term debt of \$10.6 million in Asia. Exercise of stock options and the related tax benefits contributed \$13.9 million.

As a result of the above, total cash and cash equivalents increased by \$23.3 million to \$272.7 million in the six months of fiscal 2014 compared to an increase of \$51.9 million to \$212.3 million in the comparable period of fiscal 2013.

All obligations under our 2011 Credit Facility are secured by, among other things, substantially all of our U.S. assets. This credit agreement contains various covenants which, absent prepayment in full of the indebtedness and other obligations, or the receipt of waivers, limit our ability to conduct certain specified business transactions, buy or sell assets out of the ordinary course of business, engage in sale and leaseback transactions, pay dividends and take certain other actions. There are no prepayment penalties on loans under this credit facility.

We are in compliance with all covenants and conditions under our credit agreement. Since we believe that we will continue to comply with these covenants and conditions, we believe that we have the financial resources and the capital available to fund the foreseeable organic growth in our business and to remain active in pursuing further acquisition opportunities. See Note 8 to the Consolidated Financial Statements included in our 2013 Annual Report on Form 10-K for a detailed description of debt.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Market Risks

Our cash flows and earnings are subject to fluctuations resulting from changes in interest rates, foreign currency exchange rates and raw material costs. We manage our exposure to these market risks through internally established policies and procedures and, when deemed appropriate, through the use of derivative financial instruments. Our policy does not allow speculation in derivative instruments for profit or execution of derivative instrument contracts for which there are no underlying exposures. We do not use financial instruments for trading purposes and are not a party to any leveraged derivatives. We monitor our underlying market risk exposures on an ongoing basis and believe that we can modify or adapt our hedging strategies as needed.

Counterparty Risks

We have entered into interest rate swap agreements, lead forward purchase contracts and foreign exchange forward contracts to manage the risk associated with our exposures. The Company's agreements are with creditworthy financial institutions. Those contracts that result in a liability position at September 29, 2013 are \$1.9 million (pretax), therefore, there is no risk of nonperformance by these counterparties. Those contracts that result in an asset position at September 29, 2013 are \$2.4 million (pre-tax) and the vast majority of these will settle within one year. The impact on the Company due to nonperformance by the counterparties has been evaluated and not deemed material.

Interest Rate Risks

We are exposed to changes in variable U.S. interest rates on borrowings under our 2011 Credit Facility. On a selective basis, from time to time, we enter into interest rate swap agreements to reduce the negative impact that increases in interest rates could have on our outstanding variable rate debt. We had no interest rate swap agreements as of September 29, 2013.

At March 31, 2013, the aggregate notional amount of interest rate swap agreements was \$65.0 million. These agreements expired in May 2013.

A 100 basis point increase in interest rates would have increased annual interest expense by approximately \$0.2 million on the variable rate portions of our debt.

Commodity Cost Risks - Lead Contracts

We have a significant risk in our exposure to certain raw materials. Our largest single raw material cost is for lead, for which the cost remains volatile. In order to hedge against increases in our lead cost, we have entered into contracts with financial institutions to fix the price of lead. A vast majority of such contracts are for a period not extending beyond one year. We had the following contracts outstanding at the dates shown below:

Date	\$'s Under Contract (in millions)	# Pounds Purchased (in millions)	Average Cost/Pound	Approximate % of Lead Requirements (1)
September 29, 2013	\$ 88.5	94.2	\$ 0.94	19%
March 31, 2013	56.6	56.3	1.00	12
September 30, 2012	40.7	46.8	0.87	10

⁽¹⁾ Based on approximate annual lead requirements for the periods then ended.

For the remaining two quarters of this fiscal year, we believe approximately 58% of the cost of our lead requirement is known. This takes into account the hedge contracts in place at September 29, 2013, lead purchased by September 29, 2013 that will be reflected in future costs under our FIFO accounting treatment, and the benefit from our lead tolling program.

We estimate that a 10% increase in our cost of lead would have increased our cost of goods sold by approximately \$4 million and \$15 million in the second quarter and six months of fiscal 2014.

Foreign Currency Exchange Rate Risks

We manufacture and assemble our products globally in the Americas, EMEA and Asia. Approximately 60% of our sales and expenses are transacted in foreign currencies. Our sales revenue, production costs, profit margins and competitive position are affected by the strength of the currencies in countries where we manufacture or purchase goods relative to the strength of the currencies in countries where our products are sold. Additionally, as we report our financial statements in U.S. dollars, our financial results are affected by the strength of the currencies in countries where we have operations relative to the strength of the U.S. dollar. The principal foreign currencies in which we conduct business are the Euro, Swiss franc, British pound, Polish zloty, Chinese renminbi and Mexican peso.

We quantify and monitor our global foreign currency exposures. Our largest foreign currency exposure is from the purchase and conversion of U.S. dollar based lead costs into local currencies in Europe. Additionally, we have currency exposures from intercompany financing and trade transactions. On a selective basis, we enter into foreign currency forward contracts and option contracts to reduce the impact from the volatility of currency movements; however, we cannot be certain that foreign currency fluctuations will not impact our operations in the future.

To hedge these exposures, we have entered into forward contracts with financial institutions to fix the value at which we will buy or sell certain currencies. The vast majority of such contracts are for a period not extending beyond one year. Forward contracts outstanding as of September 29, 2013 and March 31, 2013 were \$99.7 million and \$73.1 million, respectively. The details of contracts outstanding as of September 29, 2013 were as follows:

Transactions Hedged	\$US juivalent millions)	Avera Rat Hedg	e	Approximate % of Annual Requirements (1)	
Sell Euros for U.S. dollars	\$ 33.5	\$/€	1.32		18%
Sell Euros for Polish zloty	31.6	PLN/€	4.28		29
Sell Euros for British pounds	22.0	£/€	0.84		28
Sell Euros for Chinese Renminbi	5.1	RMB/€	8.13		50
Sell Australian dollars for U.S. dollars	2.4	\$/AUD	0.93		19
Sell U.S. dollars for Mexican pesos	2.5	MXN/\$	12.21		50
Sell Australian dollars for Euros	1.5	€/AUD	1.46		8
Other	1.1				
Total	\$ 99.7				

⁽¹⁾ Based on the fiscal year currency requirements.

Foreign exchange translation adjustments are recorded in the consolidated condensed statements of comprehensive income.

Based on changes in the timing and amount of interest rate and foreign currency exchange rate movements and our actual exposures and hedges, actual gains and losses in the future may differ from our historical results.

ITEM 4. CONTROLS AND PROCEDURES

(a) Disclosure Controls and Procedures. Our management, with the participation of our Chief Executive Officer and Chief Financial Officer, has evaluated the effectiveness of our disclosure controls and procedures (as such term is defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended (the "Exchange Act")) as of the end of the period covered by this report. Based on such evaluation, our Chief Executive Officer and Chief Financial Officer have concluded that, as of the end of such period, our disclosure controls and procedures are effective.

(b) Internal Control Over Financial Reporting. There have not been any changes in our internal control over financial reporting (as such term is defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) during the quarter to which this report relates that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II - OTHER INFORMATION

Item 1. Legal Proceedings

From time to time, we are involved in litigation incidental to the conduct of our business. We do not expect that any of this litigation, individually or in the aggregate, will have a material adverse effect on our financial condition, results of operations or cash flow.

Item 1A. Risk Factors

In addition to the other information set forth in this Form 10-Q, you should carefully consider the factors discussed in Part I, Item 1A. Risk Factors in our Annual Report on Form 10-K for the year ended March 31, 2013, which could materially affect our business, financial condition or future results.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

Purchases of Equity Securities by the Issuer and Affiliated Purchasers

The following table summarizes the number of shares of common stock we purchased from participants in our equity incentive plans as well as repurchases of common stock authorized by the Board of Directors. As provided by our equity incentive plans, vested options outstanding may be exercised through surrender to the Company of option shares or vested options outstanding under the Plan to satisfy the applicable aggregate exercise price (and any withholding tax) required to be paid upon such exercise.

Purchases of Equity Securities

Period	(a) Total number of shares (or units) purchased	(b) Average price paid per share (or unit)	(c) Total number of shares (or units) purchased as part of publicly announced plans or programs	(d) Maximum number (or approximate dollar value) of shares (or units) that may be purchased under the plans or programs (1) (2)
July 1 – July 28, 2013	65,279	\$ 51.21	65,279	\$ 58,362,211
July 29 – August 25, 2013	72,705	52.85	72,705	54,519,444
August 26 – September 29, 2013	81,720	54.01	81,720	50,105,433
Total	219,704	\$ 52.80	219,704	

⁽¹⁾ The Company's Board of Directors has authorized the Company to repurchase up to such number of shares as shall equal the dilutive effects of any equity-based award granted during such fiscal year under the 2010 Equity Incentive Plan and the number of shares exercised through stock option awards during such fiscal year. This repurchase program has been exhausted for fiscal year 2014.

⁽²⁾ The Company's Board of Directors authorized the Company to repurchase up to \$65 million of its common stock. This authorization expires on March 31, 2014.

Item 4. Mine Safety Disclosures

Not applicable.

ITEM 6. EXHIBITS

Exhibit Number	Description of Exhibit
2.1	Amended and Restated Agreement and Plan of Merger, dated as of September 30, 2013, by and among EnerSys Capital Inc., ECI Merger Corporation, Purcell Systems, Inc. and Fortis Advisors LLC, as representative and with respect to Article VII thereof only (the schedules and exhibits have been omitted pursuant to Item 601(b)(2) of Regulation S-K).
3.1	Fifth Restated Certificate of Incorporation (incorporated by reference to Exhibit 3.1 to Amendment No. 3 to EnerSys' Registration Statement on Form S-1 (File No. 333-115553) filed on July 13, 2004).
3.2	Amended and Restated Bylaws (incorporated by reference to Exhibit 3.2 to EnerSys' Quarterly Report on Form 10-Q (File No. 001-32253) filed on August 8, 2012).
10.1	Credit Agreement, dated as of March 29, 2011, among EnerSys, Bank of America, N.A., as Administrative Agent, Wells Fargo Bank, National Association, as Syndication Agent, RB International Finance (USA) LLC and PNC Bank, National Association, as Co-Documentation Agents and Co-Managers and the various lending institutions party thereto (incorporated by reference to Exhibit 10.1 to EnerSys' Current Report on Form 8-K (File No. 001-32253) filed on March 29, 2011).
10.2	Amendment to the Credit Facility, dated as of August 2, 2013, among EnerSys, Bank of America, N.A., as Administrative Agent, Wells Fargo Bank, National Association, as Syndication Agent, RB International Finance (USA) LLC and PNC Bank, National Association, as Co-Documentation Agents and Co-Managers, and the various lending institutions (incorporated by reference to Exhibit 10.2 to EnerSys' Current Report on Form 8-K (File No. 001-32253) filed on August 6, 2013).
31.1	Certification of Principal Executive Officer pursuant to Rule 13a-14(a)/15d-14(a) Under the Securities Exchange Act of 1934 (filed herewith).
31.2	Certification of Principal Financial Officer pursuant to Rule 13a-14(a)/15d-14(a) Under the Securities Exchange Act of 1934 (filed herewith).
32.1	Certification of the Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (furnished herewith).
101.INS	XBRL Instance Document
101.SCH	XBRL Taxonomy Extension Schema Document
101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document
101.DEF	XBRL Taxonomy Extension Definition Linkbase Document
101.LAB	XBRL Taxonomy Extension Label Linkbase Document
101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

ENERSYS (Registrant)

By /s/ Michael J. Schmidtlein

Michael J. Schmidtlein

Senior Vice President Finance & Chief Financial Officer

Date: November 6, 2013

EnerSys

EXHIBIT INDEX

Exhibit Number	Description of Exhibit
2.1	Amended and Restated Agreement and Plan of Merger, dated as of September 30, 2013, by and among EnerSys Capital Inc., ECI Merger Corporation, Purcell Systems, Inc. and Fortis Advisors LLC, as representative and with respect to Article VII thereof only (the schedules and exhibits have been omitted pursuant to Item 601(b)(2) of Regulation S-K).
3.1	Fifth Restated Certificate of Incorporation (incorporated by reference to Exhibit 3.1 to Amendment No. 3 to EnerSys' Registration Statement on Form S-1 (File No. 333-115553) filed on July 13, 2004).
3.2	Amended and Restated Bylaws (incorporated by reference to Exhibit 3.2 to EnerSys' Quarterly Report on Form 10-Q (File No. 001-32253) filed on August 8, 2012).
10.1	Credit Agreement, dated as of March 29, 2011, among EnerSys, Bank of America, N.A., as Administrative Agent, Wells Fargo Bank, National Association, as Syndication Agent, RB International Finance (USA) LLC and PNC Bank, National Association, as Co-Documentation Agents and Co-Managers and the various lending institutions party thereto (incorporated by reference to Exhibit 10.1 to EnerSys' Current Report on Form 8-K (File No. 001-32253) filed on March 29, 2011).
10.2	Amendment to the Credit Facility, dated as of August 2, 2013, among EnerSys, Bank of America, N.A., as Administrative Agent, Wells Fargo Bank, National Association, as Syndication Agent, RB International Finance (USA) LLC and PNC Bank, National Association, as Co-Documentation Agents and Co-Managers, and the various lending institutions (incorporated by reference to Exhibit 10.2 to EnerSys' Current Report on Form 8-K (File No. 001-32253) filed on August 6, 2013).
31.1	Certification of Principal Executive Officer pursuant to Rule 13a-14(a)/15d-14(a) Under the Securities Exchange Act of 1934 (filed herewith).
31.2	Certification of Principal Financial Officer pursuant to Rule 13a-14(a)/15d-14(a) Under the Securities Exchange Act of 1934 (filed herewith).
32.1	Certification of the Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (furnished herewith).
101.INS	XBRL Instance Document
101.SCH	XBRL Taxonomy Extension Schema Document
101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document
101.DEF	XBRL Taxonomy Extension Definition Linkbase Document
101.LAB	XBRL Taxonomy Extension Label Linkbase Document
101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document

Exhibit 2.1

AMENDED AND RESTATED AGREEMENT AND PLAN OF MERGER

BY AND AMONG

ENERSYS CAPITAL INC.

ECI MERGER CORPORATION

AND

PURCELL SYSTEMS, INC.

AND WITH RESPECT TO ARTICLE VII ONLY

FORTIS ADVISORS LLC

AS REPRESENTATIVE

Dated as of September 30, 2013

TABLE OF CONTENTS

	Page
Article I THE MERGER	2
1.1 The Merger	2
1.2 Effective Time; Closing	2
1.3 Effect of the Merger	2
1.4 Certificate of Incorporation and Bylaws	2
1.5 Directors and Officers	3
1.6 Effect on Company Securities	3
1.7 Surrender of Certificates	6
1.8 No Further Ownership Rights in Capital Stock	8
1.9 Lost, Stolen or Destroyed Certificates	8
1.10 Further Action	8
Article II REPRESENTATIONS AND WARRANTIES OF THE COMPANY	9
2.1 Organization; Standing and Power; Charter Documents; Subsidiaries	9
2.2 Capital Structure	9
2.3 Authority; Non-Contravention; Necessary Consents	10
2.4 Financial Statements	11
2.5 Undisclosed Liabilities	12
2.6 Absence of Certain Changes or Events	12
2.7 Taxes	13
2.8 Intellectual Property	15
2.9 Compliance; Permits	17
2.10 Litigation	17
2.11 Brokers' and Finders' Fees	17
2.12 Employee Benefit Plans	17
2.13 Real Property	19
2.14 Assets	19
2.15 Environmental Matters	19
2.16 Contracts	20
2.17 Insurance	22
2.18 Order Backlog	22
2.19 Warranty Claims	22
2.20 Certain Business Practices	23
2.21 Export Control Laws	23
2.22 Books and Records	23

23

2.23 Bank Accounts; Powers of Attorney

2.24 Material Customers and Material Suppliers	24
2.25 Receivables; Inventory	24
2.26 Disclosure	24
Article III REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB	25
3.1 Organization; Standing and Power; Charter Documents	25
3.2 Merger Sub	25
3.3 Authority; Non-Contravention; Necessary Consents	25
3.4 Compliance	26
3.5 Availability of Funds	26
Article IV CONDUCT BY THE COMPANY PRIOR TO THE EFFECTIVE TIME	26
4.1 Conduct of Business by the Company and its Subsidiaries	26
4.2 Supplementation and Correction and Information	28
Article V ADDITIONAL AGREEMENTS	28
5.1 Required Stockholder Approval and Notification	28
5.2 Acquisition Proposals	29
5.3 Confidentiality; Access to Information; No Modification of Representations, Warranties or Covenants	29
5.4 Public Disclosure	30
5.5 Reasonable Efforts	30
5.6 Regulatory Approvals	30
5.7 Employee Matters	31
5.8 Indemnification	31
5.9 Certain Tax Matters	33
5.10 Fees and Expenses	33
5.11 FIRPTA Compliance	33
5.12 Third Party Consents	33
5.13 Merger Sub Compliance	33
5.14 Tax Matters	33
Article VI CONDITIONS TO THE MERGER	34
6.1 Conditions to the Obligations of Each Party to Effect the Merger	34
6.2 Additional Conditions to the Obligations of the Company	34
6.3 Additional Conditions to the Obligations of Parent and Merger Sub	35
Article VII SURVIVAL OF REPRESENTATIONS AND WARRANTIES; INDEMNITY ESCROW	36

36

7.1 Survival of Representations and Warranties	36
7.2 Escrow Amount Deposit; Representative Escrow Amount Deposit	37
7.3 Indemnification From Escrow Fund and Working Capital Fund; Representative Fund	37
7.4 Exclusive Remedy	39
7.5 Distribution of Escrow Fund, Working Capital Fund and Representative Fund	39
7.6 Representative	40
7.7 Third Party Claims	42
7.8 Warranty Claims	43
Article VIII TERMINATION, AMENDMENT AND WAIVER	43
8.1 Termination	43
8.2 Notice of Termination; Effect of Termination	44
8.3 Amendment	5
8.4 Extension; Waiver	45
Article IX GENERAL PROVISIONS	45
9.1 Notices	45
9.2 Definitions	46
9.3 Interpretation	57
9.4 Counterparts	58
9.5 Entire Agreement; Third-Party Beneficiaries	58
9.6 Severability	59
9.7 Other Remedies	59
9.8 Governing Law	59
9.9 Consent to Jurisdiction	59
9.10 Waiver of Jury Trial	59
9.11 Rules of Construction	59
9.12 Assignment	60
9.13 Expenses	60
9.14 No Personal Liability	60
9.15 Guarantee	60

60

9.16 Effectiveness of Amendment and Restatement

EXHIBITS

Exhibit A Form of Written Consent

Exhibit B Form of Escrow Agreement

Exhibit C Form of Agreement to Not Compete

Exhibit D Form of Certificate of Merger

Exhibit E Form of Letter of Transmittal

AMENDED AND RESTATED AGREEMENT AND PLAN OF MERGER

This AMENDED AND RESTATED AGREEMENT AND PLAN OF MERGER (this "Agreement") is made and entered into as of September 30, 2013, by and among EnerSys Capital Inc., a Delaware corporation ("Parent"), ECI Merger Corporation, a Delaware corporation and direct wholly-owned subsidiary of Parent ("Merger Sub"), and Purcell Systems, Inc., a Delaware corporation (the "Company"), and with respect to Article VII only, Fortis Advisors LLC, a Delaware limited liability company, as Representative (the "Representative"). Capitalized terms used herein shall have the meaning set forth in Section 9.2.

RECITALS

- A. Parent, Merger Sub, the Company and the Representative have previously entered into an Agreement and Plan of Merger dated as of September 15, 2013 (the "*Prior Agreement*").
- B. The parties to the Prior Agreement believe it is in the best interests of all parties to effect certain changes to the terms of the Prior Agreement by amending and restating the Prior Agreement as set forth in this Agreement, effective as of the date of this Agreement.
- C. The respective Boards of Directors of Parent, Merger Sub and the Company have deemed it advisable and in the best interests of their respective corporations and stockholders that Parent and the Company consummate the business combination and other transactions provided for herein.
- D. The respective Boards of Directors of Parent, Merger Sub and the Company have approved, in accordance with applicable provisions of the laws of the state of Delaware, including the DGCL ("*Delaware Law*"), this Agreement and the transactions contemplated hereby, including the Merger.
- E. As a condition and inducement to Parent's willingness to enter into this Agreement and to complete the Merger and the other transactions contemplated hereby, the Company has secured and delivered to Parent a written consent irrevocably (1) adopting the Prior Agreement and (2) approving the consummation of the transactions contemplated by the Prior Agreement, including, but not limited to, the Merger, in substantially the form attached hereto as **Exhibit A** (the "Written Consent"), from the holders of (a) at least seventy seven percent (77%) of the voting power of the outstanding shares of Capital Stock, voting together as a single class on an as converted into Common Stock basis, and (b) ninety percent (90%) of the voting power of the outstanding shares of Preferred Stock, voting together as a single class on an as converted into Common Stock basis (the "Required Stockholder Approvals").
- F. Concurrently with, and as a condition to, the Closing, Parent, the Company, the Representative and the Escrow Agent shall enter into an escrow agreement substantially in the form attached hereto as **Exhibit B** (with such changes as the Escrow Agent may reasonably request and which Parent and the Representative shall have approved the "*Escrow Agreement*") to become effective upon the Closing, pursuant to which a portion of the aggregate Merger

Consideration otherwise payable to holders of Company Securities shall be placed in an escrow account at the Closing to secure the obligations set forth in **Section 1.6(h)(vi)** and **Article VII**.

- G. Concurrently with, and as a condition to, the Closing, Parent and each Designated Person shall have entered into an Agreement Not to Compete in substantially the form attached hereto as **Exhibit C** (the "Non-Competition Agreement"), to become effective upon the Closing.
- H. Parent, Merger Sub and the Company desire to make certain representations, warranties and agreements in connection with the Merger and also to prescribe certain conditions to the Merger.

NOW, THEREFORE, in consideration of the covenants, promises and representations set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree to amend and restate in its entirety the Prior Agreement as follows:

Article I

THE MERGER

- 1.1 **The Merger**. At the Effective Time and subject to and upon the terms and conditions of this Agreement and the applicable provisions of Delaware Law, Merger Sub shall be merged with and into the Company (the "Merger"), the separate corporate existence of Merger Sub shall cease and the Company shall continue as the surviving corporation. The Company, as the surviving corporation after the Merger, is hereinafter sometimes referred to as the "Surviving Corporation."
- 1.2 Effective Time; Closing. Subject to the provisions of this Agreement, the parties hereto shall cause the Merger to be consummated by filing a Certificate of Merger, substantially in the form attached hereto as Exhibit D (the "Certificate of Merger"), with the Secretary of State of the State of Delaware in accordance with the relevant provisions of Delaware Law (the time of such filing with the Secretary of State of the State of Delaware (or such later time as may be agreed in writing by the Company and Parent and specified in the Certificate of Merger) being the "Effective Time") as soon as practicable on or after the Closing Date (as defined below). The closing of the Merger (the "Closing") will be coordinated by means of overnight delivery, electronic mail or facsimile (rather than by means of the attendance of the parties hereto in person) at the offices of Wilson Sonsini Goodrich & Rosati, Professional Corporation, located at 701 Fifth Avenue, Suite 5100, Seattle, Washington on the third Business Day after the satisfaction or waiver of the conditions set forth in Article VI (other than those that by their terms are to be satisfied or waived at the Closing, but subject to the satisfaction or waiver of such conditions), or at such other time, date and location as the parties hereto agree in writing. The date on which the Closing occurs is referred to herein as the "Closing Date."
- 1.3 *Effect of the Merger*. At the Effective Time, the effect of the Merger shall be as provided in this Agreement and the applicable provisions of Delaware Law, including Section 259 of the Delaware General Corporation Law, as amended ("*DGCL*"). Without limiting the generality of the foregoing, and subject thereto, at the Effective Time all the property, rights, privileges, powers and franchises of the Company and Merger Sub shall vest in the Surviving Corporation, and all debts, liabilities and duties of the Company and Merger Sub shall become the debts, liabilities and duties of the Surviving Corporation.
- 1.4 *Certificate of Incorporation and Bylaws*. At the Effective Time, the Certificate of Incorporation of the Surviving Corporation shall be amended and restated in its entirety to be identical to the Certificate of Incorporation of Merger Sub, until thereafter amended in accordance with Delaware Law and as

provided in such Certificate of Incorporation; *provided*, however, that at the Effective Time, **Article I** of the Certificate of Incorporation of the Surviving Corporation shall be amended and restated in its entirety to read as follows: "The name of the corporation is "Purcell Systems, Inc." and the Certificate of Incorporation shall be amended so as to comply with **Section 5.8(b)**. At the Effective Time, the Bylaws of the Surviving Corporation shall be amended and restated in their entirety to be identical to the Bylaws of Merger Sub, until thereafter amended in accordance with Delaware Law and as provided in such Bylaws; *provided*, *however*, that at the Effective Time, if necessary, the Bylaws shall be amended so as to comply with **Section 5.8(b)**.

1.5 *Directors and Officers*. The initial directors of the Surviving Corporation shall be the directors of Merger Sub immediately prior to the Effective Time, until their respective successors are duly elected or appointed and qualified. The initial officers of the Surviving Corporation shall be the officers of Merger Sub immediately prior to the Effective Time, until their respective successors are duly appointed.

1.6 Effect on Company Securities.

- (a) *Capital Stock*. Subject to the terms and conditions of this Agreement, at the Effective Time, by virtue of the Merger and without any action on the part of Parent, Merger Sub, the Company or the holders of any shares of Capital Stock of the Company, the following shall occur:
- (i) Each share of Series A Preferred Stock issued and outstanding immediately prior to the Effective Time, other than any shares to be canceled pursuant to **Section 1.6(e)**, will be canceled and extinguished and automatically converted into the right to receive, without interest, the Per Share Amount, upon surrender of the certificate representing such share of Series A Preferred Stock in the manner provided in **Section 1.7** (or in the case of a lost, stolen or destroyed certificate, upon delivery of an affidavit and indemnity, if required in the manner provided in **Section 1.9**).
- (ii) Each share of Series B Preferred Stock issued and outstanding immediately prior to the Effective Time, other than any shares to be canceled pursuant to **Section 1.6(e)** will be canceled and extinguished and automatically converted into the right to receive, without interest, the Per Share Amount, upon surrender of the certificate representing such share of Series B Preferred Stock in the manner provided in **Section 1.7** (or in the case of a lost, stolen or destroyed certificate, upon delivery of an affidavit and indemnity, if required in the manner provided in **Section 1.9**).
- (iii) Each share of Common Stock issued and outstanding immediately prior to the Effective Time, other than any shares to be canceled pursuant to **Section 1.6(e)** will be canceled and extinguished and automatically converted into the right to receive, without interest, the Per Share Amount, upon surrender of the certificate representing such share of Common Stock in the manner provided in **Section 1.7** (or in the case of a lost, stolen or destroyed certificate, upon delivery of an affidavit and indemnity, if required in the manner provided in **Section 1.9**).
- (b) Stock Options. At the Effective Time, by virtue of the Merger and without any action on the part of Parent, Merger Sub, the Company, or the holders of any Company Options, each Company Option, will be converted into and will become a right to receive an amount in cash, without interest, with respect to each share of Common Stock subject thereto, equal to the excess, if any, of the Per Share Amount over the per share exercise price of such Company Option (such amount being hereinafter referred to as the "Option Consideration"), and each such Company Option will automatically and irrevocably terminate at the Effective Time. The payment of the Option Consideration to the holder of a Company Option will be reduced by any required tax withholdings. Notwithstanding anything in this Section 1.6(b) to the contrary, prior to the Effective Time, the Company shall take all actions necessary to effect the transactions anticipated by this Section 1.6(b) under all Company Option agreements and any other plan or arrangement of the Company (whether written or oral, formal or informal), including delivering all required notices and obtaining all waivers or consents from the holders of the Company Options which are necessary to cause the Company Options to terminate at the Effective Time as provided in this

Section 1.6(b), which waivers and consents, if any, shall be in a form reasonably satisfactory to Parent in all respects.

- (c) *Warrant Termination*. Upon the Effective Time, the Company Warrant outstanding immediately prior to the Effective Time shall be converted into the right to receive an amount of cash, without interest, with respect to each share of Capital Stock underlying such Company Warrant, equal to the excess of the Per Share Amount over the exercise price per share of such Company Warrant (the "*Warrant Termination Amount*").
- (d) *Calculations*. For purposes of calculating the amount of cash to be paid to each Company Stockholder and each holder of outstanding Company Options and/or outstanding Company Warrants pursuant to this **Section 1.6**, respectively, all amounts payable to such holder (without rounding) shall be aggregated and after such aggregation, such amount shall be rounded to the nearest whole cent. For purposes of illustration, the distribution provisions set forth in this **Section 1.6** are intended to result in the distribution of the Merger Consideration substantially in the manner set forth in **Schedule 1.6(d)**, which may be updated by the Company at the Closing and the Representative following the Closing to update the estimated amounts set forth therein with the actual amounts.
- (e) Cancellation of Treasury and Company Owned Stock. Each share of Capital Stock held by the Company or any direct or indirect wholly-owned Subsidiary of the Company immediately prior to the Effective Time shall be canceled and extinguished without any conversion thereof.

(f) [Intentionally Omitted]

(g) **Stock of Merger Sub**. Each share of common stock of Merger Sub issued and outstanding immediately prior to the Effective Time shall be converted into one (1) validly issued fully paid and nonassessable share of common stock of the Surviving Corporation.

(h) Merger Consideration Adjustment.

(i) At least three (3) Business Days prior to the Closing Date, the Company shall deliver to Parent (A) an unaudited consolidated balance sheet of the Company and its Subsidiaries prepared in accordance with GAAP applied on a basis consistent with the Company Balance Sheet and the principles and methodologies set forth on **Schedule 1.6(h)** and setting forth the Company's good faith estimate of the consolidated balance sheet of the Company as of the Closing Date (the "Estimated Closing Balance Sheet") and (B) a statement (the "Estimated Closing Statement") setting forth in reasonable detail (1) a good faith estimate of the Closing Working Capital based on the Estimated Closing Balance Sheet (such estimate, the "Estimated Closing Working Capital"), (2) the amount of the Closing Working Capital Deficit, if any, based on the amount of the Estimated Closing Working Capital (the "Estimated Closing Working Capital Deficit"), or the amount of the Closing Working Capital Surplus, if any, based on the amount of the Estimated Closing Working Capital (the "Estimated Closing Working Capital Surplus"), (3) the Total Debt Amount, comprising the Closing Debt Amount determined by reference to the pay-off letters referred to in Section 6.3(d)(i) and the Post-Closing Debt Amount determined by reference to the debt confirmation letter referred to in Section 6.3(d)(ii), (4) the Closing Transaction Expenses (if any) that are unpaid as of the Closing determined by reference to the invoices, statements and releases referred to in Section 6.3(d)(iii), and (5) a good faith estimate of the Closing Cash Amount based on the Estimated Closing Balance Sheet (the "Estimated Closing Cash Amount"). At the time of the Closing, the Merger Consideration shall be calculated based upon the Total Debt Amount (comprising the Closing Debt Amount determined by reference to the pay-off letters referred to in Section 6.3(d)(i) and the Post-Closing Debt Amount determined by reference to the debt confirmation letter referred to in Section 6.3(d)(ii)), the Closing Transaction Expenses (if any) that are unpaid as of the Closing (determined by reference to the invoices, statements, and releases referred to in Section 6.3(d)(iii)), the Estimated Closing Working Capital Deficit (if any), the Estimated Closing Working Capital Surplus (if any) and the Estimated Closing Cash Amount as set forth in the Estimated Closing Statement, subject to the

reasonable review and approval of Parent, and the Company shall (I) provide Parent and the authorized representatives of Parent copies of, or reasonable access during normal business hours to, all relevant financial information to the extent required to complete Parent's review of the Estimated Closing Statement and (II) cooperate with the reasonable requests of Parent and the authorized representatives of Parent with respect to the review of the Estimated Closing Statement, including by providing all information reasonably necessary in reviewing the Estimated Closing Statement.

(ii) Within sixty (60) Business Days following the Closing, Parent shall (A) prepare (or caused to be prepared) a consolidated balance sheet of the Company as of the Closing Date (the "Closing Balance Sheet") and a statement (the "Closing Statement" and, together with the Closing Balance Sheet, the "Closing Financial Data") indicating Parent's calculation (based on the Closing Balance Sheet and consistent with the definitions and terms set forth in this Agreement) of the Closing Working Capital Components and reconciling the Estimated Closing Working Capital Deficit (if any), the Estimated Closing Working Capital Surplus (if any) and the Estimated Closing Cash Amount as set forth in the Estimated Closing Statement and (B) deliver a copy of the Closing Financial Data to the Representative. The Closing Balance Sheet shall be prepared in accordance with GAAP applied on a basis consistent with the Company Balance Sheet and the principles and methodologies set forth on Schedule 1.6(h). Following such sixty (60) Business Day period, Parent shall be deemed to have accepted and agreed to the Company's calculation of each Closing Working Capital Component as set forth in the Estimated Closing Balance Sheet and the Estimated Closing Statement except with respect to each individual Closing Working Capital Component that is expressly disputed during such period by Parent in the Closing Financial Data and any undisputed Closing Working Capital Components shall be deemed final, binding and conclusive on Parent; provided, however, that Parent shall be deemed to have expressly disputed any Closing Working Capital Component if and to the extent that the net amount of such Closing Working Capital Component as set forth in the Estimated Closing Statement.

(iii) After receipt of the Closing Financial Data, the Representative shall have thirty (30) Business Days to review the Closing Financial Data. Parent shall (A) provide the Representative and the authorized representatives of the Representative copies of, or reasonable access during normal business hours to, all relevant financial information to the extent required to complete the Representative's review of the Closing Financial Data and (B) cooperate with the reasonable requests of the Representative and the authorized representatives of the Representative with respect to the review of the Closing Financial Data, including by providing all information reasonably necessary in reviewing the Closing Financial Data. Unless the Representative delivers a written dispute notice signed by the Representative to Parent on or prior to the thirtieth (30th) Business Day after the Representative's receipt of the Closing Financial Data (a "Dispute Notice") disputing that the Closing Financial Data is not mathematically correct, or was not prepared in accordance with GAAP applied on a basis consistent with the Company Balance Sheet and the principles and methodologies set forth on Schedule 1.6(h), then the Representative shall be deemed to have accepted and agreed to, on behalf of the Company Securityholders and the Other Participants, Parent's calculation of the Closing Working Capital Components as set forth in the Closing Financial Data, and such calculation shall be final, binding and conclusive on Parent, the Company, all Company Securityholders and the Other Participants.

(iv) If the Representative timely delivers a Dispute Notice, the Representative and Parent shall, within thirty (30) Business Days (or such longer period as Parent and the Representative may agree in writing) following receipt of such Dispute Notice (the "Resolution Period"), attempt in good faith to resolve their differences and any resolution in writing signed by each of them as to any disputed amounts shall be final, binding and conclusive.

(v) If, at the conclusion of the Resolution Period, there are any amounts remaining in dispute, then such amounts remaining in dispute may be submitted by either the Representative or Parent for binding resolution to a nationally recognized independent certified public accounting firm (which firm shall be subject to the mutual approval of Parent and the Representative and may not be Parent's independent registered certified public accounting firm or perform audit services for the Representative, or any Company Securityholders

holding more than ten percent (10%) of the issued and outstanding Company Securities, as determined on a fully-diluted basis, immediately prior to the Effective Time), appointed by the mutual agreement of Parent and the Representative (the "Independent Accountant") within ten (10) Business Days after the expiration of the Resolution Period. The Independent Accountant shall act as an arbitrator to determine whether the Closing Working Capital Components set forth in the Closing Financial Data that remain disputed following the Resolution Period (the "Disputed Components") are not mathematically correct or were not prepared in accordance with GAAP applied on a basis consistent with the Company Balance Sheet and the principles and methodologies set forth on Schedule 1.6(h) by choosing the position of Parent or the Representative. The Independent Accountant shall not have the authority to determine the accuracy of any Closing Working Capital Components other than the Disputed Components and, for the avoidance of doubt, shall not have the authority to alter any Closing Working Capital Component in favor of a party if such Closing Working Capital Component has become final, binding and conclusive on such party pursuant to this Section 1.6(h) or has been resolved by Parent and the Representative in writing during the Resolution Period. The Independent Accountant's resolution of the Disputed Components shall be made within twenty (20) Business Days following its engagement, shall be set forth in a written statement delivered to the Representative and Parent and shall be final, binding and conclusive, absent manifest error. The fees, costs and expenses of the Independent Accountant shall be paid (A) out of the Representative Fund if the amount of the Merger Consideration last proposed by Parent and (B) by Parent if the amount of the Merger Consideration calculated after substituting for the Disputed Amounts the Independent Accountant's calculation of such Disputed Amounts, is closer to the amount of the Merger Considera

(vi) If it is determined pursuant to the foregoing provisions of this Section 1.6(h) that the Closing Working Capital Components as calculated by the Company and set forth in the Estimated Closing Statement shall have been inaccurate, then (A) if such inaccuracy resulted in the Merger Consideration being overstated, Parent shall be entitled to deduct, from the Working Capital Fund, the amount by which the Merger Consideration shall have been overstated (the "Merger Consideration Excess Amount") and (B) if such inaccuracy resulted in the Merger Consideration being understated, Parent shall deliver to the Company Securityholders and the Other Participants (pursuant to the written instructions from the Representative) the amount by which the Merger Consideration was understated (the "Merger Consideration Shortfall Amount"), less the Pro Rata Portion thereof allocable to any Company Securityholder which has theretofore failed to comply with Section 1.7, and the Representative shall direct the Escrow Agent to distribute to (1) the Other Participants the respective amounts of the Merger Consideration Shortfall Amount as set forth on Schedule 1.6(d) and (2) each Company Securityholder (subject to such Company Securityholder's compliance with Section 1.7) an amount equal to such Company Securityholder's Pro Rata Portion of the Merger Consideration Shortfall Amount, after deducting the amount payable to the Other Participants pursuant to Section 1.6(h)(vi)(1) as set forth on Schedule 1.6(d), in each case if the Merger Consideration Shortfall Amount had been included in the original calculation of Merger Consideration at the time of Closing. Thereafter, the Representative shall direct the Escrow Agent to distribute to (I) the Other Participants the respective amounts of any amount remaining in the Working Capital Fund as set forth on Schedule 1.6(d) and (II) the Company Securityholders (subject to such Company Securityholder's compliance with Section 1.7) an amount equal to such Company Securityholder's Pro Rata Portion of any amount remaining in the Working Capital Fund, after deducting the amount payable to the Other Participants pursuant to Section 1.6(h)(vi)(I), as set forth on Schedule 1.6(d). Notwithstanding the foregoing, any written instructions by the Representative to the Escrow Agent shall be made in accordance with the terms of the Escrow Agreement and the agreements with each of Lazard and the Other Participants.

1.7 Surrender of Certificates.

- (a) [Intentionally Omitted]
- (b) *Parent to Provide Cash*. Parent shall make available to the Company Securityholders for exchange in accordance with this **Article I**, the Merger Consideration, less an amount of cash withheld from the amount otherwise payable to the Company Securityholders equal to the sum of the Escrow Amount, the Working

Capital Escrow Amount and the Representative Escrow Amount, which Parent shall deposit in the Escrow Fund, the Working Capital Fund and the Representative Fund, respectively, in exchange for Capital Stock, outstanding Company Options and outstanding Company Warrants. The Escrow Amount, the Working Capital Escrow Amount, the Representative Escrow Amount, the Total Debt Amount, and the Closing Transaction Expenses (if any) that are unpaid as of the Closing, shall be funded through the Merger Consideration otherwise payable to the holders of Company Securities, and the amount of cash that each holder of Company Securities would otherwise be entitled to receive following the Effective Time with respect to such holder's shares of Company Securities shall be reduced by such Company Securityholder's Pro Rata Portion of the Escrow Amount, the Working Capital Escrow Amount, the Representative Escrow Amount, the Total Debt Amount and the Closing Transaction Expenses (if any) that are unpaid as of the Closing. Any cash distributable to the Company Securityholders at Closing shall hereinafter be referred to as the "Exchange Fund." At the Closing Parent shall, for and on behalf of and for the account and responsibility of the Company, pay (i) the Closing Debt Amount to the respective holders of the indebtedness of the Company and its Subsidiaries comprising the Closing Debt Amount in accordance with the terms of the respective payoff letters delivered by the Company to Parent pursuant to Section 6.3(d)(i) and (ii) the Closing Transaction Expenses (if any) that are unpaid as of the Closing to the respective Persons entitled thereto in accordance with the respective final invoices, statements, and releases delivered by the Company to Parent pursuant to Sections 6.3(d)(iii) and 6.3(d)(iv). Following the Closing, Parent shall, for and on behalf of and for the account and responsibility of the Company, assume the obligations of the Post-Closing Debt Amount in accordance with the terms of the agreements for such indebtedness, and

(C) Exchange Procedures. At least five (5) Business Days prior to the Effective Time, Parent shall upon request make available, and within two (2) Business Days following the Effective Time, Parent shall mail to each Company Securityholder: (i) a letter of transmittal substantially in the form attached hereto as Exhibit E (which shall specify that delivery shall be effected, and risk of loss and title to the certificate or certificates or instrument or instruments (the "Certificates") shall pass, only upon delivery of the Certificates by the Company Stockholders to Parent) and (ii) instructions for use in effecting the surrender of the Certificates in exchange for cash constituting the applicable portion of the Merger Consideration (less the amounts of cash to be deposited in the Escrow Fund, the Working Capital Fund and the Representative Fund with respect to shares of Company Securities of such Company Securityholder in accordance with Section 1.7(b)), including a means of hand-delivery. Upon surrender of Certificates for cancellation to Parent required for all Company Stockholders, together with such letter of transmittal, duly completed and validly executed in accordance with the instructions thereto and such other documents as may reasonably be required by Parent for all Company Securityholders, the holder of record of such Company Securities shall be entitled to receive in exchange therefor the amount of cash constituting the portion of the Exchange Fund such holder is entitled pursuant to Section 1.6 and the Certificates so surrendered by the Company Stockholders shall forthwith be canceled. Parent shall make the foregoing payment by check or wire transfer (as elected by such Company Securityholder): (A) with respect to Certificates and transmittal letters that are delivered at least three (3) Business Days prior to the Closing Date to Parent along with the materials required by this Section 1.7(c), on the Closing Date and (B) with respect to Certificates and transmittal letters that are delivered less than three (3) Business Days prior to, or on or after, the Closing Date to Parent along with the materials required by this Section 1.7(c), within two (2) Business Days after delivery of such materials to Parent. Until so surrendered, outstanding Certificates will be deemed from and after the Effective Time, for all corporate purposes, to evidence only the ownership of the portion of the Merger Consideration into which such securities shall have been so converted.

(d) **Required Withholding**. The Surviving Corporation or its Subsidiaries, as appropriate, shall be entitled to deduct and withhold from any consideration payable or otherwise deliverable pursuant to this Agreement to any holder or former holder of Capital Stock, Company Options, or Company Warrants or to any Other Participants such amounts as may be required to be deducted or withheld therefrom under the Internal Revenue Code of 1986, as amended (the "Code") or under any provision of state, local or foreign Tax law, provided that any amounts so withheld are promptly paid over to the appropriate Governmental Entity. To the extent such amounts are so deducted or withheld and paid over, the amount of such consideration shall be treated for all

purposes under this Agreement as having been paid to the Person to whom such consideration would otherwise have been paid.

- (e) *No Liability*. Notwithstanding anything to the contrary in this **Section 1.7**, none of the Surviving Corporation or any party hereto shall be liable to a holder of shares of Capital Stock or any Company Option or Company Warrant for any amount properly paid to a public official pursuant to any applicable abandoned property, escheat or similar Legal Requirements.
- (f) Investment of Exchange Fund. Parent shall invest any cash included in the Exchange Fund in its sole discretion; provided, however, that no such investment or loss thereon shall affect the amounts payable to holders of the Company Securities pursuant to this Article I. Any interest and other income resulting from such investment shall become a part of the Exchange Fund, and any amounts in excess of the amounts payable to holders of the Company Securities pursuant to this Article I shall be paid to Parent.
- (g) Termination of Exchange Fund. Any holders of the Certificates who have not surrendered such Certificates and letters of transmittal in compliance with this Section 1.7 shall after such delivery to Surviving Corporation look only to Parent and the Surviving Corporation for the amount of cash constituting the portion of the Merger Consideration pursuant to Section 1.6(a) with respect to the shares of Capital Stock, Company Options, or Company Warrants formerly represented thereby. If any Certificate shall not have been surrendered immediately prior to such time as such amounts would otherwise escheat to or become property of any Governmental Entity, any such portion of the Exchange Fund remaining unclaimed by holders of the Company Securities immediately prior to twelve (12) months after the Effective Time shall, to the extent permitted by applicable Legal Requirements, become the property of the Surviving Corporation free and clear of any claims or interest of any Person previously entitled thereto.
- 1.8 No Further Ownership Rights in Capital Stock. Subject in all respects to the provisions of the last sentence of Section 1.7(c), all cash paid upon the surrender for exchange of shares of Capital Stock, Company Options and Company Warrants in accordance with the terms hereof shall be deemed to have been issued in full satisfaction of all rights pertaining to such shares of Capital Stock, Company Options and Company Warrants and there shall be no further registration of transfers on the records of the Surviving Corporation of shares of Capital Stock which were outstanding immediately prior to the Effective Time. If, after the Effective Time, Certificates are presented to the Surviving Corporation for any reason, they shall be canceled and exchanged as provided in this Article I.
- 1.9 Lost, Stolen or Destroyed Certificates. In the event any Certificates shall have been lost, stolen or destroyed, Parent shall transfer in exchange for such lost, stolen or destroyed Certificates, upon the making of an affidavit of that fact, which shall include usual and customary indemnities in favor of the Surviving Corporation, by the holder thereof, such amount of cash constituting the applicable portion of the Merger Consideration pursuant to Section 1.6(a) with respect to the shares of Capital Stock, Company Options or Company Warrants formerly represented thereby.
- 1.10 *Further Action*. At and after the Effective Time, the officers and directors of Parent and the Surviving Corporation will be authorized to execute and deliver, in the name and on behalf of the Company and Merger Sub, any deeds, bills of sale, assignments or assurances and to take and do, in the name and on behalf of the Company and Merger Sub, any other actions and things to vest, perfect or confirm of record or otherwise in the Surviving Corporation any and all right, title and interest in, to and under any of the rights, properties or assets acquired or to be acquired by the Surviving Corporation as a result of, or in connection with, the Merger.

ARTICLE II

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to Parent and Merger Sub as of the date of the Prior Agreement (and, accordingly, the phrases "as of the date hereof," "as of the date of this Agreement," or any similar terms as used in this **Article II** will be deemed to refer to the date of the Prior Agreement) and as of the Closing Date, except as set forth in the Company Disclosure Letter delivered to the Parent as of the date of the Prior Agreement, which information shall be deemed to be disclosed for purposes of the corresponding Section or subsection of this Agreement and no disclosure made in any particular part of the Company Disclosure Letter shall be deemed made in any other part unless (a) expressly made therein (by cross-reference or otherwise) or (b) it is readily apparent on the face of such disclosure that such disclosure applies to such other representations and warranties, as follows:

2.1 Organization; Standing and Power; Charter Documents; Subsidiaries.

- (a) *Organization; Standing and Power*. The Company and each of its Subsidiaries (i) is a corporation or other organization duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation or organization (except, in the case of good standing, for entities organized under the laws of any jurisdiction that does not recognize such concept), (ii) has the requisite power and authority to own, lease and operate its properties and to carry on its business as now being conducted, and (iii) is duly qualified or licensed and in good standing to do business in each jurisdiction in which the nature of its business or the ownership or leasing of its properties makes such qualification or licensing necessary.
- (b) *Charter Documents*. The Company has delivered or made available to Parent in the Virtual Data Room: (i) a true and correct copy of the Certificate of Incorporation and Bylaws of the Company, each as amended to date (collectively, the "*Company Charter Documents*") and (ii) the Certificate of Incorporation and Bylaws, or like organizational documents (collectively, "*Subsidiary Charter Documents*"), of each of its Subsidiaries, and each such instrument is in full force and effect. The Company is not in violation of any of the provisions of the Company Charter Documents and each Subsidiary is not in violation of its respective Subsidiary Charter Documents.
- (c) Subsidiaries. Section 2.1(c) of the Company Disclosure Letter sets forth each Subsidiary of the Company as of the date hereof. All the outstanding shares of capital stock of, or other equity or ownership interests in, each such Subsidiary have been validly issued and are fully paid and nonassessable and are owned by the Company, a wholly-owned Subsidiary of the Company, or the Company and another wholly-owned Subsidiary of the Company, free and clear of all pledges, liens, charges, encumbrances and security interests of any kind or nature whatsoever (collectively, "Liens"), other than Permitted Liens, and are duly authorized, validly issued, fully paid and nonassessable. Other than the Subsidiaries of the Company, neither the Company nor any of its Subsidiaries owns any capital stock of, or other equity or ownership interests of any nature in, or any interest convertible, exchangeable or exercisable for, capital stock of, or other equity or ownership interests of any nature in, any other Person, nor has the Company or any of its Subsidiaries made any debt investment in any other Person.

2.2 Capital Structure.

(a) *Capital Stock*. The authorized capital stock of the Company consists of: (i) 10,000,000 shares of Common Stock, and (ii) 6,441,335 shares of Preferred Stock, of which 601,205 shares have been designated as Series A Preferred Stock and 5,840,130 shares have been designated as Series B Preferred Stock. As of the date hereof: 1,092,136 shares of Common Stock were issued and outstanding; 455,661 shares of Series A

Preferred Stock were issued and outstanding and 5,730,130 shares of Series B Preferred Stock were issued and outstanding; 977,663 shares of Common Stock were reserved for issuance upon exercise of Options granted pursuant to the Company Option Plan and 240,000 shares of Common Stock that were reserved for issuance upon exercise of an option granted outside of the Company Option Plan; and 110,000 shares of Series B Preferred Stock were reserved for issuance upon exercise of the Company Warrant. Except as otherwise set forth in this **Section 2.2**, as of the date hereof, there are no securities, options, warrants, calls, rights, Contracts, obligations or undertakings of any kind to which the Company or any of its Subsidiaries is a party or by which any of them is bound obligating the Company or any of its Subsidiaries to (including on a deferred basis) issue, deliver or sell, or cause to be issued, delivered or sold, additional shares of capital stock, voting debt or other securities of the Company or any of its Subsidiaries, or obligating the Company or any of its Subsidiaries to issue, grant, extend or enter into any such security, option, warrant, call, right, Contract, obligation or undertaking.

- (b) Stock Options and Warrants. Section 2.2(b) of the Company Disclosure Letter sets forth a true, complete and correct list of all Persons who, at the close of business on the date hereof, hold outstanding Company Warrants or Company Options indicating, with respect to each Company Option and Company Warrant then outstanding, the number of shares of Common Stock and Preferred Stock subject to such Company Option or Company Warrant, and the exercise price applicable thereto and the date of grant, vesting schedule and expiration date thereof, including the extent to which any vesting had occurred as of the date of this Agreement and whether the vesting of such Company Option or Company Warrant will be accelerated in any way by the consummation of the transactions contemplated by this Agreement or by the termination of employment or engagement or change in position of any holder thereof following or in connection with the consummation of the Merger.
- (c) Compliance with Law and Other Securities Representations. All outstanding shares of Capital Stock, all outstanding Company Options, all outstanding Company Warrants, and all outstanding shares of capital stock or other equity or ownership interests of each Subsidiary of Company have been issued and granted in compliance in all material respects with all applicable securities laws and all other applicable Legal Requirements. No shares of Capital Stock are owned or held by any Subsidiary of the Company. All of the outstanding shares of Capital Stock of the Company are, and all shares of Capital Stock of the Company which may be issued as contemplated or permitted by this Agreement will be, when issued, duly authorized and validly issued, fully paid and nonassessable and not subject to any preemptive rights. There are not any outstanding Contracts of the Company or any of its Subsidiaries to (i) repurchase, redeem or otherwise acquire any shares of capital stock of, or other equity or ownership interests in, the Company or any of its Subsidiaries or (ii) dispose of any shares of the capital stock of, or other equity or ownership interests in, the Company or any of its Subsidiaries and there are no irrevocable proxies and no voting agreements, voting trusts, rights plans or anti-takeover plans with respect to any shares of the capital stock of, or other equity or ownership interests in, the Company or any of its Subsidiaries.

2.3 Authority; Non-Contravention; Necessary Consents.

(a) Authority. The Company has all requisite corporate power and authority to enter into this Agreement and to consummate the transactions contemplated hereby. The Company's execution and delivery of this Agreement and the consummation of the transactions contemplated hereby by the Company have been duly authorized by all necessary corporate action on the part of the Company and no other corporate proceedings on the part of the Company are necessary to authorize the execution and delivery of this Agreement or to consummate the Merger and the other transactions contemplated hereby, subject only to the adoption of this Agreement by the Company's stockholders and the filing of the Certificate of Merger pursuant to Delaware Law. This Agreement has been duly executed and delivered by the Company and, assuming due execution and delivery by Parent and Merger Sub, constitutes a valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except (i) as enforcement may be limited by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other Legal Requirements affecting the rights of creditors generally and general equitable principles (whether considered in a proceeding in equity or at law), and (ii) as the remedy of specific

performance and injunctive and other forms of equitable relief may be subject to equitable defenses and to the discretion of a court of competent jurisdiction before which any proceeding may be brought.

- (b) **Board Approval**. As of the date hereof, the Board of Directors of the Company has, by resolutions duly adopted at a meeting duly called and held (which resolutions have not been subsequently rescinded or modified in any way), (i) determined the Merger to be advisable, (ii) approved this Agreement and the transactions contemplated hereby, including the Merger, and (iii) recommended that the stockholders of the Company vote in favor of the adoption of this Agreement and directed that such matter be submitted to Company's stockholders for approval.
- (c) Non–Contravention. The execution and delivery of this Agreement by the Company does not, and performance of this Agreement by the Company will not: (i) conflict with or violate the Company Charter Documents or any Subsidiary Charter Documents of any Subsidiary of the Company, (ii) subject to obtaining the Required Stockholder Approvals, conflict with or violate any material Legal Requirement applicable to the Company or any of its Subsidiaries or by which the Company or any of its Subsidiaries or any of their respective properties is bound or affected, or (iii) result in any breach of or constitute a default (or an event that with notice or lapse of time or both would become a default) under, or impair the Company's rights or alter the rights or obligations of any third party under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of a Lien on any of the properties or assets of the Company or any of its Subsidiaries pursuant to any Company Material Contract.
- (d) Necessary Consents. No consent, approval, order or authorization of, or registration, declaration or filing with any supranational, national, state, local or foreign government, any instrumentality, subdivision, court, administrative agency or commission or other governmental authority or instrumentality (a "Governmental Entity") is required to be obtained or made by the Company or any of its Subsidiaries in connection with the execution and delivery of this Agreement or the consummation of the Merger and other transactions contemplated hereby, except for: (i) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware and appropriate documents with the relevant authorities of other states in which the Company and/or Merger Sub are qualified to do business set forth in Section 2.3(d)(i) of the Company Disclosure Letter, (ii) the consents, approvals, orders, authorizations, registrations, declarations and filings with a Governmental Entity required under applicable federal, foreign and state securities (or related) laws set forth in Section 2.3(d)(ii) of the Company Disclosure Letter and the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act") and satisfaction of such other requirements of comparable Legal Requirements of foreign jurisdictions set forth in Section 2.3(d)(ii) of the Company Disclosure Letter, (iii) such consents, approvals, orders, authorizations, registrations and filings as may be required under applicable state securities or "blue sky" laws and the securities laws of any foreign country set forth in Section 2.3(d)(ii) of the Company Disclosure Letter, (iv) the consents, authorizations, filings, approvals and registrations with a Governmental Entity, which if not obtained or made would not reasonably be expected to have a Material Adverse Effect on the Company. The consents, approvals, orders, authorizations, registrations, declarations and filings with a Governmental Entity set forth in (i) through (v), inclusive, are referred to
- 2.4 *Financial Statements*. Section 2.4 of the Company Disclosure Letter includes (a) a true and complete copy of the Company's consolidated audited financial statements at, and for the years ended, December 31, 2010, December 31, 2011, and December 31, 2012 and (b) a true and complete copy of the Company's consolidated unaudited financial statements at, and for the seven months ended, July 31, 2013 (collectively, the *"Financial Statements"*). The Financial Statements have been prepared in accordance with United States generally accepted accounting principles (*"GAAP"*) (except that the Company's unaudited financial statements at, and for the seven months ended, July 31, 2013 do not have notes thereto) applied on a consistent basis throughout the periods indicated and present fairly in all material respects the financial condition, operating results, and cash flows of the Company and its consolidated Subsidiaries as of the dates, and for the periods, indicated therein, subject to normal year-end audit adjustments in the case of the Company's unaudited financial statements at, and for the seven months ended, July 31, 2013 only. The Company maintains with respect to the Company Business

a standard system of accounting established and administered in accordance with GAAP, and the respective books of account, record books, and other books and records of the Company and its Subsidiaries are complete and correct in all material respects. The Company's unaudited balance sheet as of July 31, 2013, is referred to as the "Company Balance Sheet."

- 2.5 *Undisclosed Liabilities*. As of the date hereof, neither the Company nor any of its Subsidiaries has any material liabilities (whether liquidated, contingent or otherwise), except for (i) liabilities set forth or provided for in the Financial Statements (or the notes thereto), (ii) current liabilities which have arisen or have been incurred in the ordinary course of business since the date of the Company Balance Sheet, (iii) liabilities expressly disclosed in Section 2.5 of the Company Disclosure Letter, or (iv) liabilities incurred in connection with this Agreement or the transactions contemplated hereby consisting solely of the Closing Transaction Expenses.
 - 2.6 *Absence of Certain Changes or Events*. Since December 31, 2012 there has not been:
 - (h) a Material Adverse Effect on the Company or any of its Subsidiaries;
 - (i) any amendment to any of the Company Charter Documents or Subsidiary Charter Documents;
- (j) any declaration, setting aside or payment of any dividend on, or other distribution (whether in cash, stock or property) in respect of, any of the Company's or any of its Subsidiaries' capital stock or other securities;
- (k) any issuance, sale or grant of any capital stock or other equity or equity-related interest of the Company or any of its Subsidiaries or any options, warrants or rights to subscribe for or purchase any such capital stock or other equity or equity-related interest (except for issuances of Company Common Stock pursuant to the exercise of Company Options or Company Warrants outstanding as of the date hereof in accordance with the terms thereof);
- (1) any direct or indirect purchase, redemption or other acquisition by the Company or any of its Subsidiaries of any of the capital stock of the Company or any of its Subsidiaries or any other securities of the Company or its Subsidiaries or any options, warrants, calls or rights to acquire any such shares or other securities except for repurchases from Employees following their termination pursuant to the terms of their pre-existing stock option or purchase agreements;
- (m) any split, combination, subdivision or reclassification of any of the Company's or any of its Subsidiaries' capital stock or other securities or change or agreement to change in any manner the rights, preferences or designations thereof;
 - (n) any resignation or termination of any officer or director of the Company or any of its Subsidiaries;
- (o) any grant or agreement to grant any increases in the compensation, perquisites or benefits (whether through the payment of, or agreement to pay, bonus amounts or otherwise) to any Employee (other than in the ordinary course of business and consistent with past practices) or the execution or amendment by the Company or any of its Subsidiaries of any employment, severance, deferred compensation or similar Contract with any Employee;
- (p) any increase of severance or increase of termination pay to any Employee of the Company or any Subsidiary of the Company (other than severance or termination paid in the ordinary course of business and consistent with past practices) or any change in severance policies or practices of the Company or any of its Subsidiaries;

- (q) any material change in the accounting methods, principles or practices of the Company or any of its Subsidiaries, except as required by concurrent changes in GAAP and disclosed in the Financial Statements;
- (r) any material revaluation by the Company or any of its Subsidiaries of any of their respective assets, including writing down the value of capitalized inventory or writing off notes or accounts receivable other than in the ordinary course of business;
- (s) any damage, destruction or loss, whether or not covered by insurance, affecting any of the assets of the Company or any of its Subsidiaries having a fair market value of Fifty Thousand Dollars (\$50,000.00) or more or otherwise materially and adversely affecting the business of the Company or any of its Subsidiaries;
- (t) any adoption, amendment or termination of any Company Benefit Plan (other than in the ordinary course of business or as otherwise required to comply with applicable Legal Requirements);
- (u) any acquisition, whether by merger or consolidation by purchase of assets, or by any other manner, of any business or any corporation, partnership, association or other business organization or division thereof by the Company or any of its Subsidiaries or any other merger by the Company or any of its Subsidiaries with or into, or consolidation by the Company or any of its Subsidiaries with, any other Person;
- (v) any acquisition or disposition of assets of or by the Company or any of its Subsidiaries (other than the sale of inventory in the ordinary course of business and consistent with past practices) or incurrence or imposition of any Lien on any assets, or any interest in any assets, of the Company or any of its Subsidiaries;
- (w) any failure to pay, or delay in paying, accounts payable or other current liabilities or obligations of the Company or any of its Subsidiaries in excess of One Hundred Thousand Dollars (\$100,000.00) when due except where disputed in good faith by the Company or any of its Subsidiaries by appropriate proceedings;
- (x) any acceleration of collection of accounts receivable of the Company or any of its Subsidiaries in excess of One Hundred Thousand Dollars (\$100,000.00) in advance of the applicable due dates;
- (y) any making of any loan, advance, investment, or capital contributions by the Company or any of its Subsidiaries in or to any Person, except employee loans or advances made in the ordinary course of business consistent with past practices;
- (z) any making or commitment to make any capital expenditures or capital additions or betterments by the Company or any of its Subsidiaries in excess of One Hundred Thousand Dollars (\$100,000.00) individually or Two Hundred Fifty Thousand Dollars (\$250,000.00) in the aggregate; or
- (aa) any execution, termination or amendment of, or receipt of notice of termination of, any Company Material Contract or the waiver or release of any material rights or remedies by the Company or any of its Subsidiaries under any Company Material Contract or otherwise.

2.7 *Taxes*.

(a) **Definition**. For the purposes of this Agreement, the term "Tax" or, collectively, "Taxes" shall mean any and all U.S. federal, state, local and non-U.S. taxes and other like governmental charges, including taxes based upon or measured by gross receipts, income, profits, sales, use and occupation, and value added, ad valorem, transfer, franchise, withholding, payroll, recapture, employment, excise and property taxes, together with all interest, penalties and additions imposed with respect to such amounts.

(b) Tax Returns and Audits.

- (i) The Company and each of its Subsidiaries have prepared and timely filed all federal, state, local and non-U.S. returns, estimates, information statements and reports ("Tax Returns") that are or were required to be filed by them pursuant to applicable Legal Requirements relating to any and all Taxes concerning or attributable to the Company, its Subsidiaries or their respective operations and have timely and fully paid, or made provision for the payment of, all Taxes due and payable by them, regardless of whether or not shown on such Tax Returns, and such Tax Returns have been completed in accordance with all applicable Legal Requirements and are true, correct, and complete.
- (ii) The Company and each of its Subsidiaries have paid or withheld with respect to their Employees (and paid over to the appropriate taxing authority) all Taxes required to be paid or withheld.
- (iii) Neither the Company nor any of its Subsidiaries has executed any outstanding waiver of any statute of limitations on or outstanding extension of the period for the assessment or collection of any Tax.
- (iv) No audit or other examination of any Tax Return of the Company or any of its Subsidiaries is presently in progress, nor has the Company or any of its Subsidiaries been notified in writing of any request for such an audit or other examination.
- (v) Neither the Company nor any of its Subsidiaries has any liabilities for unpaid Taxes which have not been fully accrued or reserved on the Company Balance Sheet in accordance with GAAP, and neither the Company nor any of its Subsidiaries has incurred any liability for Taxes since the date of the Company Balance Sheet to the date of this Agreement other than in the ordinary course of business.
- (vi) There are no Liens on any of the assets of the Company or any of its Subsidiaries relating to or attributable to Taxes other than Permitted Liens.
- (vii) Neither the Company nor any of its Subsidiaries is a "United States Real Property Holding Corporation" within the meaning of Section 897(c)(2) of the Code.
- (viii) Neither the Company nor any of its Subsidiaries (a) has ever been a member of an affiliated group (within the meaning of Code § 1504(a)) filing a consolidated federal income Tax Return (other than a group the common parent of which was Company), (b) owes any amount under any Tax sharing, indemnification or allocation agreement, or (c) has any liability for the Taxes of any Person (other than Company or any of its Subsidiaries) under Treas. Reg. § 1.1502-6 (or any similar provision of state, local or non-U.S. law), as a transferee or successor, by Contract, or otherwise.
- (ix) Neither the Company nor any of its Subsidiaries has constituted either a "distributing corporation" or a "controlled corporation" in a distribution of stock intended to qualify for tax-free treatment under Section 355 of the Code.
- (x) Set forth in **Section 2.7(b)(x)** of the Company Disclosure Letter is a list of (A) all Tax Returns filed by the Company and its Subsidiaries during the three (3) year period immediately preceding the date of this Agreement, (B) all Tax Returns required to be filed by the Company and its Subsidiaries for any Tax period ending on or before the Closing Date that are required to be filed after the Closing Date (taking into account any valid and effective extensions of due dates), and (C) all Tax Returns required to be filed by the Company and its Subsidiaries for any Tax period ending after, but which includes, the Closing Date, that are required to be filed after the Closing Date (taking into account any valid and effective extensions of due dates), in each case setting forth the applicable jurisdiction in which such Tax Returns are required to be filed by the Company

or any of its Subsidiaries, the applicable due date or the extended due date requested or obtained (if applicable) and the amount of Tax (if any) paid in connection with any request for or extension of any due date.

(xi) Neither the Company nor any of its Subsidiaries has engaged in any transaction that, at the time undertaken, was a "reportable transaction" (within the meaning of Treas. Reg. § 1.6011-4).

2.8 Intellectual Property.

(a) Definitions.

- (xii) "Company Business" means the design, development, manufacture, marketing and sale of Company Products as conducted as of the date of this Agreement.
- (xiii) "Company Intellectual Property Rights" means Intellectual Property Rights owned by the Company or any of its Subsidiaries.
- (xiv) "Company Products" means the products or services manufactured, marketed, licensed or sold by the Company or any of its Subsidiaries as of the date of this Agreement.
- (XV) "Intellectual Property" means, regardless of form: (1) published and unpublished works of authorship, including without limitation audiovisual works, collective works, computer programs, compilations, databases, derivative works, literary works, mask works, and sound recordings ("Works of Authorship"); (2) inventions and discoveries, including without limitation articles of manufacture, business methods, compositions of matter, improvements, machines, methods, and processes and new uses for any of the preceding items ("Inventions"); (3) words, names, symbols, devices, designs, and other designations, and combinations of the preceding items, used to identify or distinguish a business, good, group, product, or service or to indicate a form of certification, including without limitation logos, product designs, and product features ("Trademarks"); and (4) information that is not generally known or readily ascertainable through proper means, whether tangible or intangible, including without limitation, customer lists, supplier lists, pricing and cost information, business and marketing plans and proposals, algorithms, ideas, designs, formulas, know-how, methods, processes, programs, prototypes, systems and techniques ("Confidential Information").
- (XVI) "Intellectual Property Rights" means all rights in, arising out of, or associated with Intellectual Property in any jurisdiction, including: (1) rights in, arising out of, or associated with Works of Authorship, including rights granted under the Copyright Act ("Copyrights"); (2) rights in, arising out of, or associated with Inventions, including rights granted under the Patent Act ("Patent Rights"); (3) rights in, arising out of, or associated with Trademarks, including rights granted under the Lanham Act ("Trademark Rights"); and (4) rights in, arising out of, or associated with Confidential Information, including rights granted under the Uniform Trade Secrets Act ("Trade Secret Rights").
- (xvii) "Registered Intellectual Property Rights" means all Intellectual Property Rights that are the subject of an application, certificate, filing, registration, or other document issued by, filed with, or recorded by, any Governmental Entity in any jurisdiction, including all applications, reissues, divisions, re-examinations, renewals, extensions, provisionals, continuations, and continuations-in-part associated with Patent Rights.
- (b) Section 2.8(b) of the Company Disclosure Letter (i) lists all Registered Intellectual Property Rights owned by, filed in the name of, or applied for by the Company or any of its Subsidiaries ("Company Registered Intellectual Property Rights"); and (ii) lists all actions that must be taken by the Company or any of its Subsidiaries within sixty (60) days of the Closing Date to maintain the Company Registered Intellectual Property Rights.

- (c) The Company or a Subsidiary of the Company, as appropriate, owns the Company Intellectual Property Rights. To the Knowledge of the Company, there are no third parties that claim to own any Company Intellectual Property Rights.
- (d) The Company Intellectual Property Rights are subsisting and, to the Knowledge of the Company, enforceable. To the Knowledge of the Company, there are no facts or circumstances that could impair the validity or enforceability of any Company Intellectual Property Rights.
- (e) To the Knowledge of the Company, the conduct of the Company Business does not infringe, misappropriate, or otherwise violate the Intellectual Property Rights of a third party. Neither the Company nor any of its Subsidiaries has received written notice of, nor to the Knowledge of the Company are there any facts or circumstances that would be reasonably likely to result in, a claim that the conduct of the Company Business infringes, misappropriates, or otherwise violates the Intellectual Property Rights owned by a third party.
- (f) Other than Contracts consisting of "shrink wrap," "click wrap" or "execute-by-opening" or similar license agreements relating to commercially available, off-the-shelf or hosted computer software licensed by the Company or its Subsidiaries (as licensee) which do not require the payment of any licensing, maintenance, royalty, or other fee for the current level of use of such software by the Company or any of its Subsidiaries or upon the transfer of any such license, the Contracts listed in **Section 2.8(f)** of the Company Disclosure Letter include all Contracts to which Company or any of its Subsidiaries is a party with respect to any license of Intellectual Property owned by any Person other than Company or its Subsidiaries ("Inbound Licenses"). The Company and each of its Subsidiaries have valid, existing licenses for all copies of computer software used by them in connection with the Company Business.
- (g) Other than Contracts consisting of "shrink wrap," "click wrap," or "execute by opening" or similar license agreements relating to commercially available, off-the-shelf or hosted computer software licensed by the Company or its Subsidiaries (as licensee) which do not require the payment of any licensing, maintenance, royalty, or other fee for the current level of use of such software by the Company or any of its Subsidiaries or upon the transfer of any such license, the Company Intellectual Property, together with the Inbound Licenses, includes all Intellectual Property used in or necessary to the conduct of the Company Business.
- (h) Other than licenses granted substantially in the form of the Company's standard forms of agreement (correct and complete copies of which standard forms have been made available to Parent on the Virtual Data Room, or Contracts for less than One Hundred Thousand Dollars (\$100,000.00), Section 2.8(h) of the Company Disclosure Letter sets forth all Contracts, agreements or other licenses pursuant to which the Company or any of its Subsidiaries grants any rights under any Company Intellectual Property to any Person ("Outbound Licenses").
- (i) To the Knowledge of the Company, there are no Inbound Licenses or Outbound Licenses under which there is any dispute regarding the scope of such agreement, or performance under such agreement, including with respect to any payments to be made or received by the Company or any of its Subsidiaries thereunder.
- (j) The Company and its Subsidiaries may exercise, transfer, or license the Company Intellectual Property Rights without restriction or payment to a third party.
- (k) The Company and each of its Subsidiaries takes reasonable steps to maintain the secrecy of Confidential Information from which the Company or any of its Subsidiaries derives independent economic value, actual or potential, from that Confidential Information not being generally known.
- (l) To the Knowledge of the Company, no third party is infringing, misappropriating, or otherwise violating the Company Intellectual Property Rights.

(m) There is no software or other material that is distributed pursuant to an open source license listed at www.opensource.org/licenses as of August 31, 2013 in Company Products ("*Open Source Materials*"), and the manufacture, sale or distribution by the Company or any of its Subsidiaries of any such Company Products does not consist of or include any Open Source Materials.

2.9 Compliance; Permits.

- (a) *Compliance*. The Company and each of its Subsidiaries has complied in all material respects, and is in compliance in all material respects, with each Legal Requirement applicable to the Company or its Subsidiaries or by which the Company or any of its Subsidiaries or any of their respective businesses or properties is bound or affected. There is no material judgment, injunction, order or decree binding upon the Company or any of its Subsidiaries which has or would reasonably be expected to have the effect of prohibiting or materially impairing the Company Business.
- (b) **Permits**. The Company and its Subsidiaries hold, to the extent legally required, all permits, licenses, variances, clearances, consents, commissions, franchises, exemptions, orders and approvals from Governmental Entities ("**Permits**") that are material to the operation of the business of the Company and its Subsidiaries taken as a whole, including Permits required under any applicable Environmental Laws (collectively, "**Company Permits**"). **Section 2.9(b)** of the Company Disclosure Letter contains a complete and accurate list of each of the Company Permits, in each case indicating the expiration date thereof, if any. As of the date hereof, no suspension or cancellation of any of the Company Permits is pending or, to the Knowledge of the Company, threatened. The Company and its Subsidiaries are in compliance in all material respects with the terms of the Company Permits.
- 2.10 *Litigation*. As of the date hereof, there are no claims, suits, actions or proceedings pending or, to the Knowledge of the Company, threatened against the Company or any of its Subsidiaries, before any court, governmental department, commission, agency, instrumentality or authority, or any arbitrator that seeks to restrain or enjoin the consummation of the transactions contemplated hereby or which would reasonably be expected, either singularly or in the aggregate with all such claims, actions or proceedings, to have a Material Adverse Effect on the Company.
- **2.11** *Brokers' and Finders' Fees*. Except for fees payable to Lazard, the Company and its Subsidiaries have not incurred, nor will any of them incur, directly or indirectly, any liability for brokerage or finders' fees or agents' commissions or any similar charges in connection with this Agreement.

2.12 Employee Benefit Plans.

- (a) *Schedule*. **Section 2.12(a)** of the Company Disclosure Letter sets forth a complete and accurate list of each Company Benefit Plan.
- (b) Benefit Plan Compliance.
- (i) Each bonus, pension, profit sharing, deferred compensation, incentive compensation, stock ownership, stock purchase, stock option, phantom stock, stock-related or performance award, retirement, vacation, severance, disability, death benefit, hospitalization, medical, loan (other than travel allowances and relocation packages), fringe benefit, disability, sabbatical and other plan or arrangement providing benefits to any current or former employees, consultants, officers, or directors of the Company or its Subsidiaries (each, an "Employee") pursuant to which the Company or its Subsidiaries have or could have liability ("Company Benefit Plans") has been administered and operated in accordance with its terms, with the applicable provisions of the Employee Retirement Income Security act of 1974, as amended ("ERISA"), the Code and all other applicable material Legal Requirements. Each Company Benefit Plan intended to qualify under Section 401(a) of the Code and each trust intended to qualify under Section 501(a) of the Code has either received a favorable determination,

opinion, notification or advisory letter from the IRS with respect to each such Company Benefit Plan as to its qualified status under the Code, or has remaining a period of time under applicable Treasury Regulations or IRS pronouncements in which to apply for such a letter and make any amendments necessary to obtain a favorable determination as to the qualified status of each such Company Benefit Plan.

- (ii) No oral or written representation or commitment with respect to any aspect of any Company Benefit Plan has been made to an Employee of the Company or any of its Subsidiaries by an authorized Employee of the Company or any of its Subsidiaries that is not in accordance with the written or otherwise preexisting terms and provisions of such Company Benefit Plans.
- (iii) To the Knowledge of the Company, there are no unresolved claims or disputes under the terms of, or in connection with, any Company Benefit Plan (other than routine undisputed claims for benefits), and no action, legal or otherwise, has been commenced with respect to any such claim.
- (c) **Defined Benefit, Multiple Employer and Multiemployer Plans.** At no time has the Company or any other Person under common control within the meaning of Section 414(b), (c), (m) or (o) of the Code (a "Controlled Group Affiliate") with the Company or any of its Subsidiaries maintained, established, sponsored, participated in, or contributed to, any (i) Company Benefit Plan subject to Title IV of ERISA, (ii) multiemployer plan (as defined in Section 3(37) of ERISA), or (iii) "multiple employer plan" as defined in ERISA or the Code.
- (d) *Continuation Coverage*. No Company Benefit Plan provides health benefits (whether or not insured), with respect to Employees after retirement or other termination of service (other than coverage mandated by applicable Legal Requirements or benefits, the full cost of which is borne by the Employee).
- (e) International Employee Plans. The Company and its Subsidiaries do not now, nor have any of them ever had the obligation to, maintain, establish, sponsor, participate in, or contribute to any International Employee Plan. As used in this Agreement, "International Employee Plan" shall mean each Company Benefit Plan that has been adopted or maintained by the Company or its Subsidiaries, whether informally or formally, or with respect to which the Company or its Subsidiaries will or may have any material liability or which the Company or any of its Subsidiaries is required to maintain or otherwise contribute to under any applicable Legal Requirements, pursuant to any Contract with any labor council, or otherwise, for the benefit of Employees who perform services outside the United States.
- (f) Effect of Transaction. The execution of this Agreement and the consummation of the transactions contemplated by this Agreement will not (either alone or in connection with the termination of employment or engagement of change of position of any Employee following or in connection with the consummation of the Merger) constitute an event under any Company Benefit Plan that will result in any payment (whether of severance pay or otherwise), acceleration of payment, forgiveness of indebtedness, vesting, distribution, increase in benefits or obligation to fund benefits with respect to any Employee. There is no Contract plan or arrangement with an Employee to which the Company or any of its Subsidiaries is a party as of the date of this Agreement, that, individually or collectively and as a result of the transactions contemplated hereby (whether alone or in connection with the termination of employment or engagement of change of position of any Employee following or in connection with the consummation of the Merger), would reasonably be expected to give rise to the payment of any amount that would not be deductible pursuant to Section 280G of the Code.
- (g) *Labor*. Neither the Company nor any of its Subsidiaries is a party to any collective bargaining agreement or union contract with respect to Employees and no collective bargaining agreement is being negotiated by the Company or any of its Subsidiaries. There is no labor dispute, strike or work stoppage against the Company or any of its Subsidiaries pending or, to the Knowledge of the Company, threatened or reasonably anticipated which may materially interfere with the respective business activities of the Company or any of its Subsidiaries. To the Knowledge of the Company, none of the Company, any of its Subsidiaries or any of their

respective representatives or Employees has committed any unfair labor practice in connection with the operation of the respective businesses of the Company or any of its Subsidiaries. There are no actions, suits, claims, labor disputes or grievances pending, or, to the Knowledge of the Company, threatened, relating to any labor, safety or discrimination matters involving any Employee, including, without limitation, charges of unfair labor practices or discrimination complaints.

(h) *Employment Matters*. The Company and each of its Subsidiaries is in compliance in all material respects with all applicable Legal Requirements respecting employment, employment practices, terms and conditions of employment, employee safety and wages and hours.

2.13 Real Property.

- (a) Neither the Company nor any of its Subsidiaries own any real property. Section 2.13 of the Company Disclosure Letter sets forth a list of the real property currently leased, subleased or licensed by or from the Company or its Subsidiaries (the "Leased Real Property"). The Company and its Subsidiaries have provided or otherwise made available to Parent in the Virtual Data Room true, correct and complete copies of all leases, lease guaranties, subleases, agreements for the leasing, use or occupancy of the Leased Property, including all amendments, terminations and modifications thereof (the "Leases"), all of which Leases are identified on Section 2.13 of the Company Disclosure Letter. There is not, under any of the Leases, any material default by the Company or any of its Subsidiaries, nor, to the Knowledge of the Company, by any other party thereto. There are no other parties occupying, or with a right to occupy, the Leased Real Property other than the Company or any of its Subsidiaries.
- (b) To the Knowledge of the Company, the heating, ventilation, air conditioning and other mechanical and electrical systems serving the Leased Real Property are in reasonable operating condition in all material respects. All leasehold improvements installed by the Company in the Leased Real Property, are in reasonable operating condition in all material respects. To the Knowledge of the Company, the buildings and structures which comprise a portion of the Leased Real Property (including the roofs thereon) are free of any material structural or engineering defects. During the past three (3) years, there has not been any significant interruption or disruption of the operations of the Company Business (excluding power outages due to force majeure events or supply interruptions associated with the applicable utility company) due to inadequate maintenance of any of the buildings, improvements or tangible personal property which comprise a portion, or which are located at or on, the Leased Real Property.
- 2.14 Assets. With respect to all of the assets that they purport to own, the Company and each of its Subsidiaries owns all such assets free and clear of all Liens, except Permitted Liens, and such assets are in operating condition (normal wear and tear excepted). For purposes of clarity, this Section 2.14 does not relate to real property (such items being the subject of Section 2.13) or Intellectual Property (such items being the subject of Section 2.8).

2.15 Environmental Matters.

(a) To the Knowledge of the Company and except as would not reasonably be expected to result in liability to the Company, no Hazardous Materials (i) are present in violation of applicable Environmental Laws in, on, or under the Leased Real Property or (ii) were present in violation of applicable Environmental Laws in, on, or under any real property formerly owned, leased, subleased or licensed by or from the Company or its Subsidiaries during the period of time any such property was owned, leased, subleased or licensed by or from the Company or its Subsidiaries. Except in compliance with Environmental Laws, the Company and its Subsidiaries have not disposed of, emitted, discharged, handled, stored, transported, used or released any Hazardous Materials, or arranged for the disposal, discharge, storage or release of any Hazardous Materials, or exposed any employee or other individual to any Hazardous Materials.

- (b) The Company, its Subsidiaries and their possession, use and operation of the Leased Real Property and their operation of the Company Business have complied in all material respects with all applicable Environmental Laws (which compliance includes, possession by the Company and its Subsidiaries of all Company Permits required in connection with the possession, use and operation of the Leased Real Property and their operation of the Company Business under all applicable Environmental Laws and the filing of all reports and notifications with applicable Governmental Entities and the maintenance of all records). To the Knowledge of the Company, no event has occurred which (i) would be reasonably likely to constitute or result in a violation by the Company or its Subsidiaries of, or a material failure on the part of the Company or its Subsidiaries to comply with, any applicable Environmental Laws; or (ii) would be reasonably likely to require the Company or its Subsidiaries to undertake, or bear all or any portion of the cost of, any corrective or other remedial or responsive action pursuant to any such applicable Environmental Laws.
- (c) No proceeding arising under or relating to any Environmental Laws is pending against the Company or its Subsidiaries, or, to the Knowledge of the Company, is threatened, against the Company, its Subsidiaries or any of the Leased Real Property. Neither the Company nor any of its Subsidiaries has received any written notice of any alleged claim, violation of or liability under any Environmental Law which has not heretofore been cured or for which there is any remaining liability.
- (d) The Company and its Subsidiaries have made available for inspection by Parent in the Virtual Data Room true, correct, and complete copies of all environmental audits and environmental assessments of any facility owned, leased or used at any time by the Company or each of its Subsidiaries in the possession or control of the Company or any of its Subsidiaries.
- (e) Except as disclosed in **Section 2.15** of the Company Disclosure Letter, neither the Company nor any of its Subsidiaries owns or operates, and, to the Knowledge of the Company, there is not present on the Leased Real Property: (A) any equipment or machinery that contains polychlorinated biphenyls or asbestos, (B) any underground or aboveground storage tanks, or (C) any surface impoundments (other than for storm water management), septic tanks, landfills, lagoons or mining or waste pits.

2.16 Contracts.

- (a) Material Contracts. For purposes of this Agreement, "Company Material Contract" shall mean:
- (i) any (A) Contract with any director, officer, employee, consultant, or independent contractor of the Company or any of its Subsidiaries, other than those Contracts that are terminable by the Company or any of its Subsidiaries for any reason or no reason on no more than thirty (30) days' notice without liability or financial obligation to the Company or its Subsidiaries, (B) indemnification or similar agreements between the Company and any Company Indemnified Parties or any other Person, and (C) Contract to or with any representative of a group of employees of the Company or any of its Subsidiaries relating to wages, hours, and other conditions or terms of employment;
- (ii) any Contract containing any covenant (A) whereby the Company or any of its Subsidiaries agrees not to engage in any line of business, compete with any Person in any line of business, or that purports to restrict the business activities of the Company or any of its Subsidiaries, including non-solicitation or similar covenants, (B) granting any exclusive distribution rights with respect to Company Products to a third party or containing any other exclusivity provisions, or (C) pursuant to which any party is required to (1) purchase or sell a stated portion of its requirements or output to another party or (2) purchase any minimum amount or volume of products or services, including any "take or pay" or similar provisions;
- (iii) any Contract relating to the disposition or acquisition by the Company or any of its Subsidiaries of a material amount of assets not in the ordinary course of business or pursuant to which the

Company or any of its Subsidiaries has any material ownership interest in any other Person or other business enterprise, including each Contract relating to the acquisition by the Company or any of its Subsidiaries of any operating business or the capital stock or all or substantially all of the assets of any other Person under which the Company or any of its Subsidiaries has any duty, liability, or obligation (including continuing indemnity obligations), or for the grant of option or preferential rights to purchase any properties, assets or operations of the Company or its Subsidiaries;

- (iv) any Contract under which the Company or any of its Subsidiaries have granted a third Person the right to sell or distribute any Company Products and have continuing material obligations to sell or distribute any Company Products and which may not be canceled by the Company or any of its Subsidiaries for any reason or no reason without penalty upon notice of ninety (90) days or less;
- (v) any Contract pursuant to which the Company or any of its Subsidiaries have continuing material obligations to develop any Intellectual Property or Intellectual Property Rights that will not be owned, in whole or in part, by the Company or any of its Subsidiaries and which may not be terminated by the Company or any of its Subsidiaries without penalty upon notice of ninety (90) days or less;
- (vi) any Contract containing any material support, maintenance or service obligation, or providing for the delivery of goods on the part of or to the Company or any of its Subsidiaries involving annual revenues, or in an amount or having a value, as appropriate, in excess of One Hundred Thousand Dollars (\$100,000.00) (including all statements of work, work orders, project addenda and other similar documents entered into in connection therewith), other than those Contracts that are terminable by the Company or any of its Subsidiaries for any reason or no reason on no more than ninety (90) days prior notice without any liability or financial obligation to the Company or its Subsidiaries;
- (vii) any joint venture, partnership, and other Contract (however named) involving a sharing of profits and losses by the Company or any of its Subsidiaries with any other Person;
- (viii) any Contract to which the Company or any Subsidiary is a party and pursuant to which any other Person is licensed any Company Intellectual Property Rights by the Company or any Subsidiary;
- (ix) any Contract to which the Company or any Subsidiary is a party and pursuant to which the Company or any Subsidiary is licensed Intellectual Property Rights that are owned by a third party (other than Contracts (A) consisting of "shrink wrap" or "execute-by-opening" or similar license agreements relating to commercially available, off-the-shelf computer software which do not require the payment of any licensing, maintenance, royalty, or other fee for the current level of use of such software by the Company or any of its Subsidiaries or upon the transfer of any such Contract, (B) that are entered into in connection with the purchase or use of equipment by the Company or any of its Subsidiaries and which are otherwise identified on Section 2.16(b) of the Company Disclosure Letter, and (C) Contracts entered into in the ordinary course of business consistent with past practices and involving payments by the Company or any of its Subsidiaries in an aggregate amount of less than One Hundred Thousand Dollars (\$100,000.00) per year);
 - (x) any Contract to which the Company or any of its Subsidiaries is a party with any Governmental Entity;
- (xi) any mortgage, indenture, guarantee, loan or credit agreement, security agreement or other Contracts to which the Company or any of its Subsidiaries is a party relating to the borrowing of money or extension of credit in a principal amount in excess of One Hundred Thousand Dollars (\$100,000.00) that is outstanding or may be incurred on the terms thereof, other than Contracts for the payment of accounts receivable or trade accounts payable in accordance with their respective terms entered into in the ordinary course of business consistent with past practices; and

- (Xii) any other Contract that involves annual payments to or from the Company or any of its Subsidiaries in excess of One Hundred Thousand Dollars (\$100,000.00) on its face in any individual case, other than those Contracts that are terminable by the Company or any of its Subsidiaries for any reason or no reason, on no more than ninety (90) days prior notice without any material liability or material financial obligation to the Company or its Subsidiaries.
- (b) *Schedule*. **Section 2.16(b)** of the Company Disclosure Letter sets forth a complete and accurate list of all Company Material Contracts to which the Company or any of its Subsidiaries is a party or is bound by as of the date hereof. Neither the Company nor any of its Subsidiaries is a party to any lease that constitutes a capital lease in accordance with GAAP.
- (c) No Breach. All Company Material Contracts are valid and in full force and effect, except (i) as enforcement may be limited by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other laws affecting the rights of creditors generally and general equitable principles (whether considered in a proceeding in equity or at law), and (ii) as the remedy of specific performance and injunctive and other forms of equitable relief may be subject to equitable defenses and to the discretion of a court of competent jurisdiction before which any proceeding may be brought. Neither the Company nor any of its Subsidiaries has violated any material provision of, or committed or materially failed to perform any act which, with or without notice, lapse of time or both would constitute a default under the provisions of, any Company Material Contract.
- 2.17 *Insurance*. As of the date of this Agreement, the Company and its Subsidiaries maintain the insurance policies set forth on **Section 2.17** of the Company Disclosure Letter. All premiums due and payable under all such policies have been paid, and the Company and its Subsidiaries are otherwise in compliance in all material respects with the terms of such policies applicable to them. To the Knowledge of the Company, there has been no threatened termination of, or premium increase with respect to, any such policy, except in accordance with the terms thereof.
- 2.18 *Order Backlog.* Section 2.18 of the Company Disclosure Letter lists all orders by third parties for Backlog that are (i) binding on Company; and (ii) unfulfilled as of August 31, 2013. The Backlog as of August 31, 2013 was as set forth on Section 2.18 of the Company Disclosure Letter. Section 2.18 of the Company Disclosure Letter also includes a list of all funnel opportunities, which list does not represent any assurance of future revenue for the Company, for the manufacture or sale of products by the Company or any of its Subsidiaries having an individual value in excess of One Hundred Thousand Dollars (\$100,000.00), which are outstanding as of August 31, 2013. Except as set forth on Section 2.18 of the Company Disclosure Letter, no customer of Company or any of its Subsidiaries has directly or indirectly paid to the Company or any of its Subsidiaries, nor has Company or any of its Subsidiaries otherwise directly or indirectly received from any Person, any unearned deposit, prepayment, advance payment or similar payment on account of any future purchase of products or services from the Company or any of its Subsidiaries that has not been appropriately accounted for as an unearned deposit, prepayment, advance payment or similar payment in accordance with GAAP and reflected in the Financial Statements.

2.19 Warranty Claims.

- (a) **Section 2.19(a)** of the Company Disclosure Letter sets forth, as of July 31, 2013, (i) the amounts for open warranty claims for Company Products and (ii) the amounts that the Company and its Subsidiaries hold in reserve for warranty claims.
- (b) Except as set forth on **Section 2.19(b)** of the Company Disclosure Letter, (i) there is no material liability or, to the Knowledge of the Company, any basis for any present or future proceeding against the Company or any of its Subsidiaries giving rise to any material liability, for the replacement or repair of any products that the Company or any of its Subsidiaries have sold, leased or produced at any time prior to the date hereof, for which an unexpired warranty obligation was outstanding as of August 31, 2013 (subject only to the reserve for

warranty claims set forth in **Section 2.19(a)** of the Company Disclosure Letter) and (ii) no claim, or series of related claims, in respect thereof in an amount in excess of Fifty Thousand Dollars (\$50,000) has been asserted by any Person during the three (3) year period immediately preceding the date of this Agreement. The Company's warranty reserve as set forth in **Section 2.19(a)** of the Company Disclosure Letter has been calculated utilizing historical warranty experience rates consistent with past practice and, to the Knowledge of the Company, is sufficient to cover the unexpired warranty liabilities of the Company and its Subsidiaries for any products (including Company Products) sold to their respective customers. Except as set forth on **Section 2.19(b)** of the Company Disclosure Letter, no product produced or sold by the Company or any of its Subsidiaries is subject to any express warranty nor any other warranty other than any warranties that are implied by applicable Legal Requirements.

- (c) Except as set forth on **Section 2.19(c)** of the Company Disclosure Letter, (i) there is no (A) material liability to the Company or any of its Subsidiaries or (B) to the Knowledge of the Company, basis for any present or future proceeding against the Company or any of its Subsidiaries that would reasonably be expected to result in any material liability, arising out of any injury to individuals or damage to property that occurred by reason of the ownership, possession or use of any product produced or sold by the Company or any of its Subsidiaries, and (ii) no claim in respect thereof has been asserted by any Person during the three (3) year period immediately preceding the date hereof.
- 2.20 *Certain Business Practices*. Neither the Company nor any of its Subsidiaries has, directly or indirectly, at any time since its inception, (a) used any funds for unlawful contributions, unlawful gifts, unlawful entertainment or other unlawful expenses, (b) made any unlawful payment to any foreign or domestic government officials or employees or to any foreign or domestic political parties or campaigns or (c) otherwise taken any action that would cause the Company or any of its Subsidiaries to be in violation of the Foreign Corrupt Practices Act of 1977, as amended, the Anti-Kickback Act of 1986 or any other applicable Legal Requirements of similar effect.
- 2.21 Export Control Laws. The Company and each of its Subsidiaries is and has at all times been in compliance with all applicable Export Control Laws. Without limiting the foregoing: (a) each of the Company and its Subsidiaries has obtained all material export licenses and other material approvals for its exports of products required by any Export Control Law and all such approvals and licenses are in full force and effect; (b) the Company and each of its Subsidiaries is in material compliance with the terms of such applicable export licenses or other approvals and (c) there are no pending or, to the Knowledge of the Company, threatened claims against the Company or any of its Subsidiaries by any Governmental Entity with respect to the failure of the Company or any of its Subsidiaries to hold such export licenses or other approvals or relating to the compliance or noncompliance by the Company or any of its Subsidiaries with any applicable Export Control Laws.
- 2.22 **Books and Records**. The Company has made available to Parent in the Virtual Data Room accurate and complete copies of (i) all stock records of the Company and each of its Subsidiaries, including their respective stock ledgers and copies of any stock certificates issued by the Company and each of its Subsidiaries; and (ii) all minutes and other records of all meetings and other proceedings (including any actions taken by written consent or otherwise without a meeting) of the Company Stockholders, the stockholders of each Subsidiary of the Company, the Board of Directors of the Company and each of its Subsidiaries and all committees thereof. The minute books of the Company and each of its Subsidiaries accurately and completely reflect all material corporate actions of the Company Stockholders, the stockholders of each Subsidiary of the Company, the board of directors of the Company and each of its Subsidiaries, as appropriate, and any committees thereof for the periods reflected therein.
- 2.23 *Bank Accounts; Powers of Attorney.* Section 2.23 of the Company Disclosure Letter sets forth a complete and correct list showing (a) all banks in which the Company or any of its Subsidiaries maintains a bank account or safe deposit box (collectively, "*Bank Accounts*"), together with, as to each such Bank Account, the account number, the names of all signatories thereof and the authorized powers of each such signatory and, with respect to each such safe deposit box, the number thereof and the names of all Persons having access thereto; and (b) the names of all Persons holding powers of attorney from the Company or any of its Subsidiaries

and a summary description of such powers of attorney, correct and complete copies of which have been made available to Parent in the Virtual Data Room.

2.24 Material Customers and Material Suppliers.

- (a) Section 2.24(a) of the Company Disclosure Letter sets forth an accurate, correct and complete list of the ten largest customers of the Company and its Subsidiaries, determined on the basis of sales revenues, for each of the calendar years ended December 31, 2012, December 31, 2011 and December 31, 2010, and for the seven month period ended July 31, 2013, and sets forth the revenue received by the Company and its Subsidiaries from each such customer during the applicable period. To the Knowledge of the Company, none of the customers listed on Section 2.24(a) of the Company Disclosure Letter has provided notice (whether written or verbal) to the Company or any of its Subsidiaries that it intends to stop or materially reduce the rate of such purchases.
- (b) Section 2.24(b) of the Company Disclosure Letter sets forth an accurate, correct and complete list of the ten largest suppliers of the Company and its Subsidiaries, determined on the basis of costs of items purchased, for each of the calendar years ended December 31, 2012, December 31, 2011 and December 31, 2010, and for the seven month period ended July 31, 2013, and sets forth (i) a general description of the raw materials, supplies, merchandise and other goods or services ordered by the Company and its Subsidiaries from each such supplier and (ii) the amounts paid by the Company and its Subsidiaries to each such supplier during the applicable period. To the Knowledge of the Company, none of the suppliers or vendors listed on Section 2.24(b) of the Company Disclosure Letter has provided notice (whether written or verbal) to the Company or any of its Subsidiaries that it will not continue to provide raw materials, supplies, merchandise and other goods to the Company or its Subsidiaries in a manner consistent with past practices or that it intends to stop or materially reduce the rate of such sales.
- (c) There is no Contract which provides that any Person will be the exclusive supplier of or distributor for the Company or any of its Subsidiaries for any particular product or product line, and none of the suppliers of the Company's or its Subsidiaries' cabinets or critical supply chain components is the sole supplier of such items to the Company or its Subsidiaries.

2.25 Receivables; Inventory.

- (a) All of the accounts receivable of the Company and its Subsidiaries as of July 31, 2013 (i) represent amounts receivable for goods actually delivered or services actually provided, (ii) have arisen from bona fide transactions in the ordinary course of business, and, except as set forth in **Section 2.25(a)** of the Disclosure Letter, to the Knowledge of the Company, are not subject to any material counterclaims or offsets, (iii) have been billed in the ordinary course of business pursuant to the standard credit terms of the Company and its Subsidiaries and (iv) will be fully collected in the ordinary course of business (without resort to legal proceedings), net of any reserves for doubtful accounts included in the Financial Statements.
- (b) That portion of the assets of the Company and its Subsidiaries that is comprised of (i) raw materials inventory consists of good and merchantable items of a quality and quantity useable in the ordinary course of business or (ii) finished goods inventory consists of items which are suitable for sale in the ordinary course of business and (iii) completed products manufactured for any particular customer pursuant to any Contract conform in all material respects to all applicable specifications and other terms and conditions required by any such Contract as of July 31, 2013, in each case except to the extent reserved for in the Company's inventory reserves as reflected in the Financial Statements.
- 2.26 *Disclosure*. No representation or warranty of the Company set forth in this Agreement and no statement of the Company set forth in the Company Disclosure Letter contains any untrue statement of a

material fact or omits to state a material fact necessary to make the statements herein or therein, in light of the circumstances in which they were made, not misleading.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Parent and Merger Sub represent and warrant to the Company as of the date of the Prior Agreement (and, accordingly, the phrases "as of the date hereof," "as of the date of this Agreement" or any similar terms as used in this **Article III** will be deemed to refer to the date of the Prior Agreement) and as of the Closing Date as follows:

- 3.1 *Organization; Standing and Power; Charter Documents*. Parent and Merger Sub are each a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation, have the requisite power and authority to own, lease and operate its properties and to carry on its business as now being conducted and is duly qualified and in good standing to do business in each jurisdiction in which the nature of its business or the ownership or leasing of its properties makes such qualification necessary other than in such jurisdictions where the failure to be so organized, existing and in good standing or so qualified, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect on Parent.
- 3.2 *Merger Sub*. The authorized capital stock of Merger Sub consists of 1,000 shares of common stock, par value \$1.00 per share, of which 1,000 shares are issued and outstanding. Parent is the sole stockholder of Merger Sub and is the legal and beneficial owner of all 1,000 issued and outstanding shares. Merger Sub was formed by counsel to Parent at the direction of Parent solely for purposes of effecting the Merger and the other transactions contemplated hereby. Except as contemplated by this Agreement, Merger Sub does not hold, nor has it held, any material assets or incurred any material liabilities nor has Merger Sub carried on any business activities other than in connection with the Merger and the transactions contemplated by this Agreement. All of the outstanding shares of capital stock of Merger Sub have been duly authorized and validly issued, and are fully paid and nonassessable and not subject to any preemptive rights.

3.3 Authority; Non-Contravention; Necessary Consents.

- (a) Authority. Each of Parent and Merger Sub has all requisite corporate power and authority to enter into this Agreement and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby by Parent and Merger Sub have been duly authorized by all necessary corporate action on the part of Parent and Merger Sub and no other corporate proceedings on the part of Parent or Merger Sub are necessary to authorize the execution and delivery of this Agreement or to consummate the Merger and the other transactions contemplated hereby, subject only to the filing of the Certificate of Merger pursuant to Delaware Law. This Agreement has been duly executed and delivered by Parent and Merger Sub and, assuming due execution and delivery by the Company, constitutes a valid and binding obligation of Parent and Merger Sub, enforceable against Parent and Merger Sub in accordance with its terms, except (i) as enforcement may be limited by bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other Legal Requirements affecting the rights of creditors generally and general equitable principles (whether considered in a proceeding in equity or at law), and (ii) as the remedy of specific performance and injunctive and other forms of equitable relief may be subject to equitable defenses and to the discretion of a court of competent jurisdiction before which any proceeding may be brought.
- (b) **Board Approval.** The Board of Directors of each of the Parent and Merger Sub have, by resolutions duly adopted at a meeting duly called and held or by written consent adopted in accordance with applicable Legal Requirements (which resolutions have not been subsequently rescinded or modified in any way) (i) determined the Merger to be advisable, and (ii) irrevocably approved this Agreement and the transactions

contemplated hereby, including the Merger. Parent has approved the Merger and this Agreement as the sole stockholder of Merger Sub.

- (c) Non–Contravention. The execution and delivery of this Agreement by Parent and Merger Sub does not, and performance of this Agreement by Parent and Merger Sub will not: (i) conflict with or violate the certificate of incorporation or bylaws of Parent or the certificate of incorporation or bylaws of Merger Sub; (ii) subject to compliance with the requirements set forth in Section 3.3(d), conflict with or violate any material Legal Requirement applicable to Parent or Merger Sub or by which Parent or Merger Sub or any of their respective properties is bound or affected; or (iii) result in any material breach of or constitute a material default (or an event that with notice or lapse of time or both would become a material default) under, or materially impair Parent's rights or alter the rights or obligations of any third party under, or give to others any rights of termination, material amendment, acceleration or cancellation of, or result in the creation of a material Lien on any of the properties or assets of Parent or Merger Sub pursuant to, any material Contract to which Parent or Merger Sub is a party the termination or breach of which would reasonably be expected to have a Material Adverse Effect on Parent.
- (d) *Necessary Consents*. No consent, approval, order or authorization of, or registration, declaration or filing with any Governmental Entity is required to be obtained or made by Parent in connection with the execution and delivery of this Agreement or the consummation of the Merger and other transactions contemplated hereby, except for the Necessary Consents.
- 3.4 *Compliance*. Parent and Merger Sub have complied in all material respects, and are in compliance in all material respects, with each Legal Requirement applicable to Parent or Merger Sub or by which Parent or Merger Sub or any of their respective businesses or properties is bound or affected. There is no material judgment, injunction, order or decree binding upon Parent or Merger Sub which has or would reasonably be expected to have the effect of prohibiting or materially impairing the business conducted by Parent or Merger Sub.
- 3.5 **Availability of Funds**. Parent currently has access to sufficient immediately available funds in cash or cash equivalents, and will at the Closing have sufficient immediately available funds, in cash, to pay the Merger Consideration and any other amounts to be paid by Parent or Merger Sub hereunder.

ARTICLE IV

CONDUCT BY THE COMPANY PRIOR TO THE EFFECTIVE TIME

- 4.1 Conduct of Business by the Company and its Subsidiaries.
- (e) *Ordinary Course*. During the period from the date of the Prior Agreement and continuing until the earlier of the termination of this Agreement pursuant to its terms or the Effective Time, the Company and each of its Subsidiaries shall, except as otherwise expressly contemplated by this Agreement or to the extent that Parent shall otherwise consent in writing (which consent shall not be unreasonably withheld or delayed), (i) carry on its business in the usual, regular and ordinary course, in substantially the same manner as heretofore conducted and in compliance with all applicable Legal Requirements, (ii) pay its debts and Taxes when due, pay or perform other material obligations when due and (iii) use commercially reasonable efforts to preserve intact its present business organization and customer and vendor relationships.
- (f) **Required Consent.** In addition, without limiting the generality of **Section 4.1(a)**, except as permitted or contemplated by the terms of this Agreement, and except as provided in **Section 4.1(b)** of the Company Disclosure Letter, without the prior written consent of Parent (which consent shall not be unreasonably withheld or delayed), during the period from the date of the Prior Agreement and continuing until the earlier of the termination of this Agreement pursuant to its terms or the Effective Time, the Company shall not do any of the following, and shall not permit any of its Subsidiaries to do any of the following:

(xviii) Cause, permit or propose any amendments to the Company Charter Documents or any of the Subsidiary Charter

Documents;

- (xix) Adopt a plan of complete or partial liquidation or dissolution;
- (XX) Declare, set aside or pay any dividends on or make any other distributions (whether in cash, stock, equity securities or property) in respect of any capital stock or other equity or ownership interests, or split, combine or reclassify any capital stock or other equity or ownership interests, or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for any capital stock or other equity or ownership interests, other than any such transaction by a wholly-owned Subsidiary of it that remains a wholly-owned Subsidiary of it after consummation of such transaction, in the ordinary course of business;
- (XXI) Purchase, redeem or otherwise acquire, directly or indirectly, any shares of its capital stock or the capital stock of its Subsidiaries, except repurchases of unvested shares in connection with the termination of the employment relationship with any employee, director, or consultant pursuant to stock option or purchase agreements in effect on the date hereof;
- (xxii) Issue, deliver, sell, authorize, pledge or otherwise encumber any shares of Capital Stock, or any securities convertible into shares of Capital Stock, or subscriptions, rights, warrants or options to acquire any shares of Capital Stock or any securities convertible into shares of Capital Stock, or enter into other Contracts of any character obligating it to issue any such securities or rights, other than: (A) issuances of Capital Stock upon the exercise or conversion of Company Options or Company Warrants in accordance with their respective terms and (B) grants of stock options to acquire Common Stock or restricted stock of the Company under the Company Option Plan at fair market value and in the ordinary course of business consistent with past practice;
- (XXIII) Acquire or agree to acquire by merging or consolidating with, or by purchasing any material equity or voting interest in or a material portion of the assets of, or by any other manner, any business or any Person or division thereof, or otherwise acquire or agree to acquire any assets which are material to the business of the Company or its Subsidiaries;
- (xxiv) Sell, lease, license, encumber or otherwise dispose of any properties or assets of the Company or its Subsidiaries except (A) the sale, lease or disposition (other than through licensing) of property or assets which are not material to the business of the Company and its Subsidiaries or (B) the sale or licensing of Company Products in the ordinary course of business;
- (XXV) Make any loans, advances or capital contributions to, or investments in, any other Person, other than: (A) loans or investments by it or a wholly-owned Subsidiary of it to or in it or any wholly-owned Subsidiary of it or (B) employee loans or advances made in the ordinary course of business which do not, in the aggregate, exceed Fifty Thousand Dollars (\$50,000.00);
- (XXVI) Except as required by GAAP as concurred in by its independent auditors and disclosed to, and approved by, Parent in writing, make any material change in its methods or principles of accounting since the date of the Company Balance Sheet;
- (xxvii) Make any Tax election or accounting method change that is reasonably likely to adversely affect in any material respect the Tax liability of the Company or any of its Subsidiaries, settle or compromise any material income Tax liability or consent to any extension or waiver of any limitation period with respect to material Taxes;
- (XXVIII) Except as specifically required by Legal Requirements, Company Benefit Plans or Contracts currently binding on the Company or its Subsidiaries, (A) increase in any manner the amount of compensation or fringe benefits of, pay any bonus to or grant severance or termination pay to any

executive officer of the Company or any Subsidiary of the Company, (B) make any material increase in or commitment to increase, in any material respect, any benefits provided under any Company Benefit Plan (including any severance plan), adopt or amend or make any commitment to adopt or amend any Company Benefit Plan or make any contribution, other than regularly scheduled contributions, to any Company Benefit Plan, (C) waive any stock repurchase rights, accelerate, amend or change the period of exercisability of Company Options, or reprice any Company Options or authorize cash payments in exchange for any Company Options, (D) enter into any employment, severance, termination or indemnification agreement with any executive officer of the Company or any Subsidiary of the Company or enter into any collective bargaining agreement or (E) enter into any agreement with any Employee the benefits of which are (in whole or in part) contingent or the terms of which are materially altered upon the occurrence of a transaction involving the Company of the nature contemplated by this Agreement; provided, however, that nothing herein shall be construed as prohibiting the Company from granting Company Options in accordance with clause (B) of Section 4.1(b)(v);

(XXIX) Grant any license (including a sublicense) under any material Company Intellectual Property Rights except in connection with the sale, licensing, or provision of Company Products in the ordinary course of business, consistent with past practice;

(XXX) Enter into or renew any Contracts containing any non-competition or exclusivity restrictions on the Company, and of its Subsidiaries, or the Company Business;

(xxxi) Terminate any Company Material Contract currently in effect other than in the usual, regular and ordinary course, in substantially the same manner as heretofore conducted;

(XXXII) Incur any indebtedness for borrowed money or guarantee any such indebtedness of another Person, issue or sell any debt securities or options, warrants, calls or other rights to acquire any debt securities of the Company or any of its Subsidiaries, guarantee any debt securities of another Person, enter into any "keep well" or other agreement to maintain any financial statement condition of any other Person (other than any whollyowned Subsidiary of it), other than in connection with the financing of ordinary course trade payables; or

(XXXIII) Agree to or commit to do any of the actions described in clauses (i) through (xv) of this **Section 4.1(b)** or take any action, or fail to take any action within its reasonable control, as a result of which any of the changes or events listed in **Section 2.6** would occur.

4.2 **Supplementation and Correction and Information**. Between the date of the Prior Agreement and the Closing Date, the Company will promptly correct and supplement the information set forth in the Company Disclosure Letter in order to cause such Company Disclosure Letter and the representations and warranties of the Company set forth in **Article II** to remain true, correct and complete in all material respects. The delivery by the Company to Parent of any corrections or supplements will not constitute an amendment to the Company Disclosure Letter unless Parent specifically consents to any such amendment in writing or such information relates to the disclosure of an action of the type referred to in **Section 4.1(b)** for which the Company received prior written consent from Parent.

ARTICLE V

ADDITIONAL AGREEMENTS

5.1 Required Stockholder Approval and Notification. On or after the third Business Day following the date of the Prior Agreement, the Company shall prepare and deliver to the Company's stockholders an Information Statement, in a form reasonably acceptable to Parent, and shall use commercially reasonable efforts, in accordance with Delaware Law and the Charter Documents, to obtain the adoption of this Agreement and the

approval of the Merger (collectively, the "Company Proposals") by stockholders of the Company who did not deliver Written Consents immediately following the execution of this Agreement.

5.2 Acquisition Proposals.

- (i) No Solicitation. From and after the date of the Prior Agreement, the Company agrees that neither it nor any of its Subsidiaries nor any of the officers and directors of it or its Subsidiaries shall, and that it shall cause its and its Subsidiaries' Employees, stockholders, agents and authorized representatives (including any investment banker, attorney or accountant retained by it or any of its Subsidiaries) not to (and shall not authorize any of them to) directly or indirectly: (i) solicit, initiate, knowingly encourage or knowingly facilitate any inquiries with respect to, or the making, submission or announcement of, any offer or proposal for an Acquisition Proposal; (ii) participate in any discussions or negotiations regarding, or furnish to any Person any nonpublic information with respect to, any Acquisition Proposal; (iii) engage in discussions with any Person with respect to any Acquisition Proposal, except as to the existence of these provisions; (iv) approve, endorse or recommend any Acquisition Proposal; or (v) enter into any letter of intent or similar document or any Contract or commitment contemplating any Acquisition Proposal. The Company and its Subsidiaries and their respective representatives will immediately cease any and all existing activities, discussions or negotiations with any third parties conducted heretofore with respect to any Acquisition Proposal.
- (j) Notification of Unsolicited Acquisition Proposals. Promptly after receipt of any Acquisition Proposal or any request for nonpublic information or inquiry which it reasonably believes would lead to an Acquisition Proposal, in each case received after the date of the Prior Agreement, the Company shall provide Parent with notice of the material terms and conditions of such Acquisition Proposal, request or inquiry, and the identity of the Person or group making any such Acquisition Proposal, request or inquiry.

5.3 Confidentiality; Access to Information; No Modification of Representations, Warranties or Covenants.

- (bb) Confidentiality. The parties acknowledge that the Company and EnerSys have previously executed a Nondisclosure Agreement dated April 29, 2013 (as amended by the letter agreement dated August 29, 2013, the "Confidentiality Agreement"), which Confidentiality Agreement will continue in full force and effect in accordance with its terms and each of Parent and the Company will hold, and will cause its respective directors, officers, employees, agents and advisors (including attorneys, accountants, consultants, bankers and financial advisors) to hold, any Confidential Information (as defined in the Confidentiality Agreement) confidential in accordance with the terms of the Confidentiality Agreement.
- (CC) Access to Information. The Company will afford Parent and Parent's accountants, counsel and other representatives reasonable access during normal business hours and upon reasonable notice to its and its Subsidiaries' properties, books, records and personnel during the period prior to the Effective Time to obtain all information concerning their respective business, properties, results of operations and personnel for purposes of this Agreement, as Parent may reasonably request; provided, however, that the Company may restrict the foregoing access to the extent that (i) any Legal Requirement, treaty, rule or regulation of any Governmental Entity applicable to the Company or its Subsidiaries requires the Company or its Subsidiaries to restrict or prohibit access to any such properties or information, or (ii) such access would (A) be in breach of any confidentiality obligation, commitment or provision by which the Company or any of its Subsidiaries is bound or affected, which confidentiality obligation, commitment or provision shall be disclosed to Parent, provided that disclosure of such obligation, commitment or provision would not itself be the breach of an obligation or commitment to a third party, or (B) result in a waiver of any legal privilege enjoyed by the Company. With respect to the exchange of competitively sensitive information, including strategic and marketing plans, pricing material and customer specific data, outside antitrust counsel will be consulted prior to the exchange of such information, and such information shall not be exchanged to the extent such counsel advises against such exchange. In addition, any information obtained from the Company or any Company Subsidiary pursuant to the access contemplated by this Section 5.3(b) shall be subject to the Confidentiality Agreement. Any access to any of the Company's offices shall be subject to the Company's reasonable security

measures, the requirements of the applicable Lease and insurance requirements and shall not include the right to perform any "invasive" testing.

- 5.4 **Public Disclosure.** Without limiting any other provision of this Agreement, Parent and the Company will provide each other the opportunity to review, comment upon and concur with, and agree on any press release with respect to this Agreement and the transactions contemplated hereby, including the Merger, and any Acquisition Proposal and will not issue any such press release prior to such consultation and agreement, except as may be required by Legal Requirements or any listing agreement with any applicable national or regional securities exchange or market. The parties shall agree to the text of the joint press release announcing the signing of this Agreement.
- 5.5 **Reasonable Efforts.** Subject to the terms and conditions provided in this Agreement, each of the parties hereto shall use commercially reasonable efforts to take, or cause to be taken, promptly, all actions, and to do, or cause to be done, promptly, all things reasonably necessary to consummate and make effective the transactions contemplated hereby, to satisfy the conditions to the obligations to consummate the Merger, to obtain all necessary waivers, consents, orders, authorizations and approvals and to effect all necessary registrations, declarations, filings, notices, petitions, statements, applications and submissions and to remove any injunctions or other impediments or delays, legal or otherwise, in order to consummate and make effective the transactions contemplated by this Agreement for the purpose of securing to the parties hereto the benefits contemplated by this Agreement.
- 5.6 *Regulatory Approvals*. To the extent applicable, as soon as may be reasonably practicable following the execution and delivery of this Agreement by the parties hereto, the Company and Parent shall make all filings, notices, petitions, statements, registrations and submissions of information, application or submission of other documents required by any Governmental Entity in connection with the Merger and the transactions contemplated hereby. Each of Parent and the Company shall cause each document that it is responsible for filing with any Governmental Entity under this Section 5.6 to comply in all material respects with applicable Legal Requirements. The Company and Parent shall use their respective commercially reasonable efforts to, and shall use their respective commercially reasonable efforts to cause any applicable Company Securityholder to, promptly (a) supply the others with any information that reasonably may be required in order to effectuate the filings contemplated by this Section 5.6, and (b) supply any additional information that reasonably may be required by the competition or merger control authorities of any other jurisdiction and that the parties hereto may reasonably deem appropriate. Except where prohibited by applicable Legal Requirements, each of the Company and Parent, as applicable, shall consult with the other prior to taking a position with respect to any such filing, shall permit the other to review and discuss in advance, and consider in good faith, the views of the other in connection with any analyses, appearances, presentations, memoranda, briefs, white papers, other materials, arguments, opinions and proposals before making or submitting any of the foregoing to any Governmental Entity in connection with any investigations or proceedings in connection with this Agreement or the transactions contemplated hereby, coordinate with the other in preparing and providing such information and, subject to entering into an appropriate joint defense agreement if either party reasonably so requests, promptly provide the other (and its legal counsel) copies of all filings, presentations and submissions (and a summary of oral presentations) made by such party with any Governmental Entity in connection with this Agreement and the transactions contemplated hereby. Parent and the Company shall cooperate with each other regarding the strategy for interacting with such Governmental Entities in connection with the matters contained in this Section 5.6. Each party hereto shall notify the other promptly upon the receipt of (i) any comments from any officials of any Governmental Entity in connection with any filings made pursuant hereto, and (ii) any request by any officials of any Governmental Entity for amendments or supplements to any filings made pursuant to, or information provided to comply in all materials respect with, applicable Legal Requirements. Whenever any event occurs that is required to be set forth in an amendment or supplement to any filing made pursuant to this Section 5.6, each party will promptly inform the other parties hereto of such occurrence, and each of the Company and Parent will cooperate with the other in filing with the applicable Governmental Entity such amendment or supplement. Without limiting the generality of the foregoing, each party hereto shall use commercially reasonable efforts to cause to be prepared and filed, as promptly as practicable following the date of this Agreement but in no event later than three (3) Business Days following the date of this Agreement, the

notifications required under the HSR Act in connection with the Merger. Each party shall elect early termination of the statutory waiting period under the HSR Act. Each party shall respond as promptly as reasonably practicable to (A) any inquiries or requests received from the Federal Trade Commission or the Department of Justice for additional information or documentation and (B) any inquiries or requests received from any state attorney general or other Governmental Entity in connection with antitrust or related matters. All filing and other fees, excluding any legal fees of counsel, required under the HSR Act shall be paid equally by the parties (i.e., fifty percent (50%) by Parent and fifty percent (50%) by the Company).

5.7 *Employee Matters*.

- (a) Following the Effective Time, Parent shall arrange for each participant in the Company Benefit Plans (including all dependents) (the "Company Participants") who becomes an employee of Parent, any Parent Subsidiary or the Surviving Corporation (or a dependent of such employee) to be eligible for the same employee benefits as those received by Parent employees with similar positions and responsibilities. Each Company Participant shall be given service credit for all purposes for his or her length of service with the Company (and its Subsidiaries and predecessors) prior to the Closing, including, without limitation, for eligibility to participate (provided that no retroactive contributions will be required), eligibility for vesting under Parent employee benefit plans and arrangements and vacation accrual. Parent shall cause any and all pre-existing condition (or actively at work or similar) limitations, eligibility waiting periods and evidence of insurability requirements under any Parent employee benefit plans and arrangements to be waived with respect to such Company Participants and shall provide them with credit for any co-payments, deductibles, and offsets (or similar payments) made during the plan year which includes the Closing Date for the purposes of satisfying any applicable deductible, out-of-pocket, or similar requirements under any Parent employee benefit plans or arrangements in which they are eligible to participate after the Closing Date.
- (b) Payment of Certain Severance Obligations. Parent agrees to assume, and/or cause the Surviving Corporation to assume the severance obligations payable to only those Employees of the Company set forth on Schedule 5.7(b) upon termination of each such Employee's employment pursuant to the terms of the written Contracts with each such Employee as in effect as of the date of this Agreement; provided, however, that in no event will Parent or the Surviving Corporation be obligated to pay (i) any such severance to the extent that such severance is comprised of, or determined based on, the Accrued Bonuses or (ii) any of the Severance Payments to any of the Other Participants. Parent also agrees to assume, and/or cause the Surviving Corporation to assume, any severance obligations payable to Employees of the Company who become employees of the Surviving Corporation (except the Other Participants) that would have been paid by the Company to such Employees in the ordinary course of business consistent with past practice upon the termination of each such Employee's employment, in accordance with the severance policies as set forth on Schedule 5.7(b) following the Closing.

5.8 *Indemnification*.

(a) For the period of six (6) years following the Effective Time, Parent will cause the Surviving Corporation to indemnify and hold harmless each person who is now, or has been at any time prior to the date of this Agreement or who becomes prior to the Effective Time, an officer or director of the Company or any of its Subsidiaries (the "Company Indemnified Parties") against all losses, claims, damages, costs, expenses, liabilities or judgments or amounts that are paid in settlement, in each case to the extent actually and reasonably incurred (the "Company Indemnified Liabilities") in connection with any claim, action, suit, proceeding or investigation by reason of the fact that such person is or was a director or officer of the Company or any of its Subsidiaries, or relating (directly or indirectly) to such person's duties as a director or officer of the Company or any of its Subsidiaries, whether pertaining to any matter existing or occurring at or prior to the Effective Time and whether asserted or claimed prior to, or at or after the Effective Time excluding, however, all Company Indemnified Liabilities based on, or relating to this Agreement or the transactions contemplated hereby (the "Company Indemnified Proceedings"), in each case to the full extent a corporation is permitted by applicable Legal Requirements to indemnify its own directors and officers, but only if and to the extent that such Company Indemnified Parties, in their capacity as an officer or director of the Company or its Subsidiaries, as appropriate, met

the applicable standard of conduct prescribed for indemnification under (i) the Company Charter Documents or the Subsidiary Charter Documents, as appropriate, (ii) applicable Legal Requirements, or (iii) the directors' and officers' or any similar liability insurance maintained by the Company or its Subsidiaries, as appropriate, in effect at the time of the act or omission giving rise to such claim for indemnification. In the event any Company Indemnified Party is or becomes involved in any Company Indemnified Proceeding, Parent shall, or shall cause the Surviving Corporation to, pay expenses in advance of the final disposition of any such Company Indemnified Proceeding to each Company Indemnified Party to the full extent permitted by law upon receipt of any undertaking contemplated by Section 145 of the DGCL. Without limiting the foregoing, in the event any such Company Indemnified Proceeding is brought against any Company Indemnified Party, (A) the Company Indemnified Parties may retain counsel of their choosing, (B) Parent shall, or shall cause the Surviving Corporation to, pay all reasonable and documented fees and expenses of one counsel for all of the Company Indemnified Parties with respect to each such Company Indemnified Proceeding unless there is, under applicable standards of professional conduct, a conflict on any significant issue between the positions of any two or more Company Indemnified Parties, in which case Parent shall pay the fees of such additional counsel required by such conflict, promptly as statements therefor are received; and (C) Parent and the Surviving Corporation will use all commercially reasonable efforts to assist in the defense of any such matters; provided, however, that neither Parent nor the Surviving Corporation shall be liable for any settlement of any claim effected without its written consent, which consent shall not be unreasonably withheld or delayed. Any Company Indemnified Party wishing to claim indemnification under this Section 5.8(a) upon becoming aware of any such Company Indemnified Proceeding shall promptly notify Parent and the Surviving Corporation (but the failure to so notify Parent or the Surviving Corporation shall not relieve Parent or the Surviving Corporation from any liability it may have under this Section 5.8(a) except to the extent such failure materially prejudices Parent or the Surviving Corporation), and shall deliver to Parent and the Surviving Corporation the undertaking contemplated by Section 145 of the DGCL.

- (b) For a period of six (6) years following the Effective Time, Parent will, and will cause the Surviving Corporation to, fulfill and honor in all respects the obligations of the Company pursuant to the indemnification agreements between the Company and the Company Indemnified Parties identified in **Section 2.16(b)** of the Company Disclosure Letter, subject to applicable Legal Requirements. For a period of six (6) years following the Effective Time, the Certificate of Incorporation and Bylaws of the Surviving Corporation will contain provisions with respect to exculpation and indemnification that are at least as favorable to the Company Indemnified Parties as those contained in the Certificate of Incorporation and Bylaws of the Company as in effect on the date hereof, which provisions will not be amended, repealed or otherwise modified in any manner that would adversely affect the rights thereunder of the Company Indemnified Parties.
- (C) For six (6) years after the Effective Time, Parent will, or will cause the Surviving Corporation to, maintain directors' and officers' liability insurance, or purchase a "tail" policy for such insurance, covering those Persons who are covered by the Company's directors' and officers' liability insurance policy as of the date hereof in an amount and on terms no less favorable than coverage under such policies in effect on the date of this Agreement and those applicable to the current directors and officers of the Company. The premium for any such insurance policy maintained by the Company will constitute a portion of the Closing Transaction Expenses payable out of the Merger Consideration at the Closing.
- (d) This **Section 5.8** is intended to be for the benefit of, and shall be enforceable by the Company Indemnified Parties and their heirs and personal representatives and shall be binding on Parent and the Surviving Corporation and its successors and assigns and shall be in addition to, and not in substitution for, any other rights to indemnification or contribution that any such person may have by contract or otherwise. In the event Parent or the Surviving Corporation or its successor or assign (i) consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or entity in such consolidation or merger or (ii) transfers all or substantially all of its properties and assets to any Person, then, and in each case, Parent or the Surviving Corporation, as appropriate, will use commercially reasonable efforts to cause proper provision to be made so that the successor and assign of Parent or the Surviving Corporation, as the case may be, agrees to honor the obligations set forth with respect to Parent or the Surviving Corporation, as the case may be, in this **Section 5.8**.

- 5.9 *Certain Tax Matters*. If the Company is obligated to make any payments, or is a party to any agreement that under certain circumstances could obligate it to make any payments, that will not be deductible under Section 280G of the Code if the stockholder approval requirements of Section 280G(b)(5)(B) are not satisfied (including those payments comprised of the Severance Payments and the Change of Control Payments but excluding the employer payroll portion of such Change of Control Payments and Severance Payments), then, on or prior to the Closing, the Company shall obtain and deliver to Parent a waiver from each other Person who is a party to such Contract pursuant to which such Person will waive his or her right to any such payments to the extent such payments constitute "parachute payments" within the meaning of the Code, unless such stockholder approval is obtained, and if that stockholder approval has not already been obtained, Parent agrees that it shall cooperate and assist the Company in obtaining the requisite stockholder approval described in Section 280G(b)(5)(B) of the Code, and the Company agrees to use commercially reasonably efforts to obtain such stockholder approval promptly after the date hereof and in any event prior to the Closing Date.
- 5.10 *Fees and Expenses*. Except as otherwise set forth herein, if the Merger is consummated, all fees and expenses incurred in connection with this Agreement and the transactions contemplated hereby, shall be the obligation of the respective party incurring such expenses. On behalf of the Company, at the Closing, the Parent shall pay the Closing Transaction Expenses that remain unpaid as of the Closing, including the portion of the Change of Control Payments outstanding as of the Closing, and the aggregate amount of the Merger Consideration payable to the Company Securityholders at the Closing will be reduced, on a Dollar-for-Dollar basis, by the amount of all such Closing Transaction Expenses paid by the Parent on behalf of the Company at Closing as provided in Section 1.7(b).
- 5.11 *FIRPTA Compliance*. On the Closing Date, the Company shall deliver to Parent a properly executed statement in a form reasonably acceptable to Parent for purposes of satisfying Parent's obligations under Treasury Regulation Section 1.1445-2(c)(3).
- 5.12 *Third Party Consents*. Except for the Necessary Consents and the consents set forth on **Schedule 6.1(d)** (to the extent not waived by Parent pursuant to **Section 8.4**), Parent acknowledges that certain consents and waivers with respect to the Merger may be required from parties to Contracts to which the Company is a party and that such consents and waivers have not been obtained. Parent agrees that the Company and its affiliates shall not have any liability whatsoever to Parent arising out of or relating to the failure to obtain any consents or waivers that may be required in connection with the Merger, other than any consent that would constitute a breach of **Section 2.3(d)** or **Section 1.6(d)**, or because of the termination of any Contract as a result thereof. Prior to the Closing, the Company shall cooperate with Parent, upon the request of Parent, in any reasonable manner in connection with Parent obtaining any such consents and waivers; provided that such cooperation shall not include any requirement of the Company or any of its affiliates to expend money, commence, defend or participate in any litigation or offer or grant any accommodation (financial or otherwise) to any third Person.
- 5.13 *Merger Sub Compliance*. Parent shall cause Merger Sub to comply with all of Merger Sub's obligations under or relating to this Agreement. Merger Sub shall not engage in any business which is not in connection with the Merger.

5.14 Tax Matters.

- (a) The Company shall prepare or cause to be prepared and file or cause to be filed all Tax Returns of the Company and its Subsidiaries for Pre-Closing Tax Periods that are required to be filed on or before the Closing Date and such Tax Returns shall be prepared and filed consistent with past practice, except as required by any Legal Requirement.
- (b) Parent shall prepare or cause to be prepared and file or cause to be filed all Tax Returns of the Company and its Subsidiaries for Pre-Closing Tax Periods (including any periods beginning before and ending after the Closing Date as allocated to the Pre-Closing Tax Period pursuant to the definition of "Unpaid Pre-Closing"

Taxes") that are required to be filed after the Closing Date and such Tax Returns shall be prepared and filed consistent with past practice, except as required by any Legal Requirement. To the extent that any such Tax Return could give rise to an indemnity claim under **Article VII** hereof, Parent shall submit such Tax Return to the Representative for review and approval at least fifteen (15) calendar days prior to the due date for filing such Tax Return (which approval shall not be unreasonably withheld, conditioned, or delayed).

- (C) Except to the extent required by any Legal Requirement, with respect to any Pre-Closing Tax Period, Parent shall not make any elections relating to the Company or file any amended Tax Return of the Company solely for the purpose of triggering an indemnity claim under **Article VII** hereof.
- (d) Parent and the Representative agree to furnish or cause to be furnished to each other, upon request, as reasonably promptly as practicable, such information (including access to books and records) and assistance relating to the Company as is reasonably necessary for the filing of any Tax Return (including any report required pursuant to applicable Legal Requirements), for the preparation of any audit, and for the prosecution or defense of any claim, suit or proceeding relating to any proposed adjustment. Parent and the Representative agree to use commercially reasonable efforts to retain or cause to be retained all books and records pertinent to the Company until the applicable period for assessment under applicable Legal Requirements (giving effect to any and all extensions or waivers) has expired, and to abide by or cause the abidance with all record retention agreements entered into with any Governmental Entity.

ARTICLE VI

CONDITIONS TO THE MERGER

- 6.1 *Conditions to the Obligations of Each Party to Effect the Merger*. The respective obligations of each party to this Agreement to effect the Merger shall be subject to the satisfaction at or prior to the Closing Date of the following conditions:
 - (k) Company Required Stockholder Approvals. The Required Stockholder Approvals shall have been obtained.
- (l) *No Order*. No Governmental Entity of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any statute, rule, regulation, executive order, decree, injunction or other order which (i) is in effect and (ii) has the effect of making the Merger illegal.
- (m) *Antitrust Approvals*. The waiting period (and any extension thereof) under the HSR Act relating to the transactions contemplated hereby shall have expired or terminated early and all foreign antitrust approvals listed on **Schedule 6.1(c)** hereto shall have been obtained.
- (n) *Third Party Consents*. The Company shall have delivered to Parent the consents set forth on **Schedule 6.1(d)** hereto, and such consents shall be in full force and effect.
- 6.2 Additional Conditions to the Obligations of the Company. The obligation of the Company to consummate and effect the Merger shall be subject to the satisfaction at or prior to the Closing Date of each of the following conditions, any of which may be waived, in writing, exclusively by the Company:
- (dd) *Representations and Warranties*. The representations and warranties of Parent and Merger Sub contained in this Agreement shall have been true and correct in all material respects (except for those representations and warranties which by their terms are subject to materiality qualifications, which representations and warranties shall be true and correct in all respects) as of the date of this Agreement and shall be true and correct in all material respects as of the Closing Date as if made on the Closing Date (except for those representations and warranties which address matters only as of a particular date which shall have been true and correct only on such

date). The Company shall have received a certificate to such effect signed on behalf of Parent, with respect to the representations and warranties of Parent, by an authorized executive officer of Parent and a certificate to such effect signed on behalf of Merger Sub, with respect to the representations and warranties of Merger Sub, by an authorized executive officer of Merger Sub.

- (ee) Agreements and Covenants. Parent and Merger Sub shall have performed or complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by it on or prior to the Closing Date (including the delivery to Parent of the waivers and consents, if any, required to be obtained from the holders of the Company Options pursuant to Section 1.6(b), and the Company shall have received a certificate to such effect signed on behalf of Parent, with respect to the covenants of Parent, by an authorized executive officer of Parent and a certificate to such effect signed on behalf of Merger Sub, with respect to the covenants of Merger Sub, by an authorized executive officer of Merger Sub.
- (ff) *Escrow Agreement*. Parent and Escrow Agent shall have executed and delivered to Representative the Escrow Agreement, substantially in the form attached hereto as **Exhibit B**, which Escrow Agreement shall be satisfactory to Representative in all respects.
- 6.3 Additional Conditions to the Obligations of Parent and Merger Sub. The obligations of Parent and Merger Sub to consummate and effect the Merger shall be subject to the satisfaction at or prior to the Closing Date of each of the following conditions, any of which may be waived, in writing, exclusively by Parent and Merger Sub:
- (c) Representations and Warranties. The representations and warranties of the Company contained in this Agreement shall have been true and correct in all material respects (except for those representations and warranties which by their terms are subject to materiality or Material Adverse Effect qualifications, which representations and warranties shall have been true and correct in all respects) as of the date of this Agreement and shall be true and correct in all material respects (except for those representations and warranties which by their terms are subject to materiality or Material Adverse Effect qualifications, which representations and warranties shall be true and correct in all respects) as of the Closing Date, as if made on the Closing Date (except for those representations and warranties which address matters only as of a particular date which shall have been true and correct only on such date). Parent shall have received a certificate to such effect signed on behalf of the Company, with respect to the representations and warranties of the Company, by an authorized executive officer of the Company.
- (d) *Agreements and Covenants*. The Company shall have performed or complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by it at or prior to the Closing Date, and Parent and Merger Sub shall have received a certificate to such effect signed on behalf of the Company by an authorized executive officer of the Company.
- (e) *No Material Adverse Effect*. During the period from the date of the Prior Agreement through the Closing Date, there shall not have occurred in any event, circumstance or change which has resulted in, or would be reasonably likely to result in, a Material Adverse Effect with respect to the Company that is continuing.
- (f) Closing Debt Amount and Closing Transaction Expenses. The Company shall have delivered to Parent (i) separate payoff letters in a form and substance reasonably satisfactory to Parent executed by each holder of any indebtedness of the Company or any of its Subsidiaries comprising a portion of the Closing Debt Amount, each of which payoff letters will (A) specify the aggregate amount of such indebtedness payable by Company or any of its Subsidiaries to such Person that is outstanding as of the Closing Date and (B) provide that, upon the payment of such amounts in accordance with the wire transfer instructions set forth in such payoff letters, all Liens (if any) in or on any of the assets of the Company or any of its Subsidiaries granted by the Company or any of its Subsidiaries as collateral security for such indebtedness will be terminated and released, (ii) a letter in form and substance reasonably satisfactory to the Parent executed by the Person which is the holder of the indebtedness of

the Company and its Subsidiaries which comprises the Post-Closing Debt Amount specifying the aggregate amount of such indebtedness (and all accrued unpaid interest thereon and all unpaid fees, costs and expenses relating thereto) as of the Closing Date, (iii) final invoices from any third party legal, accounting, tax, financial advisor or consultants to which any Closing Transaction Expenses are payable setting forth the respective amounts of the Closing Transaction Expenses payable to such Persons and a statement from such Persons indicating that upon the payment thereof no further payment is due except with respect to the Escrowed Transaction Expenses, and (iv) separate releases duly executed by each Person to which any Change of Control Payments or Severance Payments are payable setting forth the respective amounts of the Change of Control Payments or Severance Payments payable to such Persons (excluding the employer payroll tax portion of such Change of Control Payments or Severance Payments) and irrevocably releasing any rights or claims, if any, that such Persons could have against the Company or its Subsidiaries or the Surviving Corporation or Parent with respect to the portion of such Change of Control Payments or Severance Payments upon the payment thereof from the Merger Consideration at the Closing (excluding the employer payroll portion of such Change of Control Payments or Severance Payments), in each case in a form and substance reasonably satisfactory to Parent.

- (g) Estimated Closing Balance Sheet and Estimated Closing Statement. The Company shall have delivered the Estimated Closing Balance Sheet and the Estimated Closing Statement to Parent in accordance with the provisions of Section 1.6(h)(i), and Parent shall have reviewed and approved the Estimated Closing Balance Sheet and the Estimated Closing Statement, which approval shall not be unreasonably withheld, conditioned, or delayed.
- (h) *No Dissenting Shares*. No Company Stockholder shall have exercised and perfected appraisal or dissenters rights (and not effectively withdrawn or lost such appraisal or dissenters rights) for such shares in accordance with Section 262 of the DGCL; *provided*, *however*, that this condition shall be deemed satisfied if, prior to the Closing Date, Company provides Parent with reasonably satisfactory written evidence that 100% of the Company Stockholders shall have approved the Merger in accordance with the applicable provisions of the DGCL.
- (i) *Agreement Not To Compete*. Each Designated Person shall have executed and delivered to Parent an Agreement Not to Compete, in the form attached hereto as **Exhibit C**, each of which Agreements Not to Compete shall be satisfactory to Parent in all respects.
- (j) *Escrow Agreement*. The Representative and the Escrow Agent shall have executed and delivered to Parent the Escrow Agreement, substantially in the form attached hereto as **Exhibit B**, which Escrow Agreement shall be satisfactory to Parent in all respects.

ARTICLE VII

SURVIVAL OF REPRESENTATIONS AND WARRANTIES; INDEMNITY ESCROW

7.1 Survival of Representations and Warranties. All representations and warranties of the Company contained in this Agreement other than the Fundamental Representations (the "General Representations") shall survive the consummation of the Merger and continue until the expiration of the eighteen (18) month period immediately following the Closing Date (the "General Representations Termination Date"), after which time such representations and warranties shall terminate. The representations and warranties of the Company set forth in Sections 2.1(a) (Organization), 2.2 (Capitalization), 2.3(a) (Authority), 2.7 (Taxes), 2.8 (Intellectual Property), 2.11 (Brokers' and Finders' Fee), 2.12 (Employee Benefits Plans), 2.14 (Assets) and 2.15 (Environmental Matters) (collectively, the "Fundamental Representations") shall survive until the date that is the three (3) year anniversary of the Closing Date (the "Escrow Termination Date"), after which time such representations and warranties shall terminate. In the event of fraud with respect to this Agreement or any of the transactions contemplated hereby by any Company Securityholder, the Representative or any other Person or any breach of any provision of a Non-Competition Agreement by the Designated Person party thereto, all claims of the Indemnified Parties shall survive indefinitely with respect to any such fraud or breach committed by such Person.

Notwithstanding the foregoing, with respect to claims made pursuant to an Officer's Certificate delivered to the Escrow Agent and the Representative on or prior to the General Representations Termination Date for any breach of any General Representations and on or prior to the Escrow Termination Date for any breach of any Fundamental Representations, the representations and warranties with respect to such claims shall survive (but only for the purpose of pursuing such claims) until such claims are finally resolved pursuant to the Escrow Agreement. The agreements, covenants and other obligations of the parties hereto shall survive the Closing and the Effective Time in accordance with their respective terms or, absent such terms, in accordance with applicable Legal Requirements.

7.2 Escrow Amount Deposit; Representative Escrow Amount Deposit. Promptly following the Closing, and without any act of any stockholder of the Company, Parent shall, as provided in Section 1.7(b), pay and deposit from the Merger Consideration an amount of cash equal to the Escrow Amount and the Working Capital Escrow Amount with the Escrow Agent and an amount of cash equal to the Representative Escrow Amount with the Representative. For purposes of determining the reduction in the amount of the Merger Consideration otherwise payable to each Company Securityholder pursuant to Section 1.6(a) through Section 1.6(c), as applicable, Parent will be deemed to have contributed on behalf of each Company Securityholder, his, her or its (a) Pro Rata Portion of the Adjusted Escrow Amount and the Adjusted Working Capital Escrow Amount to the Escrow Fund to be governed under the terms set forth in this Agreement and the Escrow Agreement and (b) Pro Rata Portion of the Adjusted Representative Escrow Amount to the Representative Fund. The parties hereto agree that, for Tax purposes, Parent is the owner of any cash in the Escrow Fund and Working Capital Fund and the Company Securityholders are the owners of any cash in the Representative Fund. All interest on or other taxable income, if any, earned from the investment of cash in the Escrow Fund and Working Capital Fund pursuant to this Agreement or the Escrow Agreement shall be treated for Tax purposes as earned by Parent. All interest on or other taxable income, if any, earned from the investment of cash in the Representative Fund pursuant to this Agreement or the Representative Agreement shall be treated for Tax purposes as earned by the Representative. Upon release of the cash in the Escrow Fund and Working Capital Fund, a portion of any cash distributed to the Company Securityholders, Lazard and Other Participants shall be treated as interest under the imputed interest rules of the Code.

7.3 Indemnification From Escrow Fund and Working Capital Fund; Representative Fund.

(n) Subject to the terms and conditions of this Agreement and the Escrow Agreement, the Indemnified Parties shall be entitled to recover from the Escrow Fund, without duplication, the amount of any Losses suffered or incurred by each such Indemnified Party resulting from (i) any breach of any representation or warranty of the Company contained in Article II of this Agreement (as modified by the Company Disclosure Letter except as otherwise provided in Section 4.2) (except for breaches of the representations and warranties of the Company set forth in Section 2.19 (Warranty Claims), which are indemnifiable and recoverable from the Escrow Fund pursuant to clause (iv) of this Section 7.3(a)), and in each case read without regard to any qualifications or limitations to any of the Company's representations or warranties based upon materiality or Material Adverse Effect, other than the representation and warranty set forth in Section 2.6(a), (ii) any failure by (A) the Company to perform or comply with any covenant of the Company contained in this Agreement prior to the Closing Date or (B) the Representative to perform or comply with any covenant of the Representative contained in this Agreement, (iii) any Unpaid Pre-Closing Taxes and (iv) any (A) breach of any representation or warranty of the Company contained in Section 2.19 (Warranty Claims), in each case read without regard to any qualifications or limitations to any of such representations or warranties of the Company based upon materiality and (B) claim asserted by any Person at any time prior to the Escrow Termination Date alleging, in whole or in part, that any product (including any Company Product) which was sold or shipped by the Company or any of its Subsidiaries (or any of their respective predecessors in interest) prior to the Closing Date is defective or fails to conform in any respect with any guarantee or warranty (whether express or implied) regarding the workmanship, materials, quality, performance or specifications of any such products, including any claims for the repair or replacement of such products or for a refund of the cost thereof (collectively, "Warranty Claims"). Subject to the terms and conditions of this Agreement and the Escrow Agreement, Parent shall be entitled to recover from the Working Capital Fund the Merger Consideration Excess Amount, if applicable. Subject to the terms and conditions of this Agreement, the Representative Fund shall be available to the Representative to cover (I) the fees of the Independent Accountant pursuant to Section 1.6(h)(v) if applicable, (II) the Representative Losses and expenses incurred by the

Representative and (III) the expenses and fees of counsel or other skilled professionals for the Representative if it elects to undertake the defense of any Third Party Claim in accordance with **Section 7.7**.

- (0) Notwithstanding anything to the contrary contained herein, the Escrow Fund shall not be available to compensate any of the Indemnified Parties for any Losses indemnifiable pursuant to Section 7.3(a)(i) for breaches of (i) the General Representations or (ii) those Fundamental Representations set forth in Section 2.7 (Taxes) and Section 2.8 (Intellectual Property), and no Indemnified Party shall be entitled to recover from the Escrow Fund for such breaches, unless and until the Indemnified Parties have incurred Losses indemnifiable pursuant to Section 7.3(a)(i) for breaches of the General Representations or breaches of those Fundamental Representations set forth in Section 2.7 (Taxes) and Section 2.8 (Intellectual Property), in excess of Two Hundred Fifty Thousand Dollars (\$250,000.00) in the aggregate (the "Basket Amount"); provided, however, that no Indemnified Party shall be entitled to recover from the Escrow Fund for any Losses for any Warranty Claims unless and until the Indemnified Parties have incurred aggregate Losses with respect to such Warranty Claims in excess of the amount set forth on Schedule 7.3(b)(i) during any applicable Warranty Claim Period (the "Warranty Deductible"). Once the total of all Losses indemnifiable pursuant to Section 7.3(a)(i) incurred by the Indemnified Parties with respect to breaches of the General Representations or the Fundamental Representations set forth in Section 2.7 (Taxes) and Section 2.8 (Intellectual Property) exceeds the Basket Amount, the Indemnified Parties shall be entitled to recover from the Escrow Fund for the entire amount of all such Losses incurred (including those taken into account in determining the Basket Amount). With respect to Warranty Claims, the Indemnified Parties shall be entitled to recover from the Escrow Fund only the amount of Losses indemnifiable pursuant to Section 7.3(a)(iv) in excess of the Warranty Deductible for each separate Warranty Claim Period and, accordingly, all Losses for Warranty Claims sustained, incurred or accrued during any Warranty Claim Period shall be aggregated for purposes of determining whether the Warranty Deductible has been exceeded for such Warranty Claim Period. All Losses attributable to Warranty Claims shall (A) be determined without regard to any general or specific reserves for warranty claims included in the Financial Statements or otherwise reflected on the Company's books of account and (B) also include Losses incurred in connection with the matters set forth on Schedule 7.3(b)(ii). The Indemnified Parties shall be entitled to indemnification for Losses attributable to Warranty Claims regardless of the accuracy of the representations and warranties of the Company set forth in Section 2.19. Despite anything to the contrary set forth in this Article VII, the limitations set forth in this Section 7.3(b) shall not apply in the event of fraud or any Losses resulting from a breach of any of the Fundamental Representations (other than the Fundamental Representations set forth in Section 2.7 (Taxes) and Section 2.8 (Intellectual Property)) or any claims for indemnifiable Losses pursuant to Section 7.3(a)(ii) or Section 7.3(a)(iii).
- (p) Notwithstanding anything to the contrary contained herein: (i) in the event that any Losses which an Indemnified Party is otherwise entitled to recover from the Escrow Fund (A) are covered by insurance which was heretofore or is hereafter maintained by or for the benefit of the Indemnified Party or (B) result in a current, cash Tax benefit to the Indemnified Party, the amount of any "Losses" that the Indemnified Party would otherwise be entitled to recover from the Escrow Fund will be reduced by the amount of (1) any insurance proceeds actually received by, or actually paid to a third party for the benefit of, the Indemnified Party less (I) any reasonable costs of the collection of such insurance and (II) any actual premium increase attributable to the claim with respect to which such insurance proceeds are paid and (2) any such current, cash Tax benefit actually realized by the Indemnified Party, as appropriate; (ii) no Losses shall be recoverable from the Escrow Fund hereunder that would have been avoided through commercially reasonable efforts to mitigate such Losses which were not taken by Parent, the Surviving Corporation and/or the Indemnified Parties; provided, however, that notwithstanding the foregoing provisions of this Section 7.3(c), in no event shall any Indemnified Party be obligated or required to take any action to pursue any insurance or realize any current, cash Tax benefit with respect to any Losses suffered or incurred by the Indemnified Party as a condition to recovering any amounts from the Escrow Fund for or on account of such Losses or otherwise; and (iii) the amount of any Loss subject to recovery under this Article VII shall be calculated net of any amounts specifically accrued or reserved for in the final Closing Balance Sheet with respect to the fact or matter giving rise to the claim resulting in such Losses.
- (q) All claims for recovery for any Loss or Losses from the Escrow Fund and the Working Capital Fund shall be made pursuant to and in accordance with, and be governed by the terms of, this Agreement

and the Escrow Agreement. All claims for recovery from the Representative Fund shall be made in accordance with, and governed by the terms of, this Agreement and the Representative Agreement.

- (r) The Indemnified Parties' right to indemnification based on representations, warranties, covenants and agreements in this Agreement shall not be affected by any knowledge (whether actual or constructive) acquired, or capable of being acquired by any of the Indemnified Parties or their respective representatives from any source (including the Virtual Data Room), at any time, whether before or after the execution and delivery of this Agreement or the Closing Date, with respect to the accuracy or inaccuracy of, or compliance with, any such representation, warranty, covenant or agreement. The waiver of any condition based on the accuracy of any such representation or warranty, or on the performance of or compliance with any such covenant or agreements, will not affect the right to indemnification or any other remedy based on such representations, warranties, covenants and agreements.
- 7.4 Exclusive Remedy. Recovery from the Escrow Fund shall be the sole and exclusive remedy of the Indemnified Parties for any Losses incurred by any Indemnified Party or otherwise arising in connection with this Agreement except for the matters recoverable from the Working Capital Fund as set forth in Section 7.3(a) and as otherwise hereinafter provided in this Section 7.4. If the Merger Consideration Excess Amount exceeds the Working Capital Escrow Amount, the Company Securityholders and Other Participants shall, severally and not jointly, make the maximum contribution to the payment and satisfaction of such amount; provided that no Company Securityholder or Other Participant's obligation hereunder shall exceed such Company Securityholder's or Other Participants pursuant to this Agreement and the Change of Control Agreements. Except for fraud and for breaches by a Designated Person of his or its Non-Competition Agreement, in which case the Indemnified Parties shall have the right to proceed directly against such Person committing such fraud or breach of such Non-Competition Agreement, no Company Securityholder or Other Participant shall be liable to Parent, the Surviving Corporation or any Indemnified Party for any damages or other remedies arising out of, in connection with or related to this Agreement or the transactions contemplated hereby or any representation or warranty, covenant or agreement contained herein.

7.5 Distribution of Escrow Fund, Working Capital Fund and Representative Fund.

- (c) Subject to the terms and conditions of the Escrow Agreement, the Escrow Fund shall promptly be delivered to the Company Securityholders, Other Participants and Lazard on the Escrow Termination Date; *provided*, *however*, that the Escrow Fund shall not terminate with respect to any amount which is necessary to satisfy any unsatisfied claims specified in any Officer's Certificate delivered to the Escrow Agent and the Representative (i) on or prior to the General Representations Termination Date with respect to claims for breaches of the General Representations or (ii) the Escrow Termination Date with respect to (A) claims for breaches of the Fundamental Representations, (B) claims for breach of covenants pursuant to Section 7.3(a) (ii), (C) claims for Unpaid Pre-Closing Taxes pursuant to Section 7.3(a)(iii) and (D) Warranty Claims (except for claims pursuant to Section 2.19 which shall terminate pursuant to Section 7.5(a)(i)), pursuant to Section 7.3(a)(iv), as applicable. As soon as all such claims have been resolved, the Escrow Agent shall deliver the remaining portion of the Escrow Fund (if any) not required to satisfy such claims to the Company Securityholders, the Other Participants and Lazard in accordance with the provisions of the Escrow Agreement.
- (d) Subject to the terms and conditions of the Escrow Agreement, the Working Capital Fund (if any) shall promptly be delivered to the Company Securityholders and Other Participants in accordance with the provisions of the Escrow Agreement following (i) the expiration of the thirty (30) Business Day period immediately following the Closing Date if the parties shall have agreed that the Estimated Closing Statement Sheet shall have been accurate in accordance with **Section 1.6(h)** or (ii) the payment of the Merger Consideration Excess Amount or Merger Consideration Shortfall Amount, as applicable, if the Estimated Closing Statement Sheet shall have not been accurate in accordance with **Section 1.6(h)**.

- (e) Upon the completion of the Representative's duties hereunder, the Representative shall wire any remaining amount in the Representative Fund to the Escrow Agent and direct the Escrow Agent to distribute such amounts to the Company Securityholders, Other Participants and Lazard in accordance with the provisions of the Representative Agreement and the Escrow Agreement.
- (f) If and only to the extent that the Company Securityholders become entitled to receive funds from the Escrow Fund, the Working Capital Fund or the Representative Fund in accordance with Sections 7.5(a), 7.5(b) or 7.5(c), respectively, deliveries from the Escrow Fund, Working Capital Fund and Representative Fund to the Company Securityholders pursuant to this Section 7.5, the Escrow Agreement and the Representative Agreement shall be made in proportion to their respective Pro Rata Portion of the remaining Adjusted Escrow Amount, remaining Adjusted Working Capital Escrow Amount and remaining Adjusted Representative Escrow Amount, with each amount rounded to the nearest whole cent (\$0.01). Deliveries of the Escrowed Transaction Expenses from the Escrow Fund, the Escrowed Working Capital Transaction Expenses from the Working Capital Fund and the Escrowed Representative Transaction Expenses from the Representative Fund to the Other Participants and Lazard, as applicable pursuant to this Section 7.5, shall be made in accordance with the Escrowed Agreement and the agreements with each respective Other Participant and Lazard, as applicable, with each amount rounded to the nearest whole cent (\$0.01).

7.6 Representative.

- (e) **Designation of Representative**; **Authority and Rights of Representative**. By virtue of the approval and adoption of this Agreement by the requisite vote of the Company Securityholders, each of the Company Securityholders shall be deemed to have agreed to irrevocably appoint Fortis Advisors LLC, a Delaware limited liability company, as the initial Representative to be its agent and attorney-in-fact for and on behalf of the Company Securityholders to (i) give and receive notices and communications, (ii) to object to claims by an Indemnified Party for the release of any amount from the Escrow Fund or by Parent for the release of any amount from the Working Capital Escrow Fund in accordance with the applicable provisions of the Escrow Agreement in satisfaction of claims against the Escrow Fund under this Article VII by any Indemnified Party or against the Working Capital Escrow Fund under **Section 1.6(h)** by Parent, (iii) to agree to, negotiate, enter into settlements and compromises of, and comply with orders of courts with respect to, such claims against the Escrow Fund or Working Capital Escrow Fund, in each case relating to this Agreement or the transactions contemplated hereby, (iv) to do or refrain from doing any further act or deed on behalf of the Company Securityholders which the Representative deems necessary or appropriate in his, her or its sole discretion relating to the subject matter of this Agreement and the Escrow Agreement as fully and completely as any of such Company Securityholder could do if personally present and acting on their own behalf and (v) to take all other actions that are either necessary or appropriate in the judgment of the Representative for the accomplishment of the foregoing or specifically mandated by the terms of this Agreement or the Escrow Agreement. Notwithstanding the foregoing, the Representative (A) cannot consent to any term providing direct liability or obligation of a Company Securityholder and (B) shall have no obligation to act on behalf of the Company Securityholders, except as expressly provided herein and in the Escrow Agreement. All of the immunities and powers granted to the Representative and the Advisory Group under this Agreement shall survive the resignation or removal of Representative or any member of the Advisory Group and the Closing and/or any termination of this Agreement and the Escrow Agreement. The grant of authority to the Representative provided for in this Agreement (1) is coupled with an interest and shall be irrevocable and survive the death, incompetence, bankruptcy or liquidation of the respective Company Securityholder and shall be binding on any successor, heir or personal representative thereto and (2) shall survive the delivery of an assignment by any Company Securityholder of the whole or any fraction of his, her or its interest in the Escrow Fund, the Working Capital Fund or the Representative Fund. The Representative shall be permitted to communicate with the Company Securityholders, including in electronic form.
- (f) **Replacement of Representative**. The Representative may be changed by the Company Securityholders upon the written approval of the holders of a majority in interest of the Pro Rata Portions of the Adjusted Escrow Amount from time to time and upon prior written notice to the Parent; *provided, however*, that the Representative may not be removed unless holders of a two-thirds interest of the Adjusted Escrow Amount agree in

writing to such removal and to the identity of the substituted agent. Notwithstanding the foregoing, the Representative may resign at any time by providing written notice of intent to resign to the Company Securityholders and the Parent, which resignation shall be effective upon the earlier of (i) thirty (30) calendar days following delivery of such written notice or (ii) the appointment of a successor by the holders of a majority in interest of the Adjusted Escrow Amount. No bond shall be required of the Representative. Any successor Representative shall be deemed automatically to be the "Representative" under both this Agreement and the Escrow Agreement. If for any reason there is no Representative at any time, all references herein to the Representative shall be deemed to refer to the designee of the holders of a majority in interest of the Pro Rata Portions of the Adjusted Escrow Amount.

(g) Representative Fund. The Representative Escrow Amount shall be wired to the Representative by Parent upon the Closing of the Merger and shall be held by the Representative as agent and for the benefit of the Company Securityholders in a segregated client bank account and shall be used for the purposes set forth in Section 7.3(a), including paying directly, or reimbursing the Representative for, any third party expenses (including counsel, accountants or other experts retained by the Representative) or Representative Losses incurred pursuant to this Agreement, the Escrow Agreement or the Representative Agreement. The Representative will hold these funds separate from its corporate funds, will not use these funds for its operating expenses or any other corporate purposes and will not voluntarily make these funds available to its creditors in the event of bankruptcy. The interest or other earnings on the Representative Escrow Amount shall be irrevocably assigned to the Representative. The Company Securityholders acknowledge that the Representative is not providing any investment supervision, recommendations or advice. The Representative shall have no responsibility or liability for any loss of principal of the Representative Escrow Amount other than as a result of its willful misconduct or bad faith. For tax purposes, the Representative Escrow Amount shall be treated as having been received and voluntarily set aside by the Company Securityholders at the time of Closing. The parties agree that the Representative is not acting as a withholding agent or in any similar capacity in connection with the Representative Fund. Each Company Securityholder agrees that Representative Losses may be deducted by the Representative from the Representative Fund, subject to the prior approval by the Advisory Group as set forth in the Representative Agreement.

(h) Limitation of Liability; Indemnification. Certain Company Securityholders have entered into the Representative Agreement with the Representative to provide direction to the Representative in connection with the performance of its services under this Agreement, the Escrow Agreement and the transactions contemplated hereby (such Company Securityholders hereinafter referred to as the "Advisory Group"). To the maximum extent permissible by applicable Legal Requirements, neither the Representative nor any member of the Advisory Group shall incur any liability of any kind to any Company Securityholder or any other Person with respect to any action or inaction taken or failed to be taken, by it or by its agents, in connection with its services as the Representative or member of the Advisory Group, except with respect to its own willful misconduct or gross negligence. The Representative and the Advisory Group may act in reliance upon any signature believed by it to be genuine and may reasonably assume that such person has proper authorization to sign on behalf of the applicable Company Securityholder or other party. In all questions arising under this Agreement, the Escrow Agreement or the transactions contemplated hereby, the Representative or any member of the Advisory Group may rely on the advice of counsel, accountants or other skilled persons, and the Representative will not be liable to any Company Securityholder or any other Person for anything done, omitted or suffered in good faith by the Representative based on such advice of counsel, accountants or other skilled persons, as the case may be. No provision of this Agreement, the Escrow Agreement or any of the transactions contemplated hereby shall require the Representative to expend or risk its own funds or otherwise incur any financial liability in the exercise or performance of any of its powers, rights, duties or privileges under this Agreement or any of the transactions contemplated hereby. Furthermore, the Representative shall not be required to take any action unless the Representative has been provided with funds, security or indemnities which, in its determination, are sufficient to protect the Representative against the costs, expenses and liabilities which may be incurred by the Representative in performing such actions. The Representative and its members, managers, directors, officers, agents and employees shall be indemnified, defended and held harmless by the Company Securityholders as set forth in this Section 7.6(d) from and against any and all Losses, claims, damages, liabilities, fees, costs, expenses (including reasonable legal fees and disbursements and costs and including costs incurred in connection with seeking recovery from insurers), judgments, fines or amounts

paid in settlement ("*Representative Losses*") arising from, based upon or with respect to the Representative's execution and performance of this Agreement, the Escrow Agreement or any of the transactions contemplated hereby, or otherwise in connection with acting as the Representative (so long as the Representative was acting in good faith in connection therewith), in each case as such Representative Loss is incurred. Any such Representative Losses shall be recovered first from the Representative Fund, second from any distribution of the Escrow Fund otherwise distributable to the Company Securityholders pursuant to the terms hereof and the Escrow Agreement at the time of distribution in accordance with written instructions delivered by the Representative to the Escrow Agent, and third directly from the Company Securityholders, severally and not jointly, in proportion to their Pro Rata Portion of the Adjusted Representative Escrow Amount. Notwithstanding anything in this Agreement to the contrary, none of Parent, the Company or any of their respective Affiliates shall have any liability or obligation to indemnify, defend or hold harmless the Representative.

(i) To the extent the Representative receives documents, spreadsheets or other forms of information from any party and the Representative is required to deliver any such document, spreadsheet or other form of information to another party, the Representative is not responsible for the content of such materials, nor is the Representative responsible for confirming the accuracy of any information contained in such materials or reconciling the content of any such materials with any other documents, spreadsheets or other information. The Representative shall not be liable to any Company Securityholder for any apportionment or distribution of payments made by it in good faith, and if any such apportionment or distribution is subsequently determined to have been made in error, the sole recourse of any Company Securityholder to whom payment was due, but not made, shall be to recover from other Company Securityholders any payment in excess of the amount to which they are determined to have been entitled pursuant to this Agreement; provided, however, that the foregoing recourse limitation shall not apply in any case where such error was due to fraud or willful misconduct by the Representative. The Representative shall be entitled to rely upon Schedule 1.6(d) setting forth any apportionment or distribution of payments required to be made pursuant to this Agreement or the Escrow Agreement.

7.7 Third Party Claims.

(i) In the event that an Indemnified Party becomes aware of a third party claim, excluding, however, any Warranty Claims of the type described in Section 7.3(a)(iv)(B) (which Warranty Claims will be administered as provided in Section 7.8) (a "Third Party Claim"), that the Indemnified Party reasonably believes may result in a demand against the Escrow Fund or for other indemnification pursuant to this Article VII, Parent shall promptly notify the Representative of such claim. The Representative may, at his or her election, undertake and conduct the defense of such Third Party Claim, but only if and to the extent that the relief or remedy requested by the claimant in such Third Party Claim is limited solely to monetary damages and such Third-Party Claim does not include a claim for any equitable, injunctive, or other non-monetary relief or remedies. If the Representative elects to undertake the defense of any Third Party Claim, the Representative (i) shall use counsel of its choice but reasonably acceptable to Parent, (ii) shall be fully responsible for the outcome thereof, and (iii) shall be solely responsible for payment of all costs and expenses of its counsel; provided, however, that notwithstanding the foregoing, the Representative's right to assume and control the defense of any Third Party Claim is conditioned upon (A) the Representative acknowledging to the Indemnified Party, in writing and without qualification or limitation (other than the applicable limitations (if any) set forth in Sections 7.2 and 7.4), the Indemnified Party's right to recover all Losses relating to such Third Party Claim from the Escrow Fund and (B) the amount to fully pay all costs of defending such Third Party Claim, including the costs and expenses of its counsel, not exceeding the amount then held in the Representative Fund. Notwithstanding the Representative's election to undertake the defense of such Third Party Claim, the Indemnified Party shall nevertheless be entitled to recover from the Escrow Fund the amount of all Losses consisting of costs and expenses incurred by the Indemnified Party in connection with the response to, or defense of, such Third Party Claim prior to the assumption of such defense by the Representative to the extent recoverable pursuant to Section 7.3(a) hereof, but the Representative shall not be liable to Parent or any other Indemnified Party for legal expenses subsequently incurred by Parent or any other Indemnified Party, as applicable, in connection with the defense thereof, unless such legal expenses are incurred as a result of the Representative's failure to defend such Third Party Claim after agreeing to assume the defense thereof.

- (30) days after the date of the Representative's receipt of Parent's notice of the existence of such Third Party Claim. If the Representative elects to undertake the defense of any Third Party Claim, Parent shall be entitled to participate in, but not to determine or conduct, the defense of such Third Party Claim. If the Representative does not, within thirty (30) days after the date of its receipt of Parent's notice of the existence of a Third Party Claim, give notice to the Parent of its assumption of the defense of such Third Party Claim, the Representative shall be deemed to have irrevocably waived its right to control the defense thereof and Parent or the Indemnified Party shall have the right to control the defense of such Third Party Claim in such manner as it may deem appropriate and shall have the right to recover all Losses relating or attributable to the defense of such Third Party Claim from the Escrow Fund to the extent recoverable pursuant to Section 7.3(a) hereof. The party not controlling the defense of such claim may participate therein at its own expense.
- (k) Neither Representative nor Parent shall have the right to settle or compromise any such claim except, in the case of a settlement or compromise by the Representative, with the prior written consent of Parent (which consent will not be unreasonably withheld or delayed) and, in the case of a settlement or compromise by Parent, with the prior consent of the Representative (which consent will not be unreasonably withheld or delayed); provided, however, that with respect to Losses relating solely to the payment of money damages and that will not result in Parent becoming subject to injunctive or other relief, if the amount of the Third Party Claim does not exceed the amount then held in the Escrow Fund (*less* the aggregate amount of (i) any pending, disputed claims and (ii) any claims which are undisputed or which have been resolved, but not yet paid from the Escrow Fund) and such settlement includes a complete release of the Indemnified Parties from all further liability and no admission of liability or culpability on the part of the Indemnified Parties, the Representative may settle or compromise any such claim without the prior written consent of Parent; provided, further, that no settlement of any such Third Party Claim with third party claimants shall be determinative of the amount of Losses relating to such matter; provided, however, that the foregoing shall not alter or affect the right of the Indemnified Party to recover any amounts actually paid in settlement of such Third Party Claim from the Escrow Fund to the extent recoverable pursuant to Section 7.3(a) hereof. The Representative may settle or compromise any Third Party Claim only if it has agreed to defend such Third Party Claim in accordance with the applicable provisions of this Section 7.7 and, if any proposed settlement or compromise of any such Third Party Claim includes as an element thereof any non-monetary remedy or relief or any other non-monetary obligation that will be binding on the Company or any other Indemnified Party, the Representative shall not accept or otherwise effect such settlement or compromise, and such settlement or compromise shall not be effective or binding upon the Company or any other Indemnified Party, without the prior written consent of Parent, which consent may withheld or granted by Parent in its sole discretion, for any reason or no reason.
- 7.8 *Warranty Claims*. The Surviving Corporation and its designated Subsidiaries, Affiliates and subcontractors shall have the exclusive right and authority to respond to, administer and settle all Warranty Claims of the type described in **Section 7.3(a)(iv)(B)** and to provide the products and services necessary to perform or otherwise discharge the obligations of the Surviving Corporation or its Subsidiaries with respect to such Warranty Claims and, accordingly, in no event will any such Warranty Claim be deemed to constitute a Third Party Claim for purposes of this Agreement.

ARTICLE VIII

TERMINATION, AMENDMENT AND WAIVER

- 8.1 *Termination*. This Agreement may be terminated at any time prior to the Effective Time, by action taken or authorized by the Board of Directors of the terminating party or parties, whether before or after the requisite approvals of the stockholders of the Company, as follows:
 - (k) by mutual written consent duly authorized by the Boards of Directors of Parent and the Company;

- (l) by either the Company or Parent if the Merger shall not have been consummated before October 9, 2013 (the "Outside Date"); provided, however, that if 100% of the Company Stockholders have not approved the Merger in accordance with the applicable provisions of the DGCL on or before the Outside Date, then the Outside Date shall be extended to the date that is twenty one (21) days following the mailing of the notice of appraisal or dissenters rights to the Company Stockholders in accordance with Section 262 of the DGCL; provided, further, that if the Closing shall not have occurred by the Outside Date, but on such date, all of the conditions to Closing set forth in Article VI other than (i) conditions that by their nature are only to be satisfied as of the Closing and (ii) the conditions set forth in Section 6.1(b) (but solely to the extent the matter giving rise to the failure of such condition is related to the HSR Act or equivalent foreign antitrust Legal Requirements, if any) or Section 6.1(c), have been satisfied or waived in writing, then the Outside Date shall be extended by one (1) month; provided, however, that the right to terminate this Agreement under this Section 8.1(b) shall not be available to any party whose action or failure to act has been a principal cause of or resulted in the failure of the Merger to occur on or before such date and such action or failure to act constitutes a breach of this Agreement;
- (m) by either the Company or Parent if a Governmental Entity of competent jurisdiction shall have issued an order, decree or ruling or taken any other action (including the failure to have taken an action), in any case, having the effect of permanently restraining, enjoining or otherwise prohibiting the Merger, which order, decree, ruling or other action is final and nonappealable;
- (n) by the Company, upon a breach of any representation, warranty, covenant or agreement on the part of Parent set forth in this Agreement, or if any representation or warranty of Parent shall have become untrue, in either case such that the conditions set forth in **Section 6.2(a)** or **Section 6.2(b)** would not be satisfied as of the time of such breach or as of the time such representation or warranty shall have become untrue, provided that if such inaccuracy in Parent's representations and warranties or breach by Parent is curable by Parent prior to the Outside Date through the exercise of commercially reasonable efforts, then the Company may not terminate this Agreement under this **Section 8.1(d)** prior to thirty (30) days following the receipt of written notice from the Company to Parent of such breach (it being understood that the Company may not terminate this Agreement pursuant to this **Section 8.1(d)** if it shall have materially breached this Agreement or if such breach by Parent is cured so that such conditions would then be satisfied); or
- (o) by Parent, upon a breach of any representation, warranty, covenant or agreement on the part of the Company set forth in this Agreement, or if any representation or warranty of the Company shall have become untrue, in either case such that the conditions set forth in **Section 6.3(a)** or **Section 6.3(b)** would not be satisfied as of the time of such breach or as of the time such representation or warranty shall have become untrue, provided that if such inaccuracy in the Company's representations and warranties or breach by the Company is curable by the Company prior to the Outside Date through the exercise of commercially reasonable efforts, then Parent may not terminate this Agreement under this **Section 8.1(e)** prior to the thirty (30) days following the receipt of written notice from Parent to the Company of such breach (it being understood that Parent may not terminate this Agreement pursuant to this **Section 8.1(e)** if it shall have materially breached this Agreement or if such breach by the Company is cured so that such conditions would then be satisfied).
- 8.2 **Notice of Termination; Effect of Termination**. Any termination of this Agreement under **Section 8.1** will be effective (subject to the cure periods provided above) immediately upon the delivery of a valid written notice of the terminating party to the other party hereto. In the event of the termination of this Agreement as provided in **Section 8.1**, this Agreement shall be void and of no further force or effect and neither party shall have any liability hereunder, except (a) as set forth in **Section 5.3(a)**, this **Section 8.2**, **Section 8.3** and **Article IX**, each of which shall survive the termination of this Agreement and (b) nothing herein shall relieve any party from liability for breach of any covenant set forth in **Article IV** or **Article V** of this Agreement or fraud or any willful and knowing breach of this Agreement; *provided*, *however*, that the Company shall have no liability for any corrections, supplements or unknown omissions made pursuant to **Section 4.2** upon termination of this Agreement. No termination of this Agreement shall affect the obligations of the parties contained in the Confidentiality Agreement, all of which obligations shall survive termination of this Agreement in accordance with their terms.

- 8.3 *Amendment*. Subject to applicable Legal Requirements, this Agreement may be amended by the parties hereto, by action taken or authorized by their respective Boards of Directors, at any time before or after obtaining the Required Stockholder Approval, provided that after obtaining the Required Stockholder Approval, no amendment shall be made which requires further approval by the stockholders of the Company without such further stockholder approval. This Agreement may not be amended except by execution of an instrument in writing signed on behalf of each of Parent, Merger Sub, the Company and the Representative.
- 8.4 Extension; Waiver. At any time prior to the Effective Time either party hereto, by action taken or authorized by their respective Board of Directors, may, to the extent legally allowed: (a) extend the time for the performance of any of the obligations or other acts of the other parties hereto; (b) waive any inaccuracies in the representations and warranties made to such party contained herein or in any document delivered pursuant hereto; and (c) waive compliance with any of the agreements or conditions for the benefit of such party contained herein; provided, however, that any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such party. Delay in exercising any right under this Agreement shall not constitute a waiver of such right.

ARTICLE IX

GENERAL PROVISIONS

- 9.1 *Notices*. All notices and other communications hereunder shall be in writing and shall be deemed duly given (a) on the date of delivery if delivered personally or (b) on the date of confirmation of receipt (or, the first Business Day following such receipt if the date is not a Business Day) if delivered by a nationally recognized courier service. All notices hereunder shall be delivered as set forth below, or pursuant to such other instructions as may be designated in writing by the party to receive such notice:
 - (a) if to Parent or Merger Sub, to:

EnerSys Capital Inc. 2366 Bernville Road Reading, PA 19605 Attention: President

Telephone No.: (610) 208 1991

(b) and to:

EnerSys 2366 Bernville Road Reading, PA 19605 Attention: General Counsel (610) 208 1991

(c) with a copy to:

Stevens & Lee, P.C. 111 North 6th Street Reading, PA 19601

Attention: John A. Harenza, Esquire Telephone No.: (610) 478-2128

(d) if to the Company, to:

Purcell Systems, Inc. 16125 E. Euclid Avenue Spokane Valley, WA 99216 Attention: Chief Executive Officer Telephone No.: (509) 789-8865

(e) with a copy to:

Wilson Sonsini Goodrich & Rosati Professional Corporation 701 Fifth Avenue Suite 5100 Seattle, Washington 98104 Attention: Patrick J. Schultheis Telephone No.: (206) 883-2500

(f) and to:

Weston Presidio One Ferry Building, Suite 350 San Francisco, California 94111 Attention: Therese Mrozek Telephone No.: (415) 398-0770

(g) if to the Representative, to:

Fortis Advisors LLC

Attention: Notice Department Facsimile No.: (858) 408-1843 Email: notices@fortisrep.com

- 9.2 *Definitions*. For purposes of this Agreement, the following terms shall have the following definitions.
- (1) "Accrued Bonuses" shall mean, as of the Closing Date, (i) all accrued, unpaid bonuses (including all applicable withholdings) which are payable to, or have otherwise been earned by, any of the Employees pursuant to any of the Company Benefit Plans, which accrued amount represents the full amount of bonuses projected to be earned by all such Employees for the period ended December 31, 2013, and (ii) all related employer payroll taxes, but excluding the Change of Control Payments and Severance Payments.
- (2) "Acquisition Proposal" shall mean any inquiry, proposal or offer from any Person relating to (i) any sale or transfer of (A) the Company Business, (B) any material portion of the property or assets of the Company or any of its Subsidiaries other than for the purchase of inventory in the ordinary course of business, or (C) any of the Capital Stock or the stock or other equity or ownership interests of any of the Company's Subsidiaries other than the exercise of Company Options and Company Warrants outstanding as of the date of this Agreement, or (ii) any merger, consolidation, business combination, or similar transaction involving the Company or any of its Subsidiaries, in each case, other than the transactions contemplated by this Agreement.

- (3) "Adjusted Escrow Amount" shall mean the Escrow Amount minus the Escrowed Transaction Expenses.
- (4) "Adjusted Representative Escrow Amount" shall mean the Representative Escrow Amount <u>minus</u> the Escrowed Representative Transaction Expenses.
- (5) "Adjusted Working Capital Escrow Amount" shall mean the Escrow Amount minus the Escrowed Working Capital Transaction Expenses.
 - (6) "Advisory Group" has the meaning set forth in Section 7.6(d).
- (7) "Affiliate" of any Person means any Person that controls, is controlled by, or is under common control with such Person. As used herein, the term "control" (including the terms "controlling", "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities or other interests, by Contract or otherwise.
- (8) "Agreement" means this Amended and Restated Agreement and Plan of Merger, together with Schedule 7.3(b)(i) and Schedule 7.3(b)
 (ii) hereto, as such Amended and Restated Agreement and Plan of Merger may be further amended, restated or modified from time-to-time in accordance with the terms hereof.
- (9) "Backlog" means, collectively, as of any given date, the aggregate unpaid, undisputed purchase price for Company Products ordered from, but not yet shipped by, the Company or any of its Subsidiaries, but only to the extent that such orders are evidenced by written Contracts which are then outstanding and in effect and pursuant to which a customer of the Company or any of its Subsidiaries has, subject to the terms and conditions of such Contract, agreed or otherwise committed to purchase such products from the Company or any of its Subsidiaries.
 - (10) "Basket Amount" has the meaning set forth in Section 7.3(b).
- (11) "Business Day" shall mean a day other than Saturday, Sunday or any day on which banks located in the State of Washington and the State of Delaware are authorized or obligated to close.
 - (12) "Capital Stock" means the Common Stock and the Preferred Stock.
 - (13) "Certificate of Incorporation" means the Amended and Restated Certificate of Incorporation of the Company, as amended.
 - (14) "Certificate of Merger" has the meaning set forth in Section 1.2.
 - (15) *"Certificates"* has the meaning set forth in Section 1.7(c).
- (16) "Change of Control Agreements" shall mean the Transaction Bonus Plan Paul Brown, attached as Exhibit A to the Employment Agreement between the Company and Paul Brown, dated May 28, 2012 and the Transaction Bonus Plan Michael T. DuBose, attached as Exhibit A to the Employment Agreement between the Company and Michael T. DuBose, dated January 16, 2012, as amended on January 29, 2011 [sic].
- (17) "Change of Control Payments" shall mean all payments (including any Tax gross-up payments and all applicable withholdings) and all related employer payroll taxes which are payable in connection with any change of control obligation arising under the individually negotiated Change of Control Agreements, including the change of control payment obligations of the Company which will arise by reason of the change of

control which will occur with respect to the Company as a result of the Merger, but excluding the Severance Payments.

- (18) "Closing" has the meaning set forth in Section 1.2.
- (19) "Closing Balance Sheet" has the meaning set forth in Section 1.6(h)(ii).
- (20) "Closing Cash Amount" means the sum of all cash and cash equivalents of the Company and its Subsidiaries as of the Closing; provided, however, that "Closing Cash Amount" (i) shall be calculated net of all outstanding and unpaid checks and wires from the Company and its Subsidiaries to third parties that have not been reflected in the cash balance of the bank accounts of the Company or its Subsidiaries as of the Closing and (ii) shall include all checks and wires from third parties to the Company and its Subsidiaries that have been delivered to the Company or its Subsidiaries (with respect to checks) or have been initiated for delivery to the Company or its Subsidiaries (with respect to wires) as of the Closing, but which have not been reflected in the cash balance of the bank accounts of the Company or its Subsidiaries as of the Closing.
 - (21) "Closing Date" has the meaning set forth in Section 1.2.
- (22) "Closing Debt Amount" shall mean the portion of the Total Debt Amount to be paid at Closing to the parties listed on Schedule 9.2(22).
 - (23) "Closing Financial Data" has the meaning set forth in Section 1.6(h)(ii).
 - (24) "Closing Statement" has the meaning set forth in Section 1.6(h)(ii).
- (25) "Closing Transaction Expenses" shall mean the following: (i) all legal, accounting, tax, financial advisory and consulting fees of third parties incurred by the Company or any of its Subsidiaries at or prior to the Closing in connection with the negotiation and effectuation of the terms and conditions of this Agreement and the transactions contemplated hereby and outstanding as of the Effective Time, (ii) the Change of Control Payments outstanding as of the Effective Time, (iii) the Accrued Bonuses, (iv) the Severance Payments, (v) the premium for the extended directors' and officers' liability insurance coverage as provided in Section 5.8(c) and (vi) the fifty percent (50%) of all filing and other fees, excluding any legal fees of counsel, required under the HSR Act to be paid by the Company.
 - (26) "Closing Working Capital" shall mean an amount equal to the Current Assets minus the Current Liabilities.
- (27) "Closing Working Capital Components" shall mean the Closing Working Capital Deficit (if any), the Closing Working Capital Surplus (if any), and the Closing Cash Amount.
- (28) "Closing Working Capital Deficit" shall mean an amount, if any, by which the Closing Working Capital is less than the Closing Working Capital Target.
- (29) "Closing Working Capital Surplus" shall mean an amount, if any, by which the Closing Working Capital is greater than the Closing Working Capital Target.
- (30) *"Closing Working Capital Target"* shall mean Fifteen Million Nine Hundred Fifty Eight Thousand Eight Hundred Dollars (\$15,958,800).
 - (31) "Code" has the meaning set forth in Section 1.7(d).

- (32) "Common Stock" means the common stock, par value \$0.001 per share, of the Company.
- (33) *"Company"* has the meaning set forth in the Preamble.
- (34) "Company Balance Sheet" has the meaning set forth in Section 2.4.
- (35) "Company Benefit Plans" has the meaning set forth in Section 2.12(b)(i).
- (36) "Company Business" has the meaning set forth in Section 2.8(a)(i).
- (37) "Company Charter Documents" has the meaning set forth in Section 2.1(b).
- (38) *"Company Disclosure Letter"* means the disclosure letter supplied by the Company to Parent dated as of the date of the Prior Agreement.
 - (39) "Company Indemnified Liabilities" has the meaning set forth in the Section 5.3(a).
 - (40) "Company Indemnified Parties" has the meaning set forth in the Section 5.8(a).
 - (41) "Company Indemnified Proceedings" has the meaning set forth in the Section 5.8(a).
 - (42) "Company Intellectual Property Rights" has the meaning set forth in Section 2.8(a)(ii).
 - (43) "Company Material Contract" has the meaning set forth in Section 2.16(a).
 - (44) "Company Option Plan" means the 2004 Stock Plan of the Company.
- (45) "Company Options" means the outstanding options to purchase shares of Company Stock issued by the Company (including, but not limited to, options issued to employees of the Company).
 - (46) *"Company Participants"* has the meaning set forth in the **Section 5.7(a)**.
 - (47) *"Company Permits"* has the meaning set forth in the **Section 2.9(b)**.
 - (48) "Company Products" has the meaning set forth in Section 2.8(a)(iii).
 - (49) "Company Proposals" has the meaning set forth in Section 5.1.
 - (50) "Company Registered Intellectual Property Rights" has the meaning set forth in Section 2.8(b).
- (51) "Company Securities" means, collectively, the Capital Stock, Company Options or Company Warrants as of immediately prior to the Effective Time.
 - (52) "Company Securityholder" shall mean a holder of Company Securities as of immediately prior to the Effective Time.
 - (53) "Company Stockholder" shall mean a holder of Capital Stock as of immediately prior to the Effective Time.

- (54) "Company Warrant" means the outstanding warrant to purchase Capital Stock.
- (55) "Confidential Information" has the meaning set forth in Section 2.8(a)(iv).
- (56) "Confidentiality Agreement" has the meaning set forth in the Section 5.3.
- (57) "*Contract*" shall mean any written, oral or other legally binding agreement, contract, subcontract, settlement agreement, lease, instrument, note, warranty, purchase order, license, sublicense, or other legally binding commitment.
 - (58) "Controlled Group Affiliate" has the meaning set forth in Section 2.12(c).
 - (59) "Copyrights" has the meaning set forth in Section 2.8(a)(v).
- (60) "Current Assets" shall mean only those current assets of the Company and its Subsidiaries itemized on Schedule 1.6(h) as of the Closing, as determined on a consolidated basis and in accordance with (i) GAAP applied on a basis consistent with the Company Balance Sheet and (ii) the principles and methodologies set forth on Schedule 1.6(h).
- (61) "Current Liabilities" shall mean only those current liabilities of the Company and its Subsidiaries itemized on Schedule 1.6(h) as of the Closing Date, as determined on a consolidated basis and in accordance with (i) GAAP applied on a basis consistent with the Company Balance Sheet and (ii) the principles and methodologies set forth on Schedule 1.6(h).
 - (62) "Delaware Law" has the meaning set forth in the Recitals.
- (63) "Designated Person" shall mean each of Weston Presidio Capital IV, L.P., WPC Entrepreneur Fund II, L.P., Michael T. DuBose and Paul Brown.
 - (64) "DGCL" has the meaning set forth in Section 1.3.
 - (65) "Dispute Notice" has the meaning set forth in Section 1.6(h)(iii).
 - (66) "Disputed Components" has the meaning set forth in Section 1.6(h)(v).
 - (67) "Effective Time" has the meaning set forth in Section 1.2.
 - (68) "Employee" has the meaning set forth in Section 2.12(b)(i).
- (69) "Environmental Laws" shall mean all Legal Requirements relating to pollution, protection of the environment or exposure of any individual to Hazardous Materials, including all Legal Requirements relating to emissions, discharges, releases or threatened releases, investigation or remediation of Hazardous Materials, or otherwise relating to the manufacture, processing, registration, distribution, labeling, recycling, use, treatment, storage, disposal, transport or handling of Hazardous Materials.
 - (70) "ERISA" has the meaning set forth in Section 2.12(b)(i).
- (71) "Escrow Agent" shall mean Union Bank / Bank of Tokyo-Mitsubishi, UFJ, or any successor as determined in accordance with the Escrow Agreement.

- (72) "Escrow Agreement" has the meaning set forth in the Recitals.
- (73) "Escrow Amount" shall mean Six Million Two Hundred Fifty Thousand Dollars (\$6,250,000.00).
- (74) "Escrowed Representative Transaction Expenses" shall mean the portion of the financial advisory fees of Lazard and the portion of the Change of Control Payments to each of Michael T. DuBose and Paul Brown (if any) payable on the release of the Representative Fund as set forth on Schedule 1.6(d) and in accordance with Section 7.5 and the terms set forth in the agreements between the Company and each respective Other Participant.
- (75) "Escrowed Transaction Expenses" shall mean the portion of the financial advisory fees of Lazard and the portion of the Change of Control Payments to each of Michael T. DuBose and Paul Brown (if any) payable on the Escrow Termination Date as set forth on Schedule 1.6(d) and in accordance with Section 7.5 and the terms set forth in the agreements between the Company and each respective Other Participant.
- (76) "Escrowed Working Capital Transaction Expenses" shall mean the portion of the Change of Control Payments to each of Michael T. DuBose and Paul Brown (if any) payable on the release of the Working Capital Fund as set forth on Schedule 1.6(d) and in accordance with Section 7.5 and the terms set forth in the agreements between the Company and each respective Other Participant.
- (77) *"Escrow Fund"* shall mean the Escrow Amount held by the Escrow Agent in accordance with this Agreement and the Escrow Agreement together with all interest on or other taxable income, if any, earned from the investment of the Escrow Amount.
 - (78) "Escrow Termination Date" has the meaning set forth in Section 7.1.
 - (79) "Estimated Closing Balance Sheet" has the meaning set forth in Section 1.6(h)(i).
 - (80) "Estimated Closing Cash Amount" has the meaning set forth in Section 1.6(h)(i).
 - (81) "Estimated Closing Statement" has the meaning set forth in Section 1.6(h)(i).
 - (82) "Estimated Closing Working Capital" has the meaning set forth in Section 1.6(h)(i).
 - (83) "Estimated Closing Working Capital Deficit" has the meaning set forth in Section 1.6(h)(i).
 - (84) "Estimated Closing Working Capital Surplus" has the meaning set forth in Section 1.6(h)(i).
 - (85) "Exchange Fund" has the meaning set forth in Section 1.7(b).
- (86) "Export Control Law" shall mean (i) all U.S. import and export Legal Requirements (including those Legal Requirements under the authority of U.S. Departments of Commerce (Bureau of Industry and Security) codified at 15 CFR, Parts 700-799; Homeland Security (Customs and Border Protection) codified at 19 CFR, Parts 1-199; State (Directorate of Defense Trade Controls) codified at 22 CFR, Parts 103, 120-130; and Treasury (Office of Foreign Assets Control) codified at 31 CFR, Parts 500-599), and (ii) all comparable applicable Legal Requirements outside the United States.
 - (87) "Financial Statements" has the meaning set forth in Section 2.4.

- (88) "Fundamental Representations" has the meaning set forth in Section 7.1.
- (89) "GAAP" has the meaning set forth in Section 2.4.
- (90) "General Representations" has the meaning set forth in Section 7.1.
- (91) "General Representations Termination Date" has the meaning set forth in Section 7.1.
- (92) "Governmental Entity" has the meaning set forth in Section 2.3(d).
- (93) "Guarantor" has the meaning set forth in Section 9.15.
- (94) "Guarantor Undertakings" has the meaning set forth in Section 9.15.
- (95) "Hazardous Materials" shall mean any substance that is defined or listed in, or otherwise classified by a Governmental Entity pursuant to, any applicable Environmental Laws or Legal Requirement as a "hazardous substance," "hazardous material," "hazardous air pollutant," "extremely hazardous substance," "hazardous waste," "toxic substance," "toxic pollutant," or any other substance that is defined, listed, regulated or classified by a Governmental Entity under Environmental Laws by reason of its potentially deleterious properties such as ignitability, flammability, corrosivity, reactivity, combustibility, dispersability, volatility, carcinogenicity, or toxicity including, radioactive materials, explosives, urea formaldehyde, asbestos, polychlorinated biphenyls and petroleum products.
 - (96) "HSR Act" has the meaning set forth in Section 2.3(d).
 - (97) "Inbound Licenses" has the meaning set forth in Section 2.8(f).
- (98) "Indemnified Parties" shall mean Parent and the Surviving Corporation and their respective officers and directors, Subsidiaries and Affiliates and successors and assigns.
 - (99) "Independent Accountant" has the meaning set forth in Section 1.6(h)(v).
 - (100) "*Intellectual Property*" has the meaning set forth in Section 2.8(a)(v).
 - (101) "Intellectual Property Rights" has the meaning set forth in Section 2.8(a)(v).
 - (102) "International Employee Plan" has the meaning set forth in Section 2.12(e).
 - (103) "*Inventions*" has the meaning set forth in Section 2.8(a)(iv).
- (104) "Knowledge" with respect to the Company, with respect to any matter in question, shall mean the actual knowledge of (i) Michael T. DuBose, (ii) Paul Brown, (iii) Robert Kjelldorff, (iv) Barry Kantner; (v) Chris Craig, or (vi) John Rokus with respect to such matter after reasonable investigation or inquiry. As used herein, the term "reasonable investigation or inquiry" means, with respect to any of the foregoing Persons, (a) a reasonable review of the relevant books, records and properties of the Company and its Subsidiaries (as appropriate) for which such Person is responsible (including the review of the Phase I Environmental Site Assessment prepared by ERM of the facility located at ul. Wojska Plskiego 3, Mielec, Poland dated March 2013 and the Phase I Environmental Site Assessment of the Spokane Valley, Washington facility by All Appropriate Inquiries Environmental Corporation dated February 18, 2013), or to which such Person has access, in the ordinary course of business consistent with past practice and (b) a reasonable inquiry following any disclosures made in the ordinary

course of the Company's business consistent with past practice by those employees of the Company and its Subsidiaries for whom such Person is responsible or who directly report to such Person; provided, however, with respect to Intellectual Property, "reasonable investigation or inquiry" does not require the Company, nor any of the foregoing Persons, to conduct, have conducted, obtain or have obtained any freedom-to-operate opinions or similar opinions of counsel or any patent, trademark or other Intellectual Property clearance searches, and the Company will not be imputed to have knowledge of any third party patents, trademarks or other Intellectual Property that were not known by the foregoing Persons but that would have been revealed solely as a result of conducting such inquiries, opinions or searches and not as a result of conducting the reasonable investigation or inquiry referred to in clauses (a) and (b) above; provided, further, with respect to environmental matters in Section 2.15 (Environmental Matters), "reasonable investigation or inquiry" does not require the Company, nor any of the foregoing Persons, to conduct any surface or subsurface invasive investigation or record search of governmental agency files or records; provided, further, with respect to (1) the first and third sentences of Section 2.13(b), (2) Section 2.15(a)(ii) and (3) the second sentence of Section 2.15(b) only, "Knowledge" shall mean the actual knowledge of the foregoing Persons without any duty of reasonable investigation or inquiry.

- (105) "Lazard" shall mean Lazard Middle Market LLC.
- (106) "Leased Real Property" has the meaning set forth in Section 2.13.
- (107) "*Leases*" has the meaning set forth in **Section 2.13**.
- (108) "Legal Requirements" shall mean any federal, state, local, foreign or other law, statute, constitution, principle of common law, ordinance, code, order, edict, decree, rule, regulation, ruling or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into legal effect by or under the authority of any Governmental Entity.
 - (109) "*Liens*" has the meaning set forth in Section 2.1(c).
- (110) "Losses" shall mean all losses, liabilities, damages, deficiencies, costs and expenses (including reasonable attorneys' fees and expenses and, with respect to Warranty Claims, all internal or external labor costs associated with the repair, replacement or other correction of any defective or nonconforming products); provided, however, that "Losses" shall not include (i) any punitive, consequential, incidental, indirect, special or exemplary damages, unless and to the extent that any of such damages are actually paid or incurred by an Indemnified Party to a third party in connection with a Warranty Claim or a Third Party Claim, or (ii) any damages resulting from lost profits or diminution in value or that are based on a "multiplier" of the amount of any Loss.
- (111) "Material Adverse Effect," when used in connection with an entity, means any change or effect that has occurred prior to the date of determination of the occurrence of the Material Adverse Effect, that is materially adverse to the business, assets, liabilities, properties, financial condition, operations, or results of operations of such entity, taken as a whole with its Subsidiaries; provided, however, in no event shall any of the following, alone or in combination, be deemed to constitute, nor shall any of the following be taken into account in determining whether there has been or will be, a Material Adverse Effect on any entity: (i) any change or effect resulting from compliance with the terms and conditions of this Agreement; (ii) any change or effect that results from changes affecting any of the industries in which such entity operates generally or the United States or worldwide economy generally (which changes in each case do not disproportionately affect such entity in any material respect); (iii) any natural disaster or any acts of terrorism, sabotage, military action or war (whether or not declared) or any escalation or worsening thereof; (iv) failure to meet internal forecasts or financial projections; (v) in the case of the Company only, any change or effect resulting solely from the announcement or pendency of the Merger, including loss of any employees, customers, suppliers, partners or distributors; or (vi) any action or failure to act by the Company that is required or permitted by this Agreement.
 - (112) "*Merger*" has the meaning set forth in Section 1.1.

- (113) "Merger Consideration" means an amount of cash equal to (a) One Hundred Fifteen Million (\$115,000,000.00), minus (b) the Total Debt Amount, minus (c) the Closing Transaction Expenses that remain unpaid as of the Closing, minus (d) the Closing Working Capital Deficit, if any, plus (e) the Closing Working Capital Surplus, if any plus (f) the Closing Cash Amount.
 - (114) "Merger Consideration Excess Amount" has the meaning set forth in Section 1.6(h)(vi).
 - (115) "Merger Consideration Shortfall Amount" has the meaning set forth in Section 1.6(h)(vi).
 - (116) "Merger Sub" has the meaning set forth in the Preamble.
 - (117) "Necessary Consents" has the meaning set forth in Section 2.3(d).
 - (118) "Non-Competition Agreement" has the meaning set forth in the Recitals.
- (119) "Officer's Certificate" shall mean a certificate signed by any officer of Parent: (i) stating that an Indemnified Party has paid, sustained or incurred Losses that the Indemnified Party is entitled to recover from the Escrow Fund pursuant to Section 7.3(a) and (ii) specifying in reasonable detail the individual items of Losses included in the amount so stated, the date each such item was paid, sustained or incurred, and the nature of the misrepresentation, breach of warranty or covenant to which such item is related.
 - (120) "Open Source Materials" has the meaning set forth in Section 2.3(d).
 - (121) "Option Consideration" has the meaning set forth in Section 1.6(b).
 - (122) "Other Participants" shall mean Michael T. DuBose and Paul Brown.
 - (123) "Outbound Licenses" has the meaning set forth in Section 2.8(h).
 - (124) "Outside Date" has the meaning set forth in the Section 8.1(b).
 - (125) *"Parent"* has the meaning set forth in the Preamble.
 - (126) "Patent Rights" has the meaning set forth in the Section 2.8(a)(v).
 - (127) "Per Share Amount" means the amount set forth on Schedule 1.6(d) for each share of Capital Stock.
 - (128) "*Permits*" has the meaning set forth in the **Section 2.9(b)**.
- (129) "Permitted Liens" shall mean (i) liens for Taxes and other similar governmental charges and assessments which are not yet due and payable, or which are being contested in good faith and for which adequate reserves have been established in accordance with GAAP, (ii) liens of landlords and liens of carriers, warehousemen, mechanics and materialmen and other like liens arising in the ordinary course of business for sums not yet due and payable, and (iii) liens and encumbrances imposed on the underlying interest in the Leased Real Property.
- (130) "Person" shall mean any individual, corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust, company (including

any limited liability company or joint stock company), firm or other enterprise, association, organization, entity or Governmental Entity.

- (131) "*Post-Closing Debt Amount*" shall the portion of the Total Debt Amount that is not to be paid at Closing, as set forth on **Schedule 1.6(h)**, and interest thereon, accrued and outstanding as of the Closing, together with any accrued or unpaid costs and expenses relating thereto.
 - (132) "Pre-Closing Tax Period" shall mean any taxable period or portion thereof that ends on or prior to the Closing Date.
 - (133) "Preferred Stock" shall mean the Series A Preferred Stock and the Series B Preferred Stock.
 - (134) "Prior Agreement" has the meaning set forth in the Recitals.
- (135) "Pro Rata Portion" with respect to each Company Securityholder who holds shares of Company Securities, the quotient obtained by dividing (i) the sum of the number of shares of Capital Stock, the number of shares of Capital Stock subject to outstanding Company Warrants and the number of shares of Capital Stock subject to outstanding Company Options held by such Company Securityholder immediately prior to the Effective Time by (ii) the sum of the number of shares of Capital Stock, the number of shares of Capital Stock subject to outstanding Company Warrants and the number of shares of Capital Stock subject to outstanding Company Options held by all Company Securityholders immediately prior to the Effective Time.
 - (136) "Registered Intellectual Property Rights" has the meaning set forth in Section 2.8(a)(vi).
 - (137) "*Representative*" has the meaning set forth in the Preamble.
- (138) "*Representative Agreement*" shall mean the letter dated September 15, 2013 between the Company, the Advisory Group and the Representative.
 - (139) "Representative Escrow Amount" shall mean an amount of cash equal to Five Hundred Thousand Dollars (\$500,000.00).
- (140) "*Representative Fund*" shall mean the Representative Escrow Amount held by the Representative in accordance with this Agreement.
 - (141) "Representative Losses" has the meaning set forth in Section 7.6(c).
 - (142) "Required Stockholder Approvals" has the meaning set forth in the Recitals.
 - (143) "Resolution Period" has the meaning set forth in Section 1.6(h)(iv).
 - (144) "Series A Preferred Stock" shall mean the Series A Preferred Stock, par value \$0.001 per share, of the Company.
 - (145) "Series B Preferred Stock" shall mean the Series B Preferred Stock, par value \$0.001 per share, of the Company.
- (146) "Severance Payments" shall mean, collectively, all payments in the nature of severance (including any Tax gross-up payments and all applicable withholdings) and all related employer payroll taxes which

are or would be, payable to each of Michael T. DuBose and Paul Brown as of the Closing Date pursuant to the respective Employment Agreements identified on **Schedule 9.2(145)** by reason of the cessation of their employment with the Company in connection with the change of control which will occur with respect to the Company as a result of the Merger.

- (147) "Subsidiary" when used with respect to any party, shall mean any corporation or other organization, whether incorporated or unincorporated, at least a majority of the securities or other interests of which having by their terms ordinary voting power to elect a majority of the Board of Directors or others performing similar functions with respect to such corporation or other organization is directly or indirectly owned or controlled by such party or by any one or more of its Subsidiaries, or by such party and one or more of its Subsidiaries.
 - (148) "Subsidiary Charter Documents" has the meaning set forth in Section 2.1(b).
 - (149) "Surviving Corporation" has the meaning set forth in Section 1.1.
 - (150) "*Tax*" or "*Taxes*" has the meaning set forth in Section 2.7(a).
 - (151) "*Tax Returns*" has the meaning set forth in Section 2.7(b)(i).
 - (152) *"Third Party Claim"* has the meaning set forth in the **Section 7.7**.
- (153) "Total Debt Amount" shall mean, as of the Closing, (i) the outstanding principal amount of all indebtedness for borrowed money or other funded debt of the Company or any of its Subsidiaries, and all interest thereon, (ii) all obligations of the Company or any of its Subsidiaries as lessee under any lease that would be required to be capitalized in accordance with GAAP, (iii) any outstanding fees, costs or expenses owed by the Company or any of its Subsidiaries with respect to any such debt or capital lease obligations and (iv) any prepayment, breakage, make-whole, termination, cancellation, or similar fees, costs, or expenses due as a result of any prepayment of any of such debt or capital lease obligations or the termination of the underlying credit facility to which they relate.
 - (154) "Trade Secret Rights" has the meaning set forth in Section 2.8(a)(v).
 - (155) "Trademark Rights" has the meaning set forth in Section 2.8(a)(v).
 - (156) "*Trademarks*" has the meaning set forth in Section 2.8(a)(iv).
 - (157) "*Trade Secrets*" has the meaning set forth in Section 2.8(a)(iv).
- (158) "Unpaid Pre-Closing Taxes" shall mean: (i) any Taxes of the Company or any of its Subsidiaries related to a Pre-Closing Tax

 Period; (ii) any Taxes of the Company or any of its Subsidiaries as a result of being (or ceasing to be) on or prior to the Closing Date a member of an affiliated or combined group pursuant to Treasury Regulations Section-1.1502 6 or any similar provision of state, local or foreign law; or (iii) any Taxes for which the Company or any of its Subsidiaries is liable as a transferee or successor by Contract or otherwise. For the purpose of computing the amount of Unpaid Pre-Closing Taxes the following principles shall be applied: (A) in the case of income Taxes and all other Taxes that are not imposed on a periodic basis, the amount of such Taxes that have accrued up to and including the Closing Date for a period beginning before and ending after the Closing Date shall be deemed to be the amount that would be payable if the taxable year or period ended on the Closing Date based on an interim closing of the books, except that exemptions, allowances or deductions that are calculated on an annual basis (including depreciation and amortization deductions) shall be allocated on a per diem basis; and (B) in the case of any Taxes that are imposed on a periodic basis for a period beginning before and ending after the Closing Date, the amount of such Taxes that have accrued up to and including the Closing Date shall be the amount of such

Taxes for the relevant period multiplied by a fraction the numerator of which shall be the number of calendar days from the beginning of the period up to and including the Closing Date and the denominator of which shall be the number of calendar days in the entire period.

- (159) "Virtual Data Room" shall mean the electronic data room established on behalf of the Company by Merrill Corporation in connection with the transactions contemplated hereby.
 - (160) "Warrant Termination Amount" has the meaning set forth in Section 1.6(c).
- (161) "*Warranty Claim Period*" shall mean each of the three (3) consecutive twelve (12) month periods which comprise the thirty-six (36) month period commencing on the Closing Date and ending on the Escrow Termination Date.
 - (162) "Warranty Claims" has the meaning set forth in Section 7.3(a).
 - (163) "Warranty Deductible" has the meaning set forth in Section 7.3(b).
 - (164) "Working Capital Escrow Amount" shall mean an amount of cash equal to Three Million Dollars (\$3,000,000.00).
- (165) "Working Capital Fund" shall mean the Working Capital Escrow Amount held by the Escrow Agent in accordance with this Agreement and the Escrow Agreement, together with all interest on or other taxable income, if any, earned from the investment of the Working Capital Escrow Amount.
 - (166) "Works of Authorship" has the meaning set forth in Section 2.8(a)(iv).
 - (167) *"Written Consent"* has the meaning set forth in the Recitals.

9.3 Interpretation.

- (g) When a reference is made in this Agreement to paragraph, subparagraph, Section, subsection, clause, subclause, Article, Recital, Exhibit or Schedule, such reference shall be to a paragraph, subparagraph, Section, subsection, clause, subclause, Article, Recital, Exhibit or Schedule to this Agreement unless otherwise indicated.
- (h) When a reference is made in this Agreement to Sections, such reference shall be to a Section of this Agreement unless otherwise indicated.
- (i) For purposes of this Agreement, the words "include," "includes" and "including," when used herein, shall be deemed in each case to be followed by the words "without limitation."
- (j) The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.
- (k) When reference is made herein to "the business of" an entity, such reference shall be deemed to include the business of all such entity and its Subsidiaries, taken as a whole.
- (l) An exception or disclosure made in the Company Disclosure Letter with regard to a representation of the Company shall be deemed made with respect to any other representation by such party in accordance with the provisions of the introductory paragraph of **Article II**.

- (m) For purposes of this Agreement, the obligation to use "reasonable best efforts" and "commercially reasonable efforts" shall not require a party to pay any consent fee, profit sharing payment or any other consideration for any consent, approval or amendment (except for filing fees, other administrative charges and other reasonable out-of-pocket expenses).
- (n) Any dollar thresholds set forth herein shall not be used as a benchmark for determinations of what is or is not "material" under this Agreement.
- (0) Unless the context of this Agreement otherwise requires (i) words of any gender include each other gender; (ii) words using the singular or plural number also include the plural or singular number, respectively; and (iii) the terms "hereof," "herein," "hereunder" and derivative or similar words refer to this entire Agreement.
- (p) The Recitals to this Agreement (including all defined terms set forth therein) are hereby incorporated by reference into this Agreement and made a part hereof as if set forth in their entirety in this **Section 9.3(j)**.
- (q) All accounting terms used in this Agreement and in the Schedules hereto or the Company Disclosure Letter shall have the meanings given to them in accordance with GAAP.
- 9.4 *Counterparts*. This Agreement may be executed in two or more counterparts, all of which shall be considered one and the same agreement and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other party, it being understood that all parties need not sign the same counterpart.
- 9.5 Entire Agreement; Third-Party Beneficiaries. This Agreement and the documents and instruments and other agreements among the parties hereto as contemplated by or referred to herein, including the Company Disclosure Letter, constitute the entire agreement among the parties with respect to the subject matter hereof and supersede all prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof, including the Prior Agreement, it being understood that the Confidentiality Agreement shall continue in full force and effect until the Closing and shall survive any termination of this Agreement. Nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement other than (a) as specifically provided in Section 5.7 and Section 5.8; (b) the rights of Company Securityholders and the Other Participants to pursue claims for damages and other relief for Parent's or Merger Sub's breach of this Agreement; (c) the rights of holders of the Company Securityholders to receive the consideration specified in this Agreement; and (d) following the Closing, the rights of payees to receive the amounts described in **Section 5.7(b)**; provided, however, that if the Closing does not occur, the rights granted pursuant to clause (b) shall only be enforceable on behalf of such Company Securityholders and the Other Participants by the Company, or, following the Closing, the Representative, in their sole and absolute discretion. Subject in all respects to Section 7.3(e), each of the Parent and Merger Sub acknowledges and agrees that it has made its own inquiry and investigation into, and, based thereon, has formed an independent judgment concerning, the Company and its businesses and operations, and Parent and Merger Sub have been furnished with or given full access to such information about the Company and its businesses and operations as they requested. Subject in all respects to Section 7.3(e), Parent and Merger Sub acknowledge and agree that (i) there are uncertainties inherent in attempting to make such projections, forecasts, estimates, plans and budgets; (ii) Parent and Merger Sub are familiar with such uncertainties; and (iii) Parent and Merger Sub are taking full responsibility for making their own evaluations of the adequacy and accuracy of all estimates, projections, forecasts, plans and budgets so furnished to them or their representatives. Subject in all respects to Section 7.3(e), Parent and Merger Sub each acknowledge that, except as expressly set forth in Article II, there are no representations or warranties of any kind, express or implied, by the Company or any other Person, including (A) with respect to the Company, its Subsidiaries, or their assets and liabilities, or the transactions contemplated hereby, and (B) as to the accuracy or completeness of any information regarding the Company furnished or made available to Parent, Merger

Sub and their representatives (including any projections, forecasts, estimates, plans or budgets). Without limiting the generality of the foregoing, except as expressly set forth in this Agreement, there are no express or implied warranties of merchantability or fitness for a particular purpose with respect to any of the Company's tangible personal property.

- 9.6 Severability. In the event that any provision of this Agreement or the application thereof, becomes or is declared by a court of competent jurisdiction to be illegal, void or unenforceable, the remainder of this Agreement will continue in full force and effect and the application of such provision to other Persons or circumstances will be interpreted so as reasonably to effect the intent of the parties hereto. The parties further agree to replace such void or unenforceable provision of this Agreement with a valid and enforceable provision that will achieve, to the greatest extent possible, the economic, business and other purposes of such void or unenforceable provision.
- 9.7 *Other Remedies*. Prior to the Closing, except as otherwise provided herein, any and all remedies herein expressly conferred upon a party will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy; *provided*, *however*, the parties hereto expressly acknowledge and agree that the remedy of specific performance or any similar equitable remedies shall not be available to the Company or the Representative for the failure of Parent or Merger Sub to complete the Merger for any reason.
- 9.8 *Governing Law*. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, regardless of the laws that might otherwise govern under applicable principles of conflicts of law thereof.

9.9 Consent to Jurisdiction.

- (d) Each of Parent, the Company and Merger Sub hereby irrevocably submits to the exclusive jurisdiction and venue of the state courts of Delaware, New Castle County and to the exclusive jurisdiction and venue of the United States District Court for the District of Delaware, for the purpose of any action or proceeding arising out of or relating to this Agreement and each of Parent, the Company and Merger Sub hereby irrevocably agrees that all claims in respect to such action or proceeding may be heard and determined exclusively in any of such courts. Each of Parent, the Company and Merger Sub agrees that a final judgment in any action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law.
- (e) Each of Parent, the Company and Merger Sub irrevocably consents to the service of the summons and complaint and any other process in any other action or proceeding relating to the transactions contemplated by this Agreement, on behalf of itself or its property, by the personal delivery of copies of such process to such party. Nothing in this **Section 9.9** shall affect the right of any party to serve legal process in any other manner permitted by law.
- 9.10 Waiver of Jury Trial. EACH OF PARENT, MERGER SUB, THE COMPANY, THE COMPANY SECURITYHOLDERS, THE REPRESENTATIVE AND THE OTHER PARTICIPANTS HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE ACTIONS OF PARENT, MERGER SUB, THE COMPANY, OR THE REPRESENTATIVE IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE AND ENFORCEMENT HEREOF.
- 9.11 *Rules of Construction*. The parties hereto agree that they have been represented by counsel during the negotiation and execution of this Agreement and, therefore, waive the application of any law,

regulation, holding or rule of construction providing that ambiguities in an agreement or other document will be construed against the party drafting such agreement or document.

- 9.12 Assignment. No party may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other parties, except that following the Closing, Parent shall have the right, upon notice to the Representative, to (a) assign and delegate all of its rights, duties and obligations hereunder to any Affiliate of Parent, provided that Parent nevertheless remains secondarily liable to the Representative for any failure of such Affiliate to perform any of Parent's obligations hereunder and (b) pledge and assign any or all of Parent's rights hereunder to the bank or financial institution providing financing to Parent as collateral security for Parent's obligations to such bank or financial institution. Any purported assignment in violation of this Section 9.12 shall be void. Subject to the preceding sentence, this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and permitted assigns.
- 9.13 *Expenses*. Except as otherwise provided in this Agreement, all fees and expenses incurred in connection with the Merger, this Agreement and the transactions contemplated by this Agreement shall be paid by the party incurring such fees or expenses, whether or not the Merger is consummated.
- 9.14 *No Personal Liability*. Except as set forth herein, this Agreement will not create or be deemed to create or permit any personal liability or obligation on the part of any direct or indirect stockholder of the Company, Parent or Merger Sub or any officer, director, employee, agent, representative or investor of any party.
- 9.15 *Guarantee*. EnerSys, a Delaware corporation and the ultimate parent corporation of Parent ("*Guarantor*"), by its execution and delivery of the Limited Joinder appended to this Agreement, hereby unconditionally, absolutely, continuing and irrevocably guarantees, as surety, the full and complete payment and performance of all obligations and liabilities, whether now in existence or hereafter arising, of Parent, Merger Sub and, following the Effective Time, the Surviving Corporation under (a) this Agreement (but only to the extent that any such obligations of the Surviving Corporation are required by this Agreement to be paid or performed by the Surviving Corporation after the Effective Time and not by the Company prior to the Effective Time) and (b) the Escrow Agreement (the "*Guarantor Undertakings*"). The obligations of Guarantor with respect to the Guarantor Undertakings shall not be subject to any reduction, limitation, impairment or termination for any reason, including any claim of waiver, release, surrender, alteration, compromise, modification or amendment. Notice of acceptance of this guarantee and of the incurring of any obligation or default of the Guarantor Undertakings, as well as demand and protest with respect to such Guarantor Undertakings, are hereby waived by Guarantor.
- 9.16 *Effectiveness of Amendment and Restatement*. This Agreement amends and restates certain provisions of the Prior Agreement and restates the terms of the Prior Agreement in their entirety. All amendments to the Prior Agreement effected by this Agreement, and all other covenants, agreements, terms and provisions of this Agreement, shall have effect as of the date of the Prior Agreement unless expressly stated otherwise, and accordingly, the phrases "as of the date hereof," "as of the date of this Agreement," or any similar terms as used in this Agreement will be deemed to refer to the date of the Prior Agreement. This Agreement shall be effective as of the date that copies hereof have been executed and delivered by each of the parties hereto.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their duly authorized respective officers as of the date first written above.

ENERSYS CAPITAL INC.

By: <u>/s/ Richard W. Zuidema</u>
Name: Richard W. Zuidema
Title: Executive Vice President

ECI MERGER CORPORATION

By: <u>/s/ Richard W. Zuidema</u> Name: Richard W. Zuidema

Title: President

PURCELL SYSTEMS, INC.

By: <u>/s/ Michael T. DuBose</u> Name: Michael T. DuBose

Title: Chairman and Chief Executive Officer

FORTIS ADVISORS LLC Signing solely in its capacity as Representative

By: <u>/s/ Adam Lezack</u>
Name: Adam Lezack
Title: Managing Director

LIMITED JOINDER OF ENERSYS

THIS LIMITED JOINDER OF ENERSYS ("<u>Joinder</u>") is made as of the 30th day of September, 2013 by the undersigned, ENERSYS, a Delaware corporation ("*Guarantor*").

Guarantor hereby joins in the foregoing Amended and Restated Agreement and Plan of Merger dated as of September 30, 2013 (together with the Disclosure Letter, Schedules and Exhibits thereto, the "Agreement") among EnerSys Capital Inc., ECI Merger Corporation, Purcell Systems, Inc., and with respect to Article VII of the Agreement only, Fortis Advisors LLC, solely for the purposes of agreeing to be bound by the provisions of Section 9.15 (*Guarantee*) of the Agreement, to the same extent as if Guarantor was a party to the Agreement for the purposes of such Section 9.15.

IN WITNESS WHEREOF, the undersigned, intending to be legally bound hereby, has executed and delivered this Joinder as of the date first set forth above.

ENERSYS

By: /s/ Richard W. Zuidema

Name: Richard W. Zuidema Title: Executive Vice President

Certification of Principal Executive Officer Pursuant To Rule 13a-14(a)/15d-14(a) Under the Securities Exchange Act Of 1934

I, John D. Craig, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of EnerSys;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f)) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

ENERSYS

By /s/ John D. Craig

John D. Craig

Chairman, President and Chief Executive Officer

Date: November 6, 2013

Certification of Principal Financial Officer Pursuant To Rule 13a-14(a)/15d-14(a) Under the Securities Exchange Act Of 1934

I, Michael J. Schmidtlein, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of EnerSys;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(f)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

ENERSYS

By /s/ Michael J. Schmidtlein

Michael J. Schmidtlein Senior Vice President Finance & Chief Financial Officer

Date: November 6, 2013

CERTIFICATION OF THE CHIEF EXECUTIVE OFFICER AND CHIEF FINANCIAL OFFICER PURSUANT TO 18. U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I certify pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report of EnerSys on Form 10-Q for the quarterly period ended September 29, 2013, fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that information contained in such Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of EnerSys.

ENERSYS

By /s/ John D. Craig

John D. Craig

Chairman, President and Chief Executive Officer

By /s/ Michael J. Schmidtlein

Michael J. Schmidtlein

Senior Vice President Finance & Chief Financial Officer

Date: November 6, 2013